

88 Notch Hill Road North Branford, Connecticut 06471 (203) 488-8000 <u>www.Evergreen-Woods.com</u>

INFORMATION BOOKLET (**DISCLOSURE STATEMENT**)

January 2021

THIS FACILITY, LIKE ALL OTHER CONTINUING CARE (LIFE-CARE) FACILITIES IN THE STATE OF CONNECTICUT, IS SUBJECT TO SECTIONS 17b-520 THROUGH 17b-535 OF THE CONNECTICUT GENERAL STATUTES CONCERNING MANAGEMENT OF CONTINUING CARE FACILITIES (THE "ACT"). REGISTRATION UNDER THE ACT DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION OR ENDORSEMENT OF THE FACILITY BY THE DEPARTMENT OF SOCIAL SERVICES OR STATE OF CONNECTICUT, NOR DOES SUCH REGISTRATION EVIDENCE THE ACCURACY OR COMPLETENESS OF THE INFORMATION IN THIS DISCLOSURE STATEMENT.



Part I: Introduction; Overview

Introduction	6
Guiding Principles	6
Evergreen Woods Retirement, LLC	7
Maxwell Group, Inc	7
Leadership	
Administration	9
Judicial Proceedings	9
Part II: The Community	
Evergreen Woods	11
The Life-Care Concept	
The Plans	
Contracts And Fees	
The Services	12
The Residency Agreement	16
Tax Consequences	17
Monthly Service Fees	
Entrance Fees	19
Schedule of Miscellaneous Charges	
Part III: Financial Information	
Entrance Fee Escrow	
Operating Reserve Escrow and Debt Service Reserve Escrow	
Promissory Note	
Financing Of The Community	

The Pro Forma Cash Flow Statement	. 34
Pro Forma Cash Flow Assumptions	. 34

Part IV: Statutory Acknowledgements

Registration	. 40
Notice To Prospective Residents	. 40
Acknowledgement Of Receipt Of Disclosure Statement	. 40

EXHIBITS

A. E	xample	Traditional	Residency	Agreement
------	--------	-------------	-----------	-----------

- B. Example Unbundled Residency Agreement
- C. Sworn Statement of Entrance Fee Escrow Agent
- D. Sworn Statement of Operating and Debt Service Reserve Escrow Agent
- E. Audited Financial Statements

CROSS REFERENCES

The following information, which is required by law to be included in this Disclosure Statement, can be found on the referenced pages:

INFORMATION

1.	Name and Address of the Provider	7, 11
2.	Officers, Directors, and Trustees	
3.	Business Experience	7-9
4.	Judicial Proceedings	9
5.	Affiliations	7
6.	Description of Property	
7.	Benefits Included	
8.	Interest on Deposits	
9.	Termination of Contract	
10.	Rights of Surviving Spouse	16
11.	Marriage of Resident	16
	Disposition of Personal Property	
	Tax Consequences	
14.	Reserve Funding – Escrows	
15.	Statement Regarding Audited Financial Statements	
16.	Pro Forma Cash Flow Statement	
	Current Occupancy Rates	
	Entrance Fees/Periodic Charges/Ancillary Charges	
	Residency Agreements	
	Acknowledgement of Receipt of Disclosure Statement	
	Notice to Prospective Resident	

PART I: INTRODUCTION; OVERVIEW

INTRODUCTION

Evergreen Woods (the "Community") brings to residents of Connecticut, aged 62 and over, a form of retirement living with a full continuum of care available. Sometimes known as "life plan", "continuing care", or "life-care". This concept offers active retirees a lifestyle that is designed to meet their unique needs while allowing them the freedom to pursue their personal interests. Communities, such as Evergreen Woods, encompass four important components: a private apartment home, a wide array of available personal services, the security of an on-site professional Health Center, offered within a sound financial plan.

One of the purposes of this Information Booklet (Disclosure Statement) is to explain to prospective residents, their families and advisors, who and what is involved in the operation of the Community. This Information Booklet was prepared on the basis of information available at the time of its publication and assumptions that were believed to be realistic as of that date. Such information and assumptions are, of course, subject to change and in particular are significantly affected by changes in inflation and interest rates. In addition, there usually will be differences between the forecasted and actual results, because events and circumstances frequently do not occur as expected and those differences may be material. Moreover, the achievement of any financial forecast is dependent upon future events, the occurrence of which cannot be assured and future changes in circumstances and in the economy may require changes in the future operation of the Community.

Although we have prepared this Information Booklet carefully and have tried to use nontechnical language, it is possible that there may be some differences between the text in this booklet and certain of the contracts summarized herein. In the event of any difference, the provisions of such contracts shall govern.

GUIDING PRINCIPLES

The following Guiding Principles are the foundation upon which we operate:

People First, Always. We approach our business with a People Over Policy mindset. This allows our team members to make the decision at any given time which puts the Members we serve first. In addition, we believe in an empowered workforce. Our product is only as good as the hands that deliver it. That's why we extend our People First principle to those we're working alongside.

We Exist to Serve Our Members. If we lose sight of the individuals that we've built our community to serve and care for, then we don't deserve to be in business. This is why we stress our mission on a daily basis to everyone from our front-line caregivers to our administrative staff.

We Have A Responsibility to be Full. We believe that we have the best community offering the best services resulting in industry-best care. As our community grows we are able to serve more and more Members for whom this quality of life is the best option. Therefore, it is our responsibility to be full. And by maintaining our position as best-in-class in all that we offer, we

believe we can help shape a changing culture that allows our person-centered, dignity-focused philosophy to trickle down to every other community in the nation.

EVERGREEN WOODS RETIREMENT, LLC

The Community is operated by Evergreen Woods Retirement, LLC, a Connecticut limited liability company (the "Provider"), whose business address is 88 Notch Hill Road, North Branford, Connecticut 06471. The Provider is a wholly-owned subsidiary of Senior Living Communities, LLC, a North Carolina limited liability company ("SLC"). The executive offices of SLC and its subsidiaries, including the Provider, are located at 3530 Toringdon Way, Suite 204, Charlotte, North Carolina 28277. Donald O. Thompson, Jr. and the Brenda U. Thompson Irrevocable Trust dated December 28, 2020 own SLC and Maxwell Group, Inc., a North Carolina corporation ("Maxwell Group"), which provides management and marketing services for the Community. Donald Thompson is the CEO of SLC and the CEO of Maxwell Group and is responsible for the management of the Provider and the Community. SLC operates nine other retirement communities in North Carolina, South Carolina, Georgia and Florida that are similar to the Community, all of which are managed by Maxwell Group.

SLC's financing partner is National Health Investors, Inc., a Maryland corporation ("NHI") that is a real estate investment trust. NHI's stock trades on the New York Stock Exchange under the symbol "NHI." NHI-REIT of Evergreen, LLC, a Delaware limited liability company that is a wholly-owned subsidiary of NHI, owns the Community and leases it to SLC under a long-term lease. SLC, in turn, subleases the Community to the Provider for day-to-day operation.

Provider is solely responsible for its financial and contractual obligations, including its obligations under the Residency Agreements. No other person or entity referred to herein, including SLC, Maxwell Group, NHI or NHI-REIT of Evergreen, LLC, has assumed any financial responsibility for the fulfillment of Evergreen Woods' agreements, except as expressly stated.

Evergreen Woods is not affiliated with any religious, charitable or other non-profit organization.

MAXWELL GROUP, INC.

The Provider has retained Maxwell Group to manage the Community. Maxwell Group was founded in 1989 and has 32 years of proven success in the management of healthcare communities and other related senior living care entities. Maxwell Group has built its reputation on developing luxury communities for senior adults. As of the date of this Information Booklet, Maxwell Group manages 15 retirement communities in six states. Eight of the communities managed by Maxwell Group, including Evergreen Woods, are continuing care retirement communities. Maxwell Group's executive offices are located at 3530 Toringdon Way, Suite 204, Charlotte, North Carolina 28277. Maxwell Group is a team of people dedicated to creating luxury senior living communities by providing excellent service to all of its' customers.

Maxwell Group has been engaged to supervise and manage the operation of the Community, including: recruiting and training an Executive Director; housekeeping; flat laundry; interior and exterior building and grounds maintenance; providing and facilitating recreation and

social programs; providing scheduled transportation; on-going sales and marketing; the purchase of supplies, drugs and solutions; supervising the licensing, equipping, and staffing of the Community; preparation of annual budgets; preparing books, records and other accounts for the Community; establishing and operating a system of financial control for the Community and supervising health care services, food service and quality accommodations throughout the Community; coordinating the Community's compliance with governmental requirements; enforcement of Residency Agreements; and any other services as may from time to time be reasonably requested by the Provider. The services described herein are furnished pursuant to a management agreement between the Provider and Maxwell Group.

Maxwell Group does not guarantee the obligations of the Provider under the Residency Agreements. Maxwell Group is entitled to full reimbursement for certain costs incurred by it in connection with providing management services to the Provider, and is also paid a percentage fee based on the total collected revenue of Evergreen Woods.

RETIREMENT COMMUNITIES MANAGED BY MAXWELL GROUP, INC.

As of January 2021

The Stratford, Carmel, Indiana Homestead Hills, Winston-Salem, North Carolina Brightwater, Myrtle Beach, South Carolina Cascades Verdae, Greenville, South Carolina Evergreen Woods, North Branford, Connecticut The Lakes at Litchfield, Pawleys Island, South Carolina Marsh's Edge, St. Simons Island, Georgia Ridgecrest, Mt. Airy, North Carolina Summit Hills, Spartanburg, South Carolina Wellmore of Tega Cay, Tega Cay, South Carolina Wellmore of Daniel Island, Charleston, South Carolina Wellmore of Lexington, Lexington, South Carolina The Charlotte, Charlotte, North Carolina Wildewood Downs, Columbia, South Carolina Osprey Village at Amelia Island, Amelia Island, Florida

More information about Maxwell Group, Inc. can be obtained at <u>www.Maxwell-Group.com</u>

LEADERSHIP

Donald O. Thompson, Jr. serves as the CEO of SLC and as the CEO of Maxwell Group. Over the last three decades, Donald Thompson has been an owner of companies providing skilled nursing, Alzheimer's care, assisted living, rehab services, group purchasing, enteral feeding, medical equipment, medical supplies, home health services, and pharmacy services.

ADMINISTRATION

Overall administrative responsibility for Evergreen Woods is charged to the Executive Director, who is currently Blair Quasnitschka. Blair has over 14 years of experience in the senior living industry and has demonstrated excellence in leadership, management, sales, and member satisfaction. Most recently, he served as the Director of Operations for iCare Management. Prior to this role, he was a senior administrator for Genesis Healthcare, one of the largest skilled nursing providers in the country. Blair also serves on the National Board of Directors for the American College of Healthcare Administrators and has presented testimonial advocacy at the State's Capital five times on behalf of Connecticut's seniors.

The Health Center is managed by Jaclyn Martinelli, an administrator licensed by the State of Connecticut. Since graduating from The University of Rhode Island, with a Bachelor of Science, Jaclyn has worked in senior health care including Assisted Living and Skilled Nursing Communities. She then continued on to the University of New Haven graduating with a Master's in Public Administration, concentration in Long Term Care, and then became a licensed Nursing Home Administrator. She was fortunate enough to be hired as a newly licensed Nursing Home Administrator at The Governor's House Nursing & Rehabilitation in Simsbury, CT.

JUDICIAL PROCEEDINGS

None of the Provider, any of its officers, directors, trustees or members, or Maxwell Group has been convicted of a felony or pleaded *nolo contendere* to a felony charge, or held liable or enjoined in a civil action by final judgment involving fraud, embezzlement, fraudulent conversion or misappropriation of property, nor are any such persons subject to a currently effective injunction or restrictive or remedial order of a court of record. Within the past five years, none of the persons described herein has had any state or federal license or permit suspended or revoked as a result of an action brought by a governmental agency or department, arising out of or relating to business activity or health care, including but not limited to, actions affecting the operation of a foster care facility, nursing home, retirement home, residential care home or any facility subject to sections 17b-520 to 17b-535 of the Connecticut General Statutes, inclusive, or a similar statute in another state or country.

PART II: THE COMMUNITY

Evergreen Woods is located on an attractive 88-acre wooded site that is conveniently located on the tri-town border of North Branford, Branford, and Guilford. The address of the Community is 88 Notch Hill Road, North Branford, Connecticut 06471.

Evergreen Woods is a continuum of care community designed to accommodate persons 62 years of age or older. The Community provides active retirees a lifestyle designed to meet their unique needs while allowing them the freedom to pursue their personal interests. Continuing-care communities, such as Evergreen Woods, encompass these important components: a private apartment home, a wide array of available personal services, and the security of an on-site professional Health Center.

The Community consists of 249 apartments, numerous common spaces and a 50-bed Health Center. Common amenities include a reception lobby, a library, an indoor swimming pool with a state-of-the-art exercise room and Jacuzzi-style whirlpool spa, an auditorium with stage, an attractive dining room with an excellent view of the woods, a private dining room, which may be reserved by residents who wish to entertain a group, and several lounges designed for card playing, games, conversation or reading. Other common amenities include a country store, an arts-and-crafts room, a billiards room, a multi-media room, a beauty/barber shop, a woodworking shop and a cocktail lounge. Outdoor facilities include garden plots, hiking trails, and scenic seating areas.

The Health Center is licensed to provide chronic and convalescent nursing home care and is designed with both private and semi-private rooms. The Health Center includes a rehabilitation therapy room, recreational therapy area, dining rooms, lounges, and a beauty/barber shop. An enclosed garden courtyard with gazebo offers residents and visitors an opportunity to enjoy the outdoors. Emphasis in the Health Center is on restorative care, with the goal of returning residents to living in their apartments. The Health Center features a comprehensive, inpatient and outpatient rehabilitation program encompassing physical, occupational and speech therapy. The Health Center also provides intravenous (IV) therapy.

The Supportive Care Neighborhood is designed to meet personal care and mobility needs while promoting independence. Often referred to as a "community within a community", this resourceful neighborhood gathers services, conveniences and social interactions into a cozy and easy-to-navigate enclave. The Supportive Care Neighborhood connects the friendly and welcoming spirit of the Community, the beautiful apartments and maintenance-free services, the convenient dining and gathering features, and favorite social activities with respectful care provided in the privacy of a resident's apartment.

Evergreen Woods supports residents in maintaining a Resident Advisory Council. Elected representatives of the residents meet monthly with administration to share ideas, make suggestions and address any concerns of residents. Regularly scheduled resident meetings are held to enable the residents to ask questions and to facilitate communication between the administration of Evergreen Woods and the residents.

THE LIFE-CARE CONCEPT

The life-care concept, as provided by Evergreen Woods, offers individuals use of an apartment, support services and long-term nursing care in an on-site Health Center if they are temporarily or permanently unable to live in their apartment. This concept has grown as the result of the increasing number of men and women reaching retirement age and the concern for providing an alternative to traditional retirement living. A resident will pay an Entrance Fee which will have various minimum refunds depending upon the plan selected by the resident (see description of the plans that we offer in the section below). After assuming residence in the Community, every resident pays a monthly service fee (the "Monthly Service Fee") and a monthly working capital fund fee. A resident will be entitled to live in a private apartment, and, if necessary, may relocate to the on-site Health Center either on a temporary or permanent basis.

THE PLANS

The Community offers two types of entry fee plans to accommodate the preferences of its residents: (1) the Traditional Residency Plan, and (2) the Unbundled Residency Plan (collectively, the "Plans"). Both plans include the same services for independent living and provide for the provision of skilled nursing services. However, the Traditional Residency Plan is priced to allow for the provision of skilled nursing expenses at a much lower cost if the resident needs those services in the future. Under the Unbundled Residency Plan, the costs of skilled nursing care charged to residents are the daily costs at the time when such services are rendered, and both the Entrance Fee and Monthly Service Fee amounts are substantially less than such fees under the Traditional Residency Plan. Both Plans have 3 different prices to reflect whether the resident chooses to have his or her Entrance Fee. Each Plan is described in more detail in the applicable Residency Agreement. (See Exhibit A for an example copy of a Traditional Residency Agreement.)

CONTRACTS AND FEES

The compensation to the Provider for the on-going ownership of the Community is comprised solely from entrance fees from new residents of the Community (including any second person entrance fee charged), the Monthly Service Fee and the monthly owner's administrative fees (which are fixed for the duration of a resident's occupancy), interest earned on deposits and/or certain escrowed monies, and other benefits generally associated with the leasing and operation of real estate. In addition, to the extent the Provider loans funds to operations that would normally be funded through the Cash Requirements Allocation Formula, if applicable, and/or funds any shortfalls between operating revenues and operating expenses, the Provider may, at its option, seek to recoup such amounts through the Monthly Service Fees.

THE SERVICES

The decision to move into a continuing-care retirement community demands careful consideration of many factors, including the services to be provided. While the services provided by Evergreen Woods to residents are listed in Sections 5 and 6 of the Residency Agreement, which

governs all such obligations, the following is a description of those services most frequently asked about.

Dining Services

Each month, each individual resident receives 600 dining credit points called "Flexible Dining Dollars" (**prorated for partial months**) for use towards:

- Restaurant dining in the Community's main dining room for lunch or dinner;
- Casual dining in the bar or café for lunch or dinner;
- Private dining events in the Community's private dining rooms;
- Take-out meals;
- Catered events in common areas or in the resident's home;
- Beverage services in the Clubhouse dining and bar areas;
- Social events and parties.

A continental breakfast service is offered daily at no additional charge.

The monthly dining credit amount is sufficient for most residents to have at least one full meal (e.g., soup or salad, an entrée with sides, dessert, a beverage and a glass of wine or beer) per day each month in the main Clubhouse dining room or other dining areas. While each dining venue in the Community's Clubhouse has a standard menu, some menus also include additional daily "Wellness" offerings for residents who wish to indulge in healthier fare. The service of the dietitian is available to residents who desire nutritional counseling.

In addition to the standard menus and Wellness offerings, the Community's chefs prepare daily entrée, appetizer and dessert specials based on market availability and seasonal offerings, and the Community hosts a variety of special meals throughout the year (e.g., Thanksgiving Dinner) and these additional offerings are customized and priced accordingly.

"Take-out" service is available for lunch and dinner meals. There is no additional charge to the resident for delivery service when the resident is ill and such delivery service is approved by Evergreen Woods.

Pricing for menu items is clearly marked and, following each meal, the resident will receive a detailed receipt for that meal, the number of Flexible Dining Dollar points used for the meal and a balance of his or her Flexible Dining Dollar points remaining for the month. If a resident exceeds his or her dining credit amount in a particular month, he or she will be charged in the amount of such excess on his or her next monthly statement. If a resident ends the month with Flexible Dining Dollar points remaining, then that balance is carried over and added to the resident's dining credit balance for the next month. For example, if Mr. Smith uses only 100 of his 600 Flexible Dining Dollar points in May, then in June, Mr. Smith will begin the month with 1,100 Flexible Dining Dollar points. There is no limit to the unused Flexible Dining Dollar points that can be carried forward.

At the end of each month, any resident who has remaining Flexible Dining Dollar points can choose to either carry his or her Flexible Dining Dollar points forward in the normal course,

or to apply those unused dining credits to his or her next Monthly Service Fee in an amount equal to 30% of the Flexible Dining Dollar points credit balance. If a resident leaves the Community (or relocates to the Health Center), he or she will receive cash back in the amount of 30% of his or her remaining balance at the time of departure.

Guests are always encouraged and residents may use their Flexible Dining Dollar points to pay for guests' meals.

Residents may also use their Flexible Dining Dollar points towards adult beverages and other entertainment in the lounge and dining areas of the Community.

Flat Laundry

Evergreen Woods provides weekly service for the residents' flat laundry (i.e., sheets, pillowcases, towels, face cloths and dish cloths) which is washed, dried, folded and returned within a set time. Although the use of permanent press linen is strongly recommended, the associates will launder nonpermanent press items, but will not iron flat laundry.

Health Center

In the event a resident requires skilled nursing care, he or she may be admitted by physician's order directly to the Health Center from his or her apartment. Residents of the Community have priority access to the Health Center over nonresident patients desiring admission. Flexible Dining Dollar points may not be used in the Health Center; instead the resident's monthly billing statements will be adjusted when he or she is admitted to the Health Center to include the costs of Health Center meals. Visitors and volunteers are encouraged to visit the patient. Friends, relatives or spouses may take meals with patients with advance notification.

The costs of skilled nursing care are charged to residents who have the Unbundled Residency Plan at the then published daily rate for skilled nursing plus fees for supplies, medication, and any rehab provided.

A resident who has the Traditional Residency Plan will pay for skilled nursing at what he or she otherwise had been paying for his or her apartment Monthly Service Fee plus fees for the extra meals provided and for supplies, medication, and any rehab provided.

Evergreen Woods is licensed to provide rehabilitative, skilled nursing care. Evergreen Woods is not licensed to provide hospital-level care or certain other services including, but not limited to, acute psychiatric care and ventilator therapy. The Monthly Service Fee does not cover, and Evergreen Woods will not be responsible for, the costs of any hospitalization required by residents.

A physician has been retained on a consulting basis to act as medical director for the Health Center. Residents in the Health Center must be seen and followed by a physician who is credentialed by Evergreen Woods and must comply with all state and federal requirements for licensed health care centers. While in the Health Center, the resident will be provided nursing care in a semi-private room. If a resident desires an available private room, the resident may obtain one, subject to availability, and upon agreement to pay the difference in cost between semi-private and private rooms, calculated based on private-pay rates. Semi-private Health Center care includes basic nursing care and emergency medical care when the medical director is summoned by Health Center associates. The resident is responsible for the services of his/her personal physician. In the event the resident's personal physician or the Community's medical director orders medication, therapy, or various supplemental services for the resident's care, the resident shall be responsible for the cost of such services and medication.

Refer to Section 7 of the Residency Agreement for more details on charges in the Health Center.

Maintenance of the Living Unit

Evergreen Woods maintains all common spaces and provides weekly housekeeping services for resident apartments. Housekeeping services include: cleaning and dusting the interior of the apartment and vacuuming, all on a weekly basis; washing and waxing hard surface floors; and the annual cleaning of ovens and windows. Carpets will be cleaned approximately every 24 months, and, at other times, spot cleaning will be done. Associates will assist residents in arranging and moving furniture as needed during the first 3 months of occupancy at no charge. Extra cleaning help will be available at additional cost.

Other Charges

Other services may be provided to residents at additional charge, which may be added to the monthly bill. Such services include guest accommodations, newspapers, personal transportation, guest meals not covered by the resident's Flex Dining Dollars point balance, nursing and personal care services provided by the Resident Services Department, beauty/barber shop services, additional resident meals (available at a discount if taken on a monthly plan), additional housekeeping, personal laundry service and such other reasonable services as requested.

Pets

In recognition of the strong role that pets play in our lives, certain pets are permitted, accordance with our Pet Policy. Please see your Lifestyle Advisor for a current copy of the policy.

Supportive Services for Apartment Residents

Those residents who do not require care in the Health Center, but who need additional personal care services to continue living in their apartments may receive the services of the Resident Services Department staffed by a professional counselor, registered and licensed nurses and certified nursing assistants. The Resident Services Department provides services such as Wellness Clinic, nursing treatments, medication management, case management, bathing, dressing, additional housekeeping, shopping, escort and laundry services to the residents at additional cost. Such services are made available to the residents at an extra charge. Evergreen Woods reserves the right to add to, delete from, or modify such additional services and charges from time to time.

Evergreen Woods is a licensed Assisted Living Services Agency (ALSA) and registered Managed Residential Community (MRC) with the Connecticut Department of Public Health. The Resident Services Department offers nursing and personal care services to apartment residents who qualify under Connecticut law. Residents are free to contract with other Connecticut-licensed home health care agencies, so long as such agencies meet certain criteria and their staff meets the same requirements of the Evergreen Woods staff (i.e., background screens, drug testing, documentation, and community specific training).

Telephone Service

Each apartment has "live" jacks in various locations. Actual telephone service is at the resident's expense.

Transportation

Scheduled bus and other transportation services are provided. Transportation to theater, outings, museums and other special events is offered and may be at extra cost to the resident. Regularly scheduled bus transportation is provided to shopping centers, medical offices and local religious services. Private transportation services are also available at an additional charge.

THE RESIDENCY AGREEMENT

Evergreen Woods' Residency Agreement (the "Residency Agreement") sets out the contractual agreements between the residents and the Community. See Exhibit A for an example copy of a Traditional Residency Agreement and Exhibit B for an example copy of an Unbundled Residency Agreement. There are three versions of each type of Residency Agreement (Traditional and Unbundled); the only substantive difference between the three versions is the refund provisions, which are described under "PROMISSORY NOTE" on page 33. Evergreen Woods reserves the right to offer to new prospective residents alternative forms of residency agreements from time to time.

Upon deciding to become a resident of the Community, a future resident will execute a Residency Agreement to reserve the apartment selected. Below is a reference of key terms which are described in the Residency Agreement.

Rights of a Surviving Spouse (or Second Persons): Section 4.5 of the Residency Agreement describes the rights of a surviving spouse (or second person).

Marriage of a Resident: Section 11.2 of the Residency Agreement describes your rights if you get married or decide to have a second person live with you.

Disposition of Personal Property: Section 11.4 of the Residency Agreement describes the rights of the resident and the Community regarding the disposition of personal property upon death or cancellation or termination of the Residency Agreement.

Evergreen Woods' Termination Rights: Section 10.1 of the Residency Agreement indicates that Evergreen Woods will not terminate the Residency Agreement except for just cause.

Limitations on Termination Rights for Financial Inability: Section 10.2 of the Residency Agreement explains limitations on termination rights for financial inability.

Notice of Termination: Section 10.3 of the Residency Agreement explains the notice of termination.

Emergency Termination: Section 10.4 of the Residency Agreement explains emergency termination.

Resident's Termination Rights: Section 9 of the Residency Agreement explains the resident's termination rights.

TAX CONSEQUENCES

Payment of an entrance fee or other transfer of assets pursuant to a continuing care contract may have significant tax consequences. You may wish to consult a qualified adviser before signing the Residency Agreement to discuss any potential tax consequences.

MONTHLY SERVICE FEES

Traditional Residency Plans							
U	NIT TYPE	2016	2017	2018	2019	2020	2021
ONE BEDROOM A:	The Maple	\$3,379	\$3,585	\$3,720	\$3,880	\$3,990	\$4,170
ONE BEDROOM B:	The Birch	\$4,033	\$4,265	\$4,420	\$4,610	\$4,750	\$4,960
ONE BEDROOM C:	The Hickory	\$4,130	\$4,365	\$4,530	\$4,730	\$4,870	\$5,080
ONE BEDROOM D:	The Cedar & The Oak	\$4,600	\$4,885	\$5,040	\$5,260	\$5,420	\$5,660
ONE BEDROOM D*:	The Elm	\$4,600	\$4,885	\$5,040	\$5,260	\$5,420	\$5,660
TWO BEDROOM E:	The Dogwood & The Cherry	\$4,785	\$5,045	\$5,230	\$5,460	\$5,620	\$5,870
TWO BEDROOM F:	The Willow & The Ash	\$4,882	\$5,145	\$5,340	\$5,570	\$5,740	\$5,990
TWO BEDROOM FC:	The Holly	\$4,882	\$5,145	\$5,340	\$5,570	\$5,740	\$5,990
TWO BEDROOM G:	The Sweetgum	\$5,159	\$5,435	\$5,640	\$5,890	\$6,060	\$6,330
TWO BEDROOM G*:	The Magnolia	\$5,159	\$5,435	\$5,640	\$5,890	\$6,060	\$6,330
TWO BEDROOM GL:	The Poplar	\$5,159	\$5,435	\$5,640	\$5,890	\$6,060	\$6,330
TWO BEDROOM GL*:	The Sycamore	\$5,159	\$5,435	\$5,640	\$5,890	\$6,060	\$6,330
SECOND PERSON		\$2,159	\$2,242	\$2,300	\$2,400	\$2,430	\$2,540

Unbundled Residency Plans

τ	JNIT TYPE	2019	2020	2021
ONE BEDROOM A:	The Maple	\$2,780	\$2,890	\$3,070
ONE BEDROOM B:	The Birch	\$3,510	\$3,650	\$3,860
ONE BEDROOM C:	The Hickory	\$3,630	\$3,770	\$3,980
ONE BEDROOM D:	The Cedar & The Oak	\$4,160	\$4,320	\$4,560
ONE BEDROOM D*:	The Elm	\$4,160	\$4,320	\$4,560
TWO BEDROOM E:	The Dogwood & The Cherry	\$4,360	\$4,520	\$4,770
TWO BEDROOM F:	The Willow & The Ash	\$4,470	\$4,640	\$4,890
TWO BEDROOM FC:	The Holly	\$4,470	\$4,640	\$4,890
TWO BEDROOM G:	The Sweetgum	\$4,790	\$4,960	\$5,230
TWO BEDROOM G*:	The Magnolia	\$4,790	\$4,960	\$5,230
TWO BEDROOM GL:	The Poplar	\$4,790	\$4,960	\$5,230
TWO BEDROOM GL*:	The Sycamore	\$4,790	\$4,960	\$5,230
SECOND PERSON		\$1,180	\$1,210	\$1,210

(Unbundled Residency Plans were not offered until 2019)

ENTRANCE FEES

Traditional Residency Plan¹

Unit Type		2016	2017	2018	2019
ONE BEDROOM A:	The Maple	\$210,000	\$210,000	\$210,000	\$210,000
ONE BEDROOM B:	The Birch	\$301,000	\$301,000	\$301,000	\$301,000
ONE BEDROOM C:	The Hickory	\$305,000	\$305,000	\$305,000	\$305,000
ONE BEDROOM D:	The Cedar & The Oak	\$338,000	\$338,000	\$338,000	\$338,000
ONE BEDROOM D*:	The Elm	\$352,000	\$352,000	\$352,000	\$352,000
TWO BEDROOM E: Cherry	The Dogwood & The	\$364,235	\$364,235	\$364,235	\$364,235
TWO BEDROOM F: Ash	The Willow & The	\$374,253	\$374,235	\$374,235	\$374,235
TWO BEDROOM FC:	The Holly	\$408,435	\$408,435	\$408,435	\$408,435
TWO BEDROOM G:	The Sweetgum	\$478,930	\$478,930	\$478,930	\$478,930
TWO BEDROOM G*:	The Magnolia	\$493,000	\$493,000	\$493,000	\$493,000
TWO BEDROOM GL:	The Poplar	\$497,765	\$497,765	\$497,765	\$497,765
TWO BEDROOM GL*:	The Sycamore	\$513,000	\$513,000	\$513,000	\$513,000
SECOND PERSON					\$22,000

Note: The above prices reflect a 70% refund Traditional Residency Plan which is no longer offered.

¹ A one-time capital charge is due at the time the resident pays his/her balance. This non-refundable fee will be used for apartment refurbishment. For the Maple, Birch and Hickory styles, the capital charge is \$3,500. For the Cedar, Oak Elm, Dogwood, Cherry, Willow, Ash and Holly styles, the capital charge is \$6,500. For the Sweetgum, Magnolia, Poplar and Sycamore styles and larger, the capital charge is \$9,000.

ENTRANCE FEES

Tailored Entrance Fee Plan²

Unit Type		2016	2017	2018	2019
ONE BEDROOM A:	The Maple	\$144,000	\$144,000	\$144,000	\$144,000
ONE BEDROOM B:	The Birch	\$226,000	\$226,000	\$226,000	\$226,000
ONE BEDROOM C:	The Hickory	\$229,000	\$229,000	\$229,000	\$229,000
ONE BEDROOM D: The Oak	The Cedar &	\$253,000	\$253,000	\$253,000	\$253,000
ONE BEDROOM D*:	The Elm	\$266,000	\$266,000	\$266,000	\$266,000
TWO BEDROOM E: & The Cherry	The Dogwood	\$273,160	\$273,160	\$273,160	\$273,160
TWO BEDROOM F: The Ash	The Willow &	\$283,160	\$283,160	\$283,160	\$283,160
TWO BEDROOM FC:	The Holly	\$305,980	\$305,980	\$305,980	\$305,980
TWO BEDROOM G: Sweetgum	The	\$359,250	\$359,250	\$359,250	\$359,250
TWO BEDROOM G*:	The Magnolia	\$372,000	\$372,000	\$372,000	\$372,000
TWO BEDROOM GL:	The Poplar	\$373,350	\$373,350	\$373,350	\$373,350
TWO BEDROOM GL*:	The Sycamore	\$386,000	\$386,000	\$386,000	\$386,000
SECOND PERSON					\$22,000

Note: The above prices reflect a 0% refund Tailored Residency Plan which is no longer offered.

 $^{^{2}}$ A one-time capital charge is due at the time the resident pays his/her balance. This non-refundable fee will be used for apartment refurbishment. For the Maple, Birch and Hickory styles, the capital charge is \$3,500. For the Cedar, Oak Elm, Dogwood, Cherry, Willow, Ash and Holly styles, the capital charge is \$6,500. For the Sweetgum, Magnolia, Poplar and Sycamore styles and larger, the capital charge is \$9,000.

2020 & 2021 ENTRANCE FEES³

		UNBUNDLED PLAN			TRA	ADITIONAL PLANS	
Unit Type		Endowment (0%) Plan	60% Plan	90% Plan	Endowment (0%) Plan	60% Plan	90% Plan
ONE BEDROOM A: ONE BEDROOM B:	The Maple The Birch	\$133,500 \$190,600	\$182,100 \$259,900	\$242,800 \$346,500	 \$157,100 \$224,200	\$214,200 \$305,800	\$285,600 \$407,700
ONE BEDROOM B:	The Hickory	\$190,000	\$259,900	\$352,900	\$228,400	\$311,400	\$407,700
ONE BEDROOM D:	The Cedar & The Oak	\$214,500	\$292,500	\$390,000	\$252,300	\$344,100	\$458,800
ONE BEDROOM D*:	The Elm	\$223,700	\$305,000	\$406,700	\$263,100	\$358,800	\$478,400
TWO BEDROOM E:	The Dogwood & The Cherry	\$231,700	\$316,000	\$421,300	\$272,600	\$371,800	\$495,700
TWO BEDROOM F:	The Willow & The Ash	\$238,000	\$324,500	\$432,700	\$280,000	\$381,800	\$509,100
TWO BEDROOM FC:	The Holly	\$259,900	\$354,400	\$472,500	\$305,700	\$416,900	\$555,900
TWO BEDROOM G:	The Sweetgum	\$304,800	\$415,600	\$554,100	\$358,500	\$488,900	\$651,900
TWO BEDROOM G*:	The Magnolia	\$313,300	\$427,300	\$569,700	\$368,700	\$502,700	\$670,300
TWO BEDROOM GL:	The Poplar	\$316,600	\$431,800	\$575,700	\$372,500	\$508,000	\$677,300
TWO BEDROOM GL*:	The Sycamore	\$326,200	\$444,800	\$593,100	\$383,700	\$523,300	\$697,700
SECOND PERSON		\$0	\$0	\$0	\$22,000	\$22,000	\$22,000

³ A one-time capital charge is due at the time the resident pays his/her balance. This non-refundable fee will be used for apartment refurbishment. For the Maple, Birch and Hickory styles, the capital charge is \$3,500. For the Cedar, Oak Elm, Dogwood, Cherry, Willow, Ash and Holly styles, the capital charge is \$6,500. For the Sweetgum, Magnolia, Poplar and Sycamore styles and larger, the capital charge is \$9,000.

SCHEDULE OF MISCELLANEOUS CHARGES

(Effective 1/1/2021)

Please note: All ancillary charges are billed from the 1st through the last day of the month.

GUEST SERVICES	CHARGE
Guest Rooms 1, 2, & 4	\$100.00 per night
• If reserved for 7 or more consecutive nights	\$80.00 per night
Guest Room 3	\$85.00 per night
• If reserved for 7 or more consecutive nights	\$68.00 per night
Cot Rental – tax included	
• Up to 7 days	\$12.00 per night
• After 1st 7 days	\$5.00 per night
Memorial Services Room Setup and Cleaning Fees are as follows	
Auditorium	\$250.00
Reis Room	\$250.00
Pavilion	\$150.00
Room Capacity:	
Auditorium	80 people
Reis Room	50 people
Pavilion	30 people
MEALS	CHARGE
Guest Meals – tax included	See prices posted
E-2 and HC guest meals	See prices posted
Meal Delivery Charges	\$5.00
Additional 2 meals for Health Center Days	\$31.20 per day (x 2 meals)
Catering Services	Individually priced

OFFICE SERVICES – All office services will be "cashless" services. All transactions will be added to your monthly bill.	CHARGE
Copier	\$0.10/copy
Postage	As metered
Laminating	\$2.00 (8-1/2 x 11)
FAX (outgoing only)	\$1.00 per page

MAINTENANCE SERVICES

Services for moving furniture, hanging pictures, etc. will be provided at no charge for a period of ninety days from your move-in date.

CHARGE

After 90 days \$40.00 per hour (min 1 hour)

After work which may require the use of an outside contractor such as an electrician will be handled in one of two ways:

- 1. The contractor will invoice you directly.
- 2. Evergreen Woods will be invoiced and the amount of the invoice plus any applicable fees will be billed to you by Evergreen Woods.

Replacement of Broken Keys	\$5.00
Replacement of Lost Keys	\$30.00
Replacement of Lost Fob	\$15.00

Replacement of Mailroom Keys will be charged based on the locksmith fee at time of replacement.

OTHER MISCELLANEOUS SERVICES	CHARGE
Garage	\$30.00/Month – (per \$5,000
	purchased garage agreement)
	\$120/Month – (per *no money
	down rental garage agreement)
Massage Therapy	See prices posted in Massage
	Room

TRANSPORTATION SERVICES		
Personal Transportation for Evergreen Woods Residents:		
1st 25 miles or 30-minute radius (round trip) no charge	, after 25 miles or 30 minutes, \$0.55 per mile	
Car	No Hourly Charge	
Wheelchair Van	No Hourly Charge	
* All New Haven locations covered.		
Private Transportation for Non-Residents (private aides, family members, etc.):		
1st 20 miles or 30-minute radius no charge, after 20 miles or 30 minutes, \$0.55 per mile		
Car (1-hour minimum)	\$22 per hour (1/2 hr. increments)	
Wheelchair Van (1-hour minimum)	\$30 per hour (1/2 hr. increments)	
* After Hours Private Transportation:		
Car (1-hour minimum)	\$30 per hour (1/2 hr. increments)	
	\$40 per hour (1/2 hr. increments)	

transportation occurring outside of the stated times are liable to be charged for driver hours and mileage. * Subject to driver availability.

ENVIRONMENTAL SERVICES	CHARGE
Bed-making	\$6.00 per bed
Carpet cleaning complete	\$24.00
Window washing	\$24.00
Re-organizing	\$24.00
Oven cleaning	\$24.00
Blinds (other than during deep cleaning)	\$24.00
Silver polishing	\$24.00
Moving area rugs and furniture	\$24.00
Personal clothing laundry service *wash & wear items only	\$1.25 per pound / 10 pound minimum
Special laundry processing	\$24.00
Carpet cleaning (spots)	\$12.00
Mattress turning	\$12.00
Refrigerator cleaning	\$12.00
Hand polishing furniture	\$12.00
Watering plants when away	\$12.00
Any additional services	per event

RESIDENT HEALTH SERVICES PROGRAM FEES (Effective 1/1/2021)

Please note: All ancillaries are billed from the 1st through the last day of the month.

RESIDENT HEALTH SERVICES provided at no additional charge for those on Services

- Wellness Counseling, Health Promotion, Disease Prevention
 - ✓ Resident Liaison /Advocate
 - ✓ Support / Counseling
 - ✓ Health education lectures
 - ✓ Education on treatments and procedures
 - ✓ Coordination of medical equipment
 - ✓ Follow-up on emergency calls
- Transportation to / from the Health Center
- Arrange for take-out meal delivery due to brief illness
- Medication delivery if needed due to illness
- Wellness Clinic Visit (nursing treatment may be at additional charge)
 - \checkmark Blood pressure checks
 - ✓ Instruction / Teaching for treatments such as eye drops, support hose, heat or cold application, etc.
- Referrals to other agency services as appropriate such as Physical Therapy

NURSING SERVICES

Nursing Services (performed by licensed nurses)

Coordination of care for residents receiving Health Services	\$55.00 / month
Admission Assessment to Assisted Living Services	\$150.00 / per admission
 Nursing Treatments /Assessments Including but not limited to: Pacemaker check Wound care In-home blood pressure checks Physical assessments In-home Assessment to follow-up on Call for Assistance Medication Re-Adjustment 120-Day Required Re-assessment 	\$30.00 per 15 minutes
 Medication pre-pours \$90.00 / pre-pour Includes ordering medications, assessing for side effects of medications, performing physical assessment (including blood pressure check and other vital signs). Nursing Treatments covered by Medicare will be referred to the Visiting Nurse Agency (VNA) of your choice. 	

RESIDENT ASSISTANT SERVICES (performed by certified aides): \$9.50 per 15 minutes		
All services are <u>SCHEDULED</u> , not on-demand.		
 Medication reminders or supervision Skin Care (non-medicated creams only) Toileting Exercise Bathing / hygiene Dressing Assistance with Support Hose Meal Preparation Grocery Shopping Errands / Escorts 	 Personal Laundry Housekeeping Ironing Unpacking Respite care "Friendly Visits" Pet Assistance (food and water, cat litter removal only – no dog walking services) Safety Checks Reading and non-financial correspondence 	

24 HOUR NOTICE MUST BE GIVEN TO CANCEL SCHEDULED CARE IN ORDER TO AVOID FULL CHARGE.

Residents who want personal laundry as their only service will be serviced by our Laundry Department, not the Resident Services Department.

Services provided on Evergreen Woods Holidays will be billed at a 50% premium to reflect the cost of staff on these days (\$14.25 per unit of service).

The Resident Services Department holds a FREE, non-emergency clinic 7 days / week in the Wellness Center from 1:00-2:00pm for sick visits, blood pressure checks, and simple treatments not accepted by the VNA.

Calls for Assistance to your apartment that are non-emergency and therefore do not result in transport to the hospital will be charged at 35.00 / 15 minutes.

For those residents that have Long Term Care Insurance: Submission of bills and other required paperwork for reimbursement are the responsibility of the resident and / or family, not Resident Services / Evergreen Woods.

MISCELLANEOUS	CHARGE
Cordmate Rental	\$50.00 per month
Emergency Linen Changes	\$25.00 (flat fee)
Dining Room Escort (round-trip)	\$18.00 (flat fee)
Dining Room Escort (one-way)	\$9.50 (flat fee)
Quiet Care Monitoring System	\$70.00 per month

SUPPORTIVE CARE ALL-INCLUSIVE CHARGES	
Supportive Care Program on 2 nd Floor of Eaton	\$1,850 per month
Quiet Care Monitoring System	\$70.00 per month

Dining Services:

- ✓ A Dining Room available right on the floor offering continental breakfast and a nutritionally wellbalanced lunch and dinner.
- ✓ Staff available at each meal to provide assistance and monitoring of dietary intake.

Activity Program:

✓ Program staff will provide stimulating recreational activities and an opportunity for socialization and interaction.

Nursing/Coordination of Care Services:

- ✓ Up to one (1) hour of scheduled incremental <u>certified aide assistance</u> with activities of daily living such as dressing, bathing etc. and supervision of self-administration of medication as necessary. (Aide services provided above and beyond the one hour of care will be provided at an additional fee)
- ✓ Certified nursing assistant on 2nd Floor of Eaton 24 hours a day
- ✓ Dietician consultant when necessary
- ✓ Health monitoring and promotion, wellness counseling and disease prevention in apartment
- ✓ Periodic nursing assessments as necessary
- ✓ Coordination with your personal physician and referrals to other health care professionals, agencies or other ancillary services as appropriate
- ✓ Follow up visits after hospitalization or other medical absences
- ✓ Coordination of care between the resident or family member in determining the appropriate level of services along the continuum of care
- ✓ Coordination of medical transportation
- ✓ Associate response to emergencies
- \checkmark Security checks during the night with medical staff response if needed.

HEALTH CENTER SCHEDULE OF FEES (Effective 1/1/2021)

Please note: All ancillaries are billed from the 1st through the last day of the month.

HEALTH CENTER ROOM RATES FOR PRIVATE PAY	CHARGE
Semi-Private	\$466.00 per day (members) \$476.00 per day (non-members)
Private	\$526.00 per day (members) \$536.00 per day (non-members)

PRIVATE ROOM DIFFERENTIAL	CHARGE
Life Care Resident	\$125.00 per day (if not medically necessary)
Medicare Resident	\$60.00 per day (if not medically necessary)

MEDICARE PART A COVERED STAY

Days 1-20 Traditional Medicare covers 100% of your Medicare Part A stay

Days 21-100 Traditional Medicare covers all costs except a co-pay of \$185.50 per day

(Secondary insurance supplemental to Medicare may cover this cost.)

50 SKILLED NURSING BEDS

Room and Board, Social Services, Therapeutic Recreation, Linen, Housekeeping, Maintenance Services and 24-hour Skilled Nursing Services.

al cost (most items covered by insurance).
 Nutritional Supplements Personal Laundry Services Personal Supplies Podiatry Private & Semi Private Rooms Private Telephone Psychiatry Services Religious Services Special Events & Entertainment Wander Alert System Wheelchairs & Recliner Chairs

BEAUTY SALON SERVICE FEES (Effective 1/1/2021)

Please note: All ancillaries are billed from the 1st through the last day of the month.

BEAUTY SALON SERVICE FEES	CHARGE
Shampoo Only	\$11.00
Ladies Haircut	\$31.00
Haircut and Shampoo	\$33.00
Haircut and Blow Dry or Set	\$48.00
Men's Haircut	\$22.00
Shampoo / Diffuser / Set	\$26.00
Shampoo / Blow-Dry / Iron	\$26.00
Permanent including Cut / Shampoo / Conditioner / Style	\$94.00
Color with Shampoo / Set / Blow Dry	\$61.00
Waxing of Brow, Lip, Face	\$11.00
Beard Trim	\$11.00
Comb Out	\$11.00
Nail Cut and File	\$11.00
Manicure including Nail Polish	\$20.00
Nail Polish Change	\$11.00

PART III: FINANCIAL INFORMATION

ENTRANCE FEE ESCROW

Under the provisions of the Act, the Provider must maintain an escrow account for certain portions of the entrance fees with a bank or trust company. These are the entrance deposits that are paid to the Provider upon execution of the Residency Agreement and prior to occupancy, referred to as the "Entrance Deposit." Such account has been established with U.S. Bank (formerly Union Trust Company), One Federal Street, Boston, Massachusetts 02110. Attached as <u>Exhibit</u> <u>C</u> to this Agreement is a sworn statement of U.S. Bank, affirming that the escrow required by Section 17b-524 of the Connecticut General Statutes has been established and maintained.

The ten percent (10%) Entrance Deposit, which prospective residents pay to reserve the apartment, is deposited into the entrance fee escrow account. The funds and interest earnings are held in escrow until all statutory requirements are met.

Prospective residents who desire to cancel their Residency Agreement are required to send a written notice of termination by registered or certified mail to the escrow agent at the above address and to Evergreen Woods Retirement, LLC, 88 Notch Hill Road, North Branford, Connecticut 06471. Prospective residents who cancel will not receive interest on the entrance fee deposit.

Interest earned on the entrance fee escrow account shall belong to the Provider.

OPERATING RESERVE ESCROW AND DEBT SERVICE RESERVE ESCROW

The Provider is required to and has established and maintains in trust an escrow account with KeyBank National Association, located at 100 Pearl Street, Hartford, Connecticut 06103 on a current basis, an operating and debt service reserve in an amount sufficient to cover: (a) all principal and interest, rental or lease payments due during the next six months on account of any first mortgage loan or other long-term financing or such lesser amount as determined by the Department of Social Services; and (b) the total cost of operation of the Community for a onemonth period, excluding debt service and certain capital expenditures. Attached as **Exhibit D** to this Agreement is a sworn statement of KeyBank National Association, affirming that the escrow required by Section 17b-525 of the Connecticut General Statutes have been established and maintained. KeyBank National Association, as Escrow Agent, may invest all amounts held by it from time to time in such obligations of any federal or state or local authority (or in shares of a registered investment company which holds a portfolio of such securities) or such certificates of deposit, savings accounts, money market accounts, deposits, or other savings or investment securities including, without limitation, those of Escrow Agent, as the CAO of SLC may from time to time direct in writing, or absent such direction, in such accounts or similar investments as the Escrow Agent shall deem advisable in its sole discretion. Interest earned on the debt service and operating reserve escrow account is payable to the Provider. The balance of Evergreen Woods operating and debt service reserve escrow account as of January 28, 2021 was \$4,062,838.63.

PROMISSORY NOTE

Under the life-care retirement concept, no ownership interest in the real estate is transferred to the resident. Rather, the resident is provided use of an apartment, subject to and under the terms of the Residency Agreement and Connecticut statutes.

With all Residency Agreements, whether a Traditional Residency Agreement or an Unbundled Residency Agreement, the resident has a choice of three different refund plans: (1) the ninety percent (90%) refund plan, (2) the sixty percent (60%) minimum refund plan, or (3) the Endowment 0% refund plan. Between zero percent (0%) and ninety percent (90%) of the Entrance Fee is refunded to the resident or his or her estate upon termination of residency in the Community depending upon the type of plan selected by the resident and the date of termination. The 90% refund plan has a higher Entrance Fee than the 60% refund plan which has a higher Entrance Fee than the Endowment 0% refund plan.

If a resident chooses the ninety percent (90%) refund plan, then ninety percent (90%) of the Entrance Fee is refundable to the resident or his or her estate upon termination of residency in the Community no matter how long he or she has lived at the community.

If a resident chooses the sixty percent (60%) minimum refund plan, then a minimum of sixty percent (60%) of the Entrance Fee is refundable to the resident or his or her estate upon termination of residency in the Community. Ninety percent (90%) is refundable if the resident ends his or her residency in the Community during the 1^{st} year after moving in. The refund amount declines to eighty percent (80%) during the second year and further declines to seventy percent (70%) during the third year. In the fourth year and all subsequent years, the amount refundable to the resident or his or her estate upon termination of residency in the Community is always sixty percent (60%).

If a resident chooses the Endowment 0% refund plan, then ninety percent (90%) of the Entrance Fee is refundable initially and during the first 5 months after occupancy, but is then amortized over the following forty-five (45) months to zero refund.

The refundable portion of the Entrance Fee for all plans is evidenced by a Promissory Note, an example of which is attached as an exhibit to the Residency Agreement executed by the resident and the Provider at occupancy.

FINANCING OF THE COMMUNITY

The Community is financed through a long-term lease from NHI-REIT of Evergreen, LLC, a wholly-owned subsidiary of National Health Investors, Inc., a Maryland corporation that is a real estate investment trust. The initial term of the lease runs to December 31, 2029, subject to two options to extend the initial term for an additional five years each. As of January 1, 2021 the annual lease payment for the Community is \$5,921,400, which will increase by 3% on January 1 each year. One half of the annual lease payment is required to be deposited in the operating and debt service reserve escrow described above.

AUDITED FINANCIAL STATEMENTS

Evergreen Woods Retirement, LLC was formed in 2016 and began operating the Community in November 2016. Audited financial statements of the Provider for years ended December 31, 2018 and December 31, 2019 are attached as Exhibit E.

Due to specific accounting rules applicable to continuing care retirement communities, specifically the treatment of the nonrefundable portion of entrance fees, our financial statements

can often be misleading to the average individual. While many individuals tend to focus primarily on the income statement, the most useful resource is going to be the statement of cash flows.

FINANCIAL PROJECTIONS

It is important for future residents, their families and their advisors to understand the financial basis on which the Community operates. The past experience of developing and managing senior living communities has been the basis for financial planning for the Community. As discussed in the Introduction to this Disclosure Statement, the achievement of a financial forecast is always dependent on future events.

THE PRO FORMA CASH FLOW STATEMENT

A budgeting system is an essential business tool. Naturally, the entire process is ongoing, as new developments allow for refinement. Such new developments may include external events over which there is no control.

The Pro Forma Cash Flow Statement is simply a projection, as of January 2021, of the estimated flow of cash into and out of the Community for the period beginning on January 1, 2021 and ending December 31, 2023. Assumptions and explanations are contained in the notes following the Pro Forma Cash Flow Statement.

The Pro Forma Cash Flow Statement is based on operations of Evergreen Woods as of January 2021. Such estimates are, of course, dependent upon assumptions concerning many items, including economic conditions such as inflation and interest rates, which are subject to change. Although there can be no guarantee of accuracy, it is believed that the Pro Forma Cash Flow Statement reflects sound financial planning and a rational set of assumptions, based upon experience and insight.

PRO FORMA CASH FLOW ASSUMPTIONS

- (a) Evergreen Woods maintains its operating cash under a sweep arrangement with a master account held by SLC. For purposes of the Pro Forma Cash Flow below, beginning cash was set at \$0 with ending cash an accumulation of projected net change in cash per the projection.
- (b) Apartment Service Fees represent the total of Monthly Service Fees for all apartments based on occupancy of 70%, increasing by 3%.
- (c) Total census in the Health Center is assumed to be 43 of 50 beds; 31 Pvt Pays and 12 Med A.
- (d) Other income represents projected income from items such as assisted living services, meals, garage rentals, guest rooms, beauty and barbershop, and application fees.
- (e) Interest income is based on estimated projections of future interest rates.
- (f) A 3% increase in operating expenses is projected.

	2021	2022	2023	
Beginning Cash	\$0	\$44,743	\$488,920	(a)
Additions:				
Entrance Fee Collections net of Refunds	5,094,300	5,538,900	5,649,700	
Monthly Service Fees	12,042,199	13,030,617	14,177,190	(b)
Net Health Center Income	853,653	1,425,298	1,468,057	(c)
Other Income	1,045,698	1,134,870	1,206,578	(d)
Interest Income	16,200	16,300	16,400	(e)
Disbursements:				
Operating Expenses	11,812,350	13,291,002	14,069,800	(f)
Capital Expenditures	1,128,000	1,161,840	1,196,695	(g)
Lease Payments	6,066,957	6,248,966	6,436,435	
Net Change	44,743	444,177	814,994	
Ending Cash available for reserves and/or distribution	\$44,743	\$488,920	\$1,303,914	

(g) Capital expenditures cover unit and common area improvements or refurbishments and building updates.

Comparison of prior projections to actual results: In the Provider's prior Disclosure Statement, (a) the projection for 2019 Entrance Fee Collections net of Refunds was \$3,200,000 and the audited result was \$403,228; (b) the projection for 2019 Monthly Service Fees was \$11,708,678 and the audited result was \$11,350,068; (c) the projection for 2019 Net Health Center Income was \$1,200,266 and the audited result was \$1,431,154; (d) the projection for 2019 Other Income was \$945,406 and the audited result was \$1,022,662; (e) the projection for 2019 Interest Income was \$32,000 and the audited result was \$76,437; (f) the projection for 2019 Operating Expenses was \$11,603,260 and the audited result was \$11,527,916; (g) the projection for 2019 Capital Expenditures was \$300,000 and the audited result was \$940,422; (h) the projection for 2019 Lease Payments was \$5,418,920 and the audited result was \$5,704,319.

In addition to the foregoing assumptions, the Pro Forma Cash Flow Statement is based the following assumptions:

- 1. The current rate schedule (see pages 18-21).
- 2. The independent living unit residential turnover rates anticipated for the next three fiscal years, are as follows:

Number of Residents Beginning of Year

2021	<u>2022</u>	<u>2023</u>
188	211	211

Anticipated Residential Turnover

<u>2021</u>	<u>2022</u>	<u>2023</u>
5.3%	9.6%	17.5%

Number of Occupied Units Beginning of Year

<u>2021</u>	2022	<u>2023</u>
149	167	167

Occupied Unit Turnover

<u>2021</u>	<u>2022</u>	<u>2023</u>
2.5%	8.7%	19.2%

3. The projected average age for the next three fiscal years for independent resident is as follows:

<u>2021</u>	<u>2022</u>	<u>2023</u>
88	87	86

4. Health care utilization rates, including admission rates and days per 100 residents by level of care anticipated for the next three fiscal years, are as follows:

Assisted Living Units

Year		tion Rate s/Percent	-	<u>sion Rate</u> t/Percent	<u>Days per 100</u> <u>Residents*</u>
2021	13.5	4.5%	4.0	2.1%	1,648
2022	14.0	4.7%	4.0	1.9%	1,709
2023	14.0	4.7%	4.0	1.9%	1,709

Skilled Nursing Facility

Year		tion Rate ts/Percent		ion Rate /Percent	<u>Days per 100</u> <u>Residents*</u>
2021	35.8	12.0%	120.0	63.8%	4,370
2022	43.0	14.4%	96.0	45.5%	5,429
2023	43.0	14.4%	97.0	46.0%	5,429

*Only one level of care – skilled nursing.

**Utilization rates include only Life-Care Residents.

5. The average occupancy rates for independent living units anticipated for the next three fiscal years, are as follows:

2021	2022	2023
67.8%	71.4%	76.1%

6. The average number of days of care by level of care anticipated for the next three fiscal years, is as follows:

Year	Assisted Living	Skilled Nursing
2021	4,928	13,067
2022	5,110	15,695
2023	5,110	15,695

*Includes life care permanent and temporary residents only. No direct admit residents are included in these numbers.

7. The number of health care admissions by level of care anticipated for the next three fiscal years, is as follows:

Year	Assisted Living	Skilled Nursing
2021	4.0	120.0
2022	4.0	96.0
2023	4.0	97.0

8. The number of permanent transfers to the assisted living units and skilled nursing facility for the most recently completed fiscal year are:

Transferring From:

Facility Transferred to:	Independent Living	Assisted Living	Total
Assisted Living	4		4
Skilled Nursing	5	1	6

9. The current occupancy rate as of Dec. 31, 2020 was:

ТҮРЕ	% Occupied
Assisted Living	87%

Skilled Nursing	46%
Independent Living	63%

PART IV: STATUTORY ACKNOWLEDGEMENTS

REGISTRATION

The Community is subject to the provisions of the Connecticut Continuing Care Facilities Act ("Act"), Conn. Gen. Stat. § 17b-520 et seq. Accordingly, Evergreen Woods files annually the following documents with the Connecticut Department of Social Services (a) a current disclosure statement (information booklet), (b) financial information, (c) escrow account verifications and escrow agreements, and (d) a filing fee. Evergreen Woods complies with all of the filing requirements under the Act.

All documents filed will be a matter of public record and may be reviewed at the office of the Department of Social Services, located at 55 Farmington Avenue in Hartford, Connecticut 06105.

NOTICE TO PROSPECTIVE RESIDENTS

Connecticut General Statutes require that the following Notice be given to prospective residents or their legal representative prior to the earlier of (i) the execution of a contract to provide continuing care (ii) the transfer of any money or other property to Evergreen Woods by or on behalf of the prospective Resident:

- A continuing care contract is a financial investment and your investment may be at risk.
- Our ability to meet our contractual obligations under such contract depends upon our financial performance.
- You are advised to consult an attorney or other professional experienced in matters relating to investments in continuing care facilities before you sign a contract for continuing care.
- The Connecticut Department of Social Services does not guarantee the security of your investment.

ACKNOWLEDGEMENT OF RECEIPT OF DISCLOSURE STATEMENT

In accordance with Sections 17b-522(b) and (f) of the Connecticut General Statutes, Evergreen Woods is required to deliver to a prospective resident or his or her legal representative a current Disclosure Statement not more than 60 days or less than 10 days before the execution of a continuing care contract or the transfer of any money or other property to Evergreen Woods by or on behalf of the prospective resident. In addition, Evergreen Woods is required to provide a revised and up-to-date Disclosure Statement to a prospective resident or his or her legal representative not more than 60 days or less than 10 days before the resident begins occupancy, or if no revisions have been made, a statement that there have been no revisions to the original Disclosure Statement.

Acknowledgement:

I, or my legal representative, have received and reviewed a copy of this notice and a copy of the continuing-care contract prior to entering into a continuing-care contract or the transfer of any money or other property to Evergreen Woods.

Signature of Prospective Resident

Date

Signature of Prospective Resident, if two

Date

EXHIBIT A

EXAMPLE TRADITIONAL RESIDENCY AGREEMENT

See attached. Note that there are three versions of the Traditional Residency Agreement that follow this form. The only substantive difference between the three versions is the refund provisions, which are described under "PROMISSORY NOTE" on page 33 of the Disclosure Statement.



TRADITIONAL RESIDENCY AGREEMENT 60% MINIMUM REFUND PLAN

88 Notch Hill Road

North Branford, Connecticut 06471

Telephone: (203) 488-8000

www.Evergreen-Woods.com

PAGE 1 of 34

GLOSSARY

The following terms are used in the accompanying Agreement. Reference to the Agreement and the context in which the terms are used is recommended to provide a fuller understanding of each of the terms:

"Agreement" means this Residency Agreement.

"Apartment" means the apartment at The Community in which you are entitled to live pursuant to the Agreement, as identified in **Section 1** of the Agreement.

"Assisted Living Services" means assistance with activities of daily living and nursing services provided by a licensed assisted living services agency ("ALSA").

"Bill of Rights" has the meaning set forth in Section 15.20 of the Agreement.

"Cash Requirements" has the meaning set forth in Section 12.3 of the Agreement.

"Cash Requirements Allocation" means your portion of The Community's Cash Requirements as determined by the Cash Requirements Allocation Formula.

"Cash Requirements Allocation Formula" means the method for determining your portion of The Community's Cash Requirements as set forth in **Section 12** of the Agreement.

"Entrance Fee" means those fees established by us and payable by a prospective resident to us for acceptance into The Community as described in **Section 1** of the Agreement.

"Evergreen Woods" or "we" or "us" means Evergreen Woods Retirement LLC, d/b/a Evergreen Woods, a limited liability company organized in the State of Connecticut and the operator of The Community.

"Monthly Charges" means all those monthly charges payable by you pursuant to the terms of the Agreement, including the Working Capital Fund Fees, the fees for optional services, fees for health care, if any, and all other fees and charges payable monthly pursuant to the terms of the Agreement, as appropriate in the particular circumstances.

"Monthly Service Fee" means that monthly fee payable in consideration for the services provided to residents, as set forth under Section 4.2 of the Agreement.

"Move-In Payment" has the meaning set forth in Section 1.3 of the Agreement.

"One-Time Capital Charge" means the one-time non-refundable charge set forth in **Section 3** of the Agreement.

"Operating Cash Requirements" has the meaning set forth in **Section 12.3.1** of the Agreement.

"Promissory Note" means the promissory note executed and delivered by Evergreen Woods to Resident pursuant to this Agreement in the form attached hereto as Exhibit A. Your rights to repayment of amounts due to you pursuant to the Promissory Note may be held by your heirs or others in trust established for your benefit if we consent to such an assignment, and our consent will not be unreasonably withheld.

"**Resident**" or "you" means the resident or residents who is (are) signatory to the Agreement, as identified in **Section 1** of the Agreement. Sometimes a second Resident (if there are two of you) is referred to in the Agreement as the "second person."

"Second Person Entrance Fee" means the fee established by us and payable by a prospective resident to us for acceptance of a second person pursuant to this Residency Agreement for acceptance into The Community as described in **Section 1** of the Agreement. The Second Person Entrance Fee is a one-time amount. The Second Person Entrance Fee is non-refundable.

"The Community" means the facility known as Evergreen Woods, and which is the subject of the Agreement, including the apartments, The Health Center at Evergreen Woods, and all common areas.

"The Health Center at Evergreen Woods" means the health center forming a part of The Community that is licensed to provide nursing care.

"Working Capital Fund Fee" means that monthly fee payable to owner on a per contract basis for occupancy in The Community, as set forth in **Section 4.1** of the Agreement.

EVERGREEN WOODS

TRADITIONAL RESIDENCY AGREEMENT (60% Minimum Refund)

1. ENTRANCE FEE AND PROMISSORY NOTE.

1.1 Entrance Fee. To assure you, _____ ("Resident"), a place at Evergreen Woods in North Branford, Connecticut (hereinafter "The Community") in accordance with all terms of this Agreement, including payment of the Monthly Charges described below, you will pay to Evergreen Woods an Entrance Fee of

If there are two of you, then there is a one-time non-refundable Second Person Entrance Fee of <u>\$</u>. This one-time non-refundable Second Person Entrance Fee is to be paid on or before the date you move into the Community.

1.2 Initial Deposit. You agree to pay ten percent (10%) of your Entrance Fee in the amount of $\underline{\$}$ when you sign this Agreement. We will hold this amount in escrow until such time as it may be released pursuant to Connecticut law.

1.3 Balance of Entrance Fee. You agree to pay the remaining balance (ninety percent (90%)) of your Entrance Fee in the amount of $\underline{\$}$ (the "Move-In Payment") to Evergreen Woods on or before the date you move into The Community. You shall not be required to move into The Community before the expiration of thirty (30) days from your execution of this Agreement. In no event shall the date you move into The Community (or pay the Move-In Payment and also the Second Person Entrance Fee, if applicable), extend beyond three (<u>3</u>) months from the date this Agreement is signed by both you and The Community.

1.4 Promissory Note. Upon receipt of the Move-In Payment, we will execute and deliver to you the Promissory Note substantially in the form attached to this Agreement as Exhibit A, and in the original principal amount of ninety percent (90%) of the Entrance Fee. Repayment of the Promissory Note will be made in accordance with its terms and the provisions of **Section 2.2** below. Your interest in

Payment of your Entrance Fee entitles you to live in the ______ Building, Apartment no. _____ ("Apartment") at The Community for as long as you are capable of meeting the requirements of apartment occupancy as set forth in our current entrance policy, and in The Health Center at Evergreen Woods when you are no longer capable of apartment occupancy, and to receive available services all in accordance with the terms of this Agreement.

the Promissory Note may not be mortgaged, sold, discounted, assigned, or otherwise transferred, except to certain trusts established for your benefit, subject to our prior approval and in our sole discretion.

2. REIMBURSEMENT OF ENTRANCE FEE.

2.1 Reimbursement of Entrance Fee Prior to Occupancy.

2.1.1 If you are not accepted for residency, as evidenced by Community-authorized signature of this Residency Agreement, the full amount of the Entrance Fee and any Second Person Entrance Fee that you have paid will be refunded to you without interest within five (5) business days.

2.1.2 You may cancel this Agreement at any time prior to occupying the Apartment for any reason. To cancel, you need to give us and the escrow agent, U.S. Bank, One Federal Street, 3rd Floor, Boston, MA 02110, written notice of cancellation by registered or certified mail and this Agreement will be canceled.

In such event, the full amount of the Entrance Fee and any Second Person Entrance Fee that you have paid will be refunded to you without interest within five (5) business days of written notice, except that we will retain an amount equal to any costs specifically incurred by us or The Community at your request (any such costs will be in a written addendum to this Agreement signed by both you and us).

2.2 Repayment of Promissory Note After Occupancy. In the event this Agreement is terminated or canceled after you have occupied your Apartment, the unpaid principal balance of the Promissory Note shall be repaid as stated following:

- a. The unpaid principal balance of the Promissory Note shall become due and payable to you or your estate, as the case may be, in accordance with the terms set forth below only when: (i) this Agreement has terminated; (ii) you have surrendered possession of the Apartment to us as required by this Agreement; and (iii) either we have received the full Entrance Fee from a substitute resident for the Apartment or a substitute resident moves into the Apartment or in accordance with **Section 2.2g**.
- b. We will be entitled to credits which will automatically reduce the principal balance of the Promissory Note in the following amounts, effective on the following dates:

A credit equal to 10% of the total Entrance Fee will be deducted from the principal balance on the Promissory Note if, on the first anniversary of the date that the Move-In Payment was made, this Agreement has not been terminated or you or the second person remains a resident in the Apartment or The Health Center at Evergreen Woods (i.e., we will then owe you an amount equal to 80% of the total Entrance Fee).

An additional credit equal to 10% of the total Entrance Fee will be deducted from the principal balance on the Promissory Note if, on the second anniversary of the date that your Move-In Payment was made, this Agreement has not been terminated or you or the second person remains a resident in the Apartment or The Health Center at Evergreen Woods (i.e., we will then owe you an amount equal to 70% of the total Entrance Fee).

A final credit equal to 10% of the total Entrance Fee will be deducted from the principal balance of the Promissory Note if, on the third anniversary of the date that your Move-In Payment was made, this Agreement has not been terminated or you or the second person remains a resident in the Apartment or The Health Center at Evergreen Woods (i.e., we will then owe you an amount equal to 60% of the total Entrance Fee).

The total of the credits which will occur on the first, second, and third anniversary of the date that your Move-in Payment was made is equal to 30% of the total Entrance Fee. Because the Promissory Note will initially be for an amount equal to 90% of the total Entrance Fee, the principal balance owed on the Promissory Note will not fall below 60% of the total Entrance Fee (less any amount you owe us under this Agreement at the time or after it terminates if we have elected to setoff such amount).

c. We agree to evaluate your financial condition, upon your request, and if need is demonstrated, to apply amounts due to you on the Promissory Note to amounts you would otherwise owe us for services in The Health Center at Evergreen Woods. Any amounts so applied will become a reduction in the unpaid principal balance of the Promissory Note (i.e., a "set-off").

- d. Upon payment to you of the unpaid principal balance due on the Promissory Note, you will execute all such documents and take all action necessary to cancel the Promissory Note.
- e. You will be responsible for payment of your own income taxes on any interest which accrues or is imputed on account of the Promissory Note pursuant to applicable state or federal law.
- f. If we have not received the Entrance Fee from a substitute resident for the Apartment or a substitute resident has not moved into the Apartment within two (2) years after this Agreement has terminated and you have surrendered possession of the Apartment to us and are not a resident in The Health Center at Evergreen Woods as required by this Agreement, then interest on the unpaid principal balance of the Promissory Note will accrue at the rate of four percent (4%) per annum and interest shall be paid to you monthly in arrears, starting on the tenth day of the month following the second anniversary of the termination date of this Agreement and continuing on the tenth day of each month thereafter until the Promissory Note is paid in full.
- g. If not sooner paid, the unpaid principal balance of the Promissory Note, plus unpaid accrued interest, shall be paid in full on the date which is three (3) years after this Agreement has terminated and you have surrendered possession of the Apartment and are no longer a resident of The Health Center at Evergreen Woods to us as required by this Agreement.
- h. Refurbishment Costs: You agree that we may deduct from the repayment to you of the principal balance our costs to refurbish the Apartment to like new condition when you leave. At a minimum, we will almost certainly replace the floor coverings and repaint the Apartment. You will be responsible for other costs in addition to paint and floor coverings for damage you, your guests, or your pets have caused to the Apartment, and such other costs will be deducted from the repayment to you of the principal balance.
- i. Tobacco Smoke Damage: You agree that the Apartment is a nonsmoking residence at all times. All of our costs (at any time) to refurbish the Apartment to like new condition which are related to tobacco smoke may be deducted from the repayment to you of the principal balance. The costs to repair smoke damages are substantial

and generally include multiple replacements (such as duct work, surface finishes, even ceilings). These costs are in addition to those of **Section 2.2h** above.

- j. Pet Damage: You agree that all of our costs (at any time) to refurbish the Apartment for damages caused by your pet(s) may be deducted from the repayment to you of the principal balance. These costs are in addition to those of **Section 2.2h** above.
- k. The provisions of this **Section 2.2** shall survive termination of this Agreement.

2.3 Right of Offset. We have the right to offset against any Entrance Fee refund or Promissory Note repayment any unpaid Monthly Charges owed by you, any unreimbursed health care expenses we have advanced on your behalf, any amounts deferred by us under **Section 10.2**, and any other sums owed by you to us.

3. ONE-TIME CAPITAL CHARGE. On or before the date you move into The Community, you will pay us the amount of (the "One-Time Capital Charge"). The One-Time Capital Charge is a one-time, non-refundable charge that will be used to fund apartment refurbishment and other capital improvements as determined by us.

4. MONTHLY CHARGES.

4.1 Working Capital Fund Fee. You shall pay a Working Capital Fund Fee of seventy-five dollars (\$75) per month for occupying your Apartment or The Health Center at Evergreen Woods. If there are two of you, a single monthly Working Capital Fund Fee shall cover both residents. There shall be no increase in your Working Capital Fund Fee during the term of this Agreement. The Working Capital Fund Fee shall be our unrestricted property, and may be used by us for any purposes.

4.2 Monthly Service Fees. You shall pay a Monthly Service Fee for the services and features provided to all residents as set forth in **Section 5**. During the current calendar year, this initial Monthly Service Fee will be $\underline{\$}$ per month for one person and an additional $\underline{\$}$ per month if there are two (2) of you occupying the Apartment. We will give you advance written notice of not less than thirty (30) days before any change in the Monthly Service Fee is implemented. Any change in the Monthly Service Fee may be made at our discretion.

4.3 Monthly Charges. During the current calendar year, you will pay fixed monthly charges of <u>\$</u>, which represent the Working Capital Fund Fee and Monthly Service Fee. In addition to the Working Capital Fund Fee and the Monthly Service Fee, you will also be charged for any additional services you may utilize.

4.4 Payment. On the day you move into The Community or as otherwise provided in Section 1.3 of this Agreement, you shall pay a prorated portion of the Working Capital Fund Fee and Monthly Service Fee. Thereafter, such fixed monthly charges will be payable on the first day of each month in advance, and your additional charges for optional services will be paid on the first day of each month for the optional services obtained during the preceding month.

4.5 Termination of Charges for Apartment. The Monthly Charges for your Apartment will terminate when (a) you vacate the Apartment and remove your possessions from it or (b) you release your Apartment pursuant to the provisions of Section 7 regarding usage of The Health Center at Evergreen Woods. In the event there are two (2) of you who occupy an Apartment and one of you dies, the second person's fees will cease and the remaining person will continue to pay the first person's Monthly Service Fee and Working Capital Fund Fee. Further, if your surviving spouse is not a resident of The Community or a party to this Agreement, then this Agreement will terminate upon your death and any fees or refunds owed will be handled pursuant to the terms herein.

4.6 Late Payment Charge; Interest and Collection Costs. Evergreen Woods reserves the right to impose a late payment charge and to assess interest at the rate of one and one-half percent (1.5%) per month for all balances that remain unpaid ten (10) days after the date on which they are due. Any account balances, including late payment charges and interest, that remain unpaid when this Agreement is terminated shall become a lien against your assets or estate after deducting any refund owed under this Agreement. Evergreen Woods also reserves the right to charge you for the costs of collection, including reasonable attorney's fees.

5. SERVICES AND FEATURES PROVIDED TO ALL APARTMENT RESIDENTS.

Evergreen Woods will furnish at The Community, so long as you reside in an apartment therein, the following services and features, which are included in the Monthly Service Fee:

5.1 Dining services and meals in the dining room per the Flexible Dining Dollar Points program;

5.2 Water, sewer, air conditioning, heating and electricity;

5.3 Basic cable TV service;

5.4 Property taxes, insurance on building and structures;

5.5 Grounds keeping, landscaping and snow removal;

5.6 Security services;

5.7 Building janitorial and maintenance service;

5.8 Weekly housekeeping service;

5.9 Weekly flat laundry service;

5.10 Planned activities - social, cultural and recreational, for those who wish to participate;

5.11 One reserved space for open parking;

5.12 Complete kitchen, including refrigerator, range with oven, microwave, garbage disposal, dishwasher and ductless hood fan;

5.13 Coin-free washers and dryers on each floor;

5.14 Scheduled local transportation;

5.15 Urgent call response system on a 24 hour basis;

5.16 Use of all common areas in The Community; and

5.17 Use of The Health Center at Evergreen Woods pursuant to the terms of this Agreement.

6. ADDITIONAL SERVICES PROVIDED FOR AN EXTRA CHARGE.

Evergreen Woods will also make available at The Community, at your request, so long as you reside in an apartment at The Community, for an extra charge at the then prevailing rates, additional meals over those provided in consideration for the Monthly Service Fee, housekeeping, personalized maintenance services, delivered meal service, certain office services such as "fax" and copier, use of guest rooms, catering for private parties, assisted living services, beauty parlor/barber shop, a limited number of garages, personal transportation and certain other services, upon special arrangements. Certain additional services for extra charge will also be available to you while you are in The Health Center at Evergreen Woods.

7. THE HEALTH CENTER AT EVERGREEN WOODS.

If, in the opinion of Evergreen Woods' Medical Director, after consultation with you and your legally authorized representative, if applicable, you need nursing care in The Health Center at Evergreen Woods, we will provide you with nursing care services to the extent authorized by our license from the State of Connecticut on the following terms:

7.1 First Ninety (90) Days of Nursing Care. We will provide nursing care in The Health Center at Evergreen Woods without additional charge for ninety (90) cumulative days for you (ninety (90) days for each of you if there are two of you, but allowance cannot be combined and used by only one of you), except that you (each of you if there are two of you) will pay the cost of the extra meals not covered by the Monthly Service Fee at the then current charge for extra meals. The Monthly Service Fees for your Apartment and the monthly Working Capital Fund Fees will continue as before.

7.2 More than Ninety (90) Days of Nursing Care When There is One of You. If there is one of you and you require nursing care beyond the ninety (90) cumulative days, your monthly charges will depend upon whether you choose to release your Apartment:

7.2.1 Release Apartment. When you release your Apartment for occupancy by someone else, the Monthly Service Fee charges for your Apartment will cease. You will pay for your nursing care monthly charges in an amount equal to the then current Monthly Service Fee. You will continue to pay your Working Capital Fund Fee. In addition, you will pay for the cost of the two meals per day not covered by the Monthly Service Fee at the then current charge for extra meals. You will not be entitled to full or partial repayment of the Promissory Note at the time you release your Apartment for relocation in The Health Center at Evergreen Woods.

7.2.2 Retain Apartment. If you choose not to release your Apartment, you will pay for your nursing care monthly charges in an amount equal to your Working Capital Fund Fee plus the then current Monthly Service Fee. In addition, you will continue to pay your Monthly Service Fee for your Apartment and

the monthly Working Capital Fund Fee. You will also pay the cost of the one meal per day not covered by the two (2) Monthly Service Fees at the then current charge for extra meals.

7.3 More than Ninety (90) Days of Nursing Care When There are Two of You. If there are two (2) of you, and one of you requires nursing care beyond the ninety (90) cumulative days, there will be no additional charge for nursing care, except that you will pay the cost of the extra meals per day not covered by the Monthly Service Fee at the then current charge for extra meals. The Monthly Service Fees for your Apartment and the monthly Working Capital Fund Fees will continue as before. If both of you require nursing care beyond the ninety (90) cumulative days, your monthly charges will depend upon whether you choose to release your Apartment:

7.3.1 Release Apartment. If you choose to release your Apartment for occupancy by someone else, the Monthly Service Fee charges for your Apartment (first and second person) will cease. You will pay for your nursing care monthly charges in an amount equal to the then current Monthly Service Fees (first and second person). You will continue to pay your Working Capital Fund Fees. In addition, you will pay for the cost of the four meals per day not covered by the Monthly Service Fees at the then current charge for extra meals. You will not be entitled to full or partial repayment of the Promissory Note at the time you release your Apartment for relocation in The Health Center at Evergreen Woods.

7.3.2 Retain Apartment. If you choose not to release your Apartment, you will pay for your nursing care monthly charges in an amount equal to your Working Capital Fund Fees plus the then current Monthly Service Fees. In addition, you will continue to pay your Monthly Service Fees (first and second person) for your Apartment and the monthly Working Capital Fund Fees. You will also pay the cost of the two (2) meals per day not covered by the four (4) Monthly Service Fees (two first person and two second person) at the then current charge for extra meals.

7.4 Outside Care. If, due to unusual circumstances, there are no beds available in The Health Center at Evergreen Woods, you agree to relocate to another health facility with which we will contract to provide nursing care. You will continue to pay applicable standard health center fees of The Community described in Section 7.3 above and other applicable monthly charges, but will incur no additional costs as a result of this relocation. You will be relocated to The Health Center at Evergreen Woods as soon as a bed is available.

7.5 Return to Apartment. If you release your Apartment because you have moved to The Health Center at Evergreen Woods, and if later you are able, in the opinion of the Medical Director or your attending physician with the concurrence of the Medical Director, to return to an apartment, we will provide you an apartment of the same type as your Apartment as soon as one becomes available. You will be permitted to remain in The Health Center at Evergreen Woods until such apartment becomes available. Upon reoccupying an apartment, your Monthly Charges will be based on the then current charges for that apartment.

7.6 Additional Health Services. We will also provide additional health services and supplies in The Health Center at Evergreen Woods such as therapy, pharmaceutical and medical supplies, personal laundry, and rental of equipment. These services and supplies are not included in the nursing care monthly charges and will be available for an extra charge.

7.7 Medical Director. We have designated a member in good standing of the New Haven County Medical Society to serve as Medical Director of The Health Center at Evergreen Woods. You will be at liberty to engage the services of the Medical Director or any other physician at your own expense. We will not be responsible for the cost of medical treatment by the Medical Director, or any other physician, nor will we be responsible for the cost of medicine, drugs, prescribed therapy, and other similar services and supplies. In the event we incur or advance costs for your medical treatment or for medicine, drugs, prescribed therapy, and other similar services and supplies, you will reimburse us for such costs.

7.8 Non-resident Use of The Health Center at Evergreen Woods. We offer nursing services in The Health Center at Evergreen Woods to qualified non-residents for a fee to the extent beds are available. Residents will be given priority to available beds.

7.9 Supplemental Insurance. You are required to maintain Medicare Part A, Medicare Part B and Medicare Part D or an equivalent insurance policy under a private or public plan. In addition you must maintain one supplemental health insurance policy or equivalent insurance coverage acceptable to us to assure your ability to fully cover a Medicare-qualified stay in The Health Center at Evergreen Woods. You shall furnish to us such evidence of coverage as we may from time to time request. Such supplemental insurance should cover Medicare co-insurance and deductibles. Should your supplemental health insurance or equivalent coverage not fully cover a Medicare-qualified stay in The Health Center at Evergreen Woods, or should you fail to purchase supplemental health insurance or equivalent coverage to fully cover a Medicare-qualified stay at The Health Center at Evergreen Woods, you

shall be financially responsible for paying deductibles, co-insurance amounts, and any other costs for each Medicare-qualified stay in The Health Center at Evergreen Woods. If you are eligible to receive the ninety (90) days of care in The Health Center at Evergreen Woods as provided in **Section 7.1** and your stay in The Health Center at Evergreen Woods is a Medicare-qualified stay, such ninety (90) days will be applied after the expiration of your Medicare-qualified stay providing you have Medicare Part A, Medicare Part B, Medicare Part D and acceptable supplemental health insurance or equivalent insurance coverage. If failure to maintain Medicare Part A, Medicare Part B, Medicare Part D (or equivalent insurance coverage) or supplemental health insurance is causing depletion of your resources and impairs your ability to meet your financial obligations, we need not defer your financial obligations as provided in **Section 10.2.3**, and we retain the right to revoke your license to reside at The Community and terminate this Agreement as provided in **Section 10**.

7.10 Managed Care. If you have chosen to participate in a managed care program as an alternative to Medicare Part A, Medicare Part B, and supplemental insurance coverage, the terms governing health center care will be as follows:

7.10.1 If The Health Center at Evergreen Woods is a participating provider with your managed care program and your stay is a Medicare-qualified stay, we agree to accept, as full payment, reimbursement at the rate negotiated with your managed care program. Such managed care stay in The Health Center at Evergreen Woods will not reduce the number of cumulative days of care that you are eligible to receive without additional charge pursuant to **Section 7.1**.

7.10.2 If The Health Center at Evergreen Woods is not a participating provider with your managed care program and you choose to receive health care services at a managed care participating provider during a Medicare-qualified stay, then you agree that you must relocate for as long as necessary for those services to be provided, and be responsible for all costs for those health care services. In addition, while receiving health care services at the managed care participating provider, you agree that unless this Agreement is terminated, you will continue to pay the Monthly Service Fee if you choose not to release your Apartment, and the Working Capital Fund Fee. Such a Medicare-qualified stay at a managed care participating provider other than The Health Center at Evergreen Woods will not reduce the number of cumulative days of care that you are eligible to receive without additional charge pursuant to **Section 7.1**.

7.10.3 If The Health Center at Evergreen Woods is not a participating provider in your managed care program and your stay is a Medicare-

qualified stay, we will attempt to negotiate an acceptable reimbursement rate with your managed care program. If we are able to negotiate an acceptable rate, we agree to accept as full payment the rate provided by your managed care program. Such a managed care stay in The Health Center at Evergreen Woods will not reduce the number of cumulative days of care that you are eligible to receive without additional charge pursuant to **Section 7.1**. If we are unable to negotiate an acceptable rate and you choose to receive health care in The Health Center at Evergreen Woods, you agree to be responsible for all charges for care in The Health Center at Evergreen Woods, as set forth in **Section 7.10.4** below.

7.10.4 If The Health Center at Evergreen Woods is not a participating provider in your managed care program and you would still like to receive health care in The Health Center at Evergreen Woods during a Medicarequalified stay, then each day of your stay in The Health Center at Evergreen Woods will reduce by one day the number of cumulative days of care that you are eligible to receive without additional charge pursuant to Section 7.1. During any such Medicare-qualified stay in The Health Center at Evergreen Woods, you agree that you will continue to pay the Monthly Service Fee if you choose not to release your Apartment, the Working Capital Fund Fee, and the charges for any additional health service you may receive pursuant to Section 7.6. If at any time during any such Medicare-qualified stay in The Health Center at Evergreen Woods you are no longer eligible to receive any of the cumulative days of care provided for in Section 7.1, then you agree to pay the per diem rate for your care in The Health Center at Evergreen Woods, the Monthly Service Fee if you choose not to release your Apartment, and the Working Capital Fund Fee, while you are assigned to The Health Center at Evergreen Woods.

7.10.5 At the conclusion of each such Medicare-qualified stay, you will be entitled to nursing care under the terms of this Agreement other than as set forth in this **Section 7.10**, as adjusted to reflect any reduction during such stay in the number of cumulative days of care that you are eligible to receive without additional charge pursuant to **Section 7.1**.

8. DURATION OF YOUR RIGHT TO OCCUPY THE APARTMENT.

You can live in your Apartment for as long as you (or either of you) live, unless you (or both of you) require services that cannot be provided or accommodated in the Apartment, or this Agreement is terminated by you or by us. If, in the opinion of the Medical Director, your physical or mental health requires that nursing care be given, you will relocate to The Health Center at Evergreen Woods where Evergreen Woods is licensed to provide such care.

9. YOUR VOLUNTARY CANCELLATION RIGHTS AFTER OCCUPANCY.

After you have moved into The Community you may cancel this Agreement at any time for any reason upon at least ninety (90) days' written notice signed by you (or both of you if there are two (2) of you) and sent to us by registered or certified mail. If you give such notice, you will pay all applicable Monthly Charges and any other daily charges pursuant to the terms of this Agreement until the later of the expiration of such ninety (90) day period or your departure from The Community. The balance due on your Promissory Note will be repaid to you pursuant to its terms.

Happiness Guarantee: You will have the right to terminate this Agreement within the first six (6) months after the earlier of when you move in or when you paid the Move-In Payment, by giving written notice to us and vacating the Apartment during that six (6) month period. In such event, we will refund the entire Entrance Fee (without interest) and refund the Second Person Entrance Fee (if applicable) that you have paid minus the following amount(s): (a) our cost to restore the Apartment to new condition and the cost for restoration for any damages you have made to the Apartment; and (b) any costs of any healthcare services provided to you. This will be paid to you upon the earliest of the following to occur: (i) our receipt of the full Entrance Fee from a substitute resident, (ii) the date a substitute resident moves into the Apartment, or (iii) the third anniversary of the date this Agreement is terminated. This Happiness Guarantee shall not apply for reasons of death. This Happiness Guarantee shall not apply if you move directly from your Apartment to a healthcare facility or you move to a healthcare facility within six (6) months of the date you move out of Evergreen Woods.

10. OUR TERMINATION RIGHTS.

10.1 Just Cause. We will not terminate this Agreement except for just cause, which includes, but is not limited to, the following:

10.1.1 Except as set forth below, failure to pay any Monthly Charges or the Entrance Fee as provided herein;

10.1.2 Creation by you of a disturbance within The Community which in our judgment is detrimental to the health, safety, comfort or peaceful lodging of others;

10.1.3 You become infected with a dangerous and contagious disease or become mentally or emotionally disturbed, and the Medical Director determines that your condition is detrimental to the health, safety, or welfare of

others and The Health Center at Evergreen Woods cannot care for you within the limits of our license from the State;

10.1.4 You do not comply with the terms of this Agreement or the published operating procedures, covenants, rules, regulations and policies now existing or later created or amended by us; or

10.1.5 You refuse medical treatment which in the opinion of the Medical Director is medically required for your health or the health or safety of others.

10.2 Limitations on Termination Rights for Financial Inability. If, after you have paid the Entrance Fee, the One-Time Capital Charge, and the Second Person Entrance Fee, if applicable, you encounter financial difficulties making it impossible for you to pay the full Monthly Charges and other charges appropriate for the Apartment or for nursing care in The Health Center at Evergreen Woods, then:

10.2.1 You may remain until any applicable Title XVIII Medicare benefits and/or third party insurance benefits received by us on your behalf have been earned;

10.2.2 You shall in any case be permitted to remain at The Community for ninety (90) days after the date of failure to pay, during which time you shall continue to pay a reduced Monthly Charge based on your current financial condition; and

10.2.3 Because it is and shall continue to be our declared policy to not terminate your residency solely by reason of your financial inability to pay the full Monthly Charges and other charges, you shall be permitted to remain at The Community at a reduced Monthly Charge based on your ability to pay for so long as you establish facts, satisfactory to us in our sole discretion, to justify deferral of the usual charges, and the deferral of such charges can, in our sole discretion, be granted without impairing our ability to operate on a sound financial basis. This provision shall be rendered inoperative if you have impaired your ability to meet your financial obligations hereunder by making unapproved gifts or other transfers or by not maintaining Medicare Part A, Medicare Part B, Medicare Part D and/or adequate supplemental insurance required under **Section 7.10**. Any such charges deferred shall be offset against the refundable portion of your Entrance Fee when it is paid to you or your estate in accordance with the Promissory Note.

10.3 Notice of Termination. Prior to any termination of the Agreement by us, we will give you notice in writing of the reasons, and you will have thirty (30) days thereafter to correct the problem. If the problem is corrected within such thirty (30) day period, this Agreement shall not be then terminated. If the problem is not corrected within such thirty (30) day period, this Agreement will be terminated and you must vacate the Apartment, remove all your possessions, and leave The Community.

10.4 Emergency Termination. Notwithstanding the above, if the Medical Director determines that either the giving of notice or the lapse of time as above provided might be detrimental to the health, safety, or welfare of you or others, then such notice and/or waiting period prior to termination and relocation to a hospital or other appropriate facility shall not be required, and termination of this Agreement shall be deemed to have occurred when you are relocated. In such event, we are expressly authorized to transfer you to a hospital or other appropriate facility and will promptly notify your representative or your attending physician.

11. MISCELLANEOUS PROVISIONS WITH RESPECT TO YOUR APARTMENT.

11.1 Use of Apartment. The Apartment is for living only and shall not be used for carrying on any business or profession, nor in any manner in violation of zoning restrictions or any other municipal, state or federal law.

11.2 Occupancy of Apartment. Except as hereinafter provided, no person other than you (or both of you) may occupy the Apartment except with our express written approval. In the event that a second person who is not a party to this Agreement is accepted for residency under this Agreement at a time subsequent to the date hereof (said acceptance to be in accordance with admission policies governing all other admissions), you shall pay an Entrance Fee and, if applicable, a Second Person Entrance Fee, as determined by us, and each month thereafter the then current additional Monthly Charges for second persons shall be paid. If such second person does not meet the requirements for residency, such second person will not be permitted to occupy the Apartment for more than thirty (30) days (except with our express written approval), and you may cancel this Agreement as provided in **Section 9**. If such second person is already a resident of The Community and the two of you would like to live in the same apartment under one residency agreement, you may cancel this Agreement as provided in **Section 9**.

11.3 Compliance Changes. We may effect changes in The Community at any time to meet the requirements of the law. You agree to temporarily relocate to

other facilities provided by us without additional cost to you if it becomes necessary to vacate your Apartment in order to make such changes.

11.4 Furnishings. Furnishings within the Apartment will not be provided by us except to the extent provided in consideration for the Monthly Service Fee. Furnishings provided by you shall not interfere with the health, safety and general welfare of other residents or the staff of The Community. If removal of your furniture and other property is not accomplished upon termination of this Agreement or within thirty (30) days after your death (if there are two of you, the death of the survivor), then we may remove and store such furniture and other property at the expense of you or your estate. We may dispose of such furniture or other property not claimed within ninety (90) days after your death (if there are two of you, the death of the survivor) or termination of this Agreement.

11.5 Alterations by You. You may not undertake any alterations to your Apartment without our prior written approval. We reserve the right, upon your release of the Apartment because you have moved to The Health Center at Evergreen Woods, or upon termination or cancellation of this Agreement, to require that any alterations made by you be removed, and that the Apartment be restored to its original design at your expense. The cost of any such restoration may be withheld from the Promissory Note when it is paid to you or your estate in accordance with its terms.

11.6 Refurbishment. We will provide maintenance and repair to your Apartment for property, furnishings, and equipment owned or leased by Evergreen Woods. Other maintenance services may be available at an additional cost. Should you desire to have the Apartment refurbished in the future (i.e., new paint, floor coverings, wall coverings, counter-tops, etc.), such services may be available at an additional cost to you. You agree to pay for any necessary refurbishments caused by you or your guests that are beyond those which are customary and normal, including, but not limited to, the costs of cleaning and removal of stains, odors, and other damage caused by smoking in your apartment.

11.7 Emergency Entry. We may enter your Apartment should it be necessary to protect your health or safety or the health or safety of other residents.

12. CASH REQUIREMENTS ALLOCATION FORMULA

The amount of your Monthly Service Fee is determined by application of the Cash Requirements Allocation Formula. The Cash Requirements Allocation

Formula has been developed to equitably apportion the costs of operating The Community among its residents who are under a traditional residency plan.

12.1 General. All projected Cash Requirements (including additions to reserves) necessary to provide accommodations and services to the residents of The Community (including residents of The Health Center at Evergreen Woods) are charged to The Community residents through the application of the Cash Requirements Allocation Formula. Generally, the projected Cash Requirements are based upon the costs incurred in connection with the operation of The Community for the preceding year with adjustments for inflation or deflation, occupancy rates, additions to reserves, and nonrecurring expenses, and reduced by projected revenues (such as fees for services, and certain investment income, etc.) for the upcoming year. Once determined, the projected Cash Requirements are then allocated among The Community residents based upon the size of the apartment that each resident occupies and the number of residents in The Community.

12.2 Benefits to Us. No portions of the costs included in the Cash Requirements Allocation Formula are paid to Evergreen Woods as fees. Our compensation for the on-going ownership of The Community is comprised solely of the Entrance Fee, the Second Person Entrance Fee, if applicable, monthly Working Capital Fund Fee (which is fixed for you for the duration of your occupancy), interest earned on deposits and/or certain escrowed monies, the appreciation of The Community, if any, and other benefits generally associated with the ownership of real estate. In the event, however, that we loan funds to operations that would normally be funded through the Cash Requirements Allocation and/or fund any differences between operating revenues and operating expenses, we may, at our option, recoup such amounts funded through the Monthly Service Fees.

12.3 Cash Requirements. As set forth in this **Section 12.3**, your Monthly Service Fee will be adjusted upward or downward each year according to the Cash Requirements of The Community, as allocated to each apartment and all second persons. The "Cash Requirements" to be allocated shall be equal to:

12.3.1 Operating Cash Requirements for the preceding year (annualized if the actual period is less than a full operating year), which shall mean all recurring and nonrecurring costs, expenses and outlays incurred in connection with the operation and administration of The Community. "Operating Cash Requirements" shall be determined by The Community management and shall include but not be limited to salaries and wages and all fringe benefits paid to employees of The Community, and other persons stationed at The Community, and other persons having day-to-day administrative, marketing and/or management

responsibilities; payroll taxes; telephone; utilities; insurance premiums; license fees, occupational taxes, property taxes, sales taxes, permits, etc., associated with the operation of The Community; lease payments, if any; legal and accounting services; bad debts and other losses related to the operation; deductibles related to insurance claims; consulting fees for specialists such as dietitians and medical directors; food and supplies; replacement and repair of equipment and all common furniture and fixtures; individual capital improvements, replacements or additions costing less than fifty thousand dollars (\$50,000) and not covered by Section 12.3.2; all interior building elements such as carpet, doors, repainting, and glass damage; management fees plus costs of reasonable expenses incurred in performing management services for The Community, such as transportation, living expenses, telephone expense and photocopying expense; cost of maintenance and normal replacement of landscaping; principal and interest payments on all loans except loans from residents; and other similar items related to the operation of The Community. Operating Cash Requirements shall not include income taxes, dividends, or individual capital improvements, and other replacements or additions that cost in excess of fifty thousand dollars (\$50,000); plus

12.3.2 An amount to maintain or increase the reserve for replacement described in **Section 12.6** shall be equal to three percent (3%) of the prior year's Operating Cash Requirements. This reserve is in addition to any operating reserves required under Connecticut law. Deposits to this reserve and earnings from this reserve may only be used for improving or replacing capital items of The Community that cost over fifty thousand dollars (\$50,000); <u>plus</u>

12.3.3 An amount projected by us to account for expected changes from the previous year in services, The Health Center at Evergreen Woods utilization, apartment occupancy, marketing, remodeling or any other material factor; <u>plus</u>

12.3.4 A reasonable estimate for inflation or deflation in the cost of operating The Community for the coming fiscal year; <u>plus</u>

12.3.5 An amount to maintain or increase the reserve for operating cash established under **Section 12.6**. Deposits to this reserve may only be used for items related to the operation of The Community; <u>plus</u>

12.3.6 An amount to maintain or increase the amount of any reserve required by the laws of the State of Connecticut; <u>minus</u>

12.3.7 The projected net revenues to be received in the coming fiscal year from (i) The Health Center at Evergreen Woods' operations; (ii) extra fees received for the furnishing of additional services to the residents as provided in **Sections 6** and **7** of this Agreement; (iii) space rentals from such support services as barber/beauty shops; (iv) income from investment of reserves required by the laws of the State of Connecticut; and (v) nonrecurring operating income.

12.4 Allocation of Cash Requirements. Allocation of the sum of the Cash Requirements described in Section 12.3 to each individual resident's apartment and all second persons shall be accomplished by:

12.4.1 Multiplying the number of apartments in The Community of each size and the number of second person residents therein by the following factors and then summing such products:

Number of One-Bedroom Alcove (Type A) Apartments X 70.6 percent factor

Number of One-Bedroom Apartments (Type B) X 84.3 percent factor

Number of One-Bedroom Custom Apartments (Type C) X 86.3 percent factor

Number of One-Bedroom Deluxe Apartments (Type D) X 96.1 percent factor

Number of Two-Bedroom Apartments (Type E) X 100.0 percent factor

Number of Two-Bedroom Custom Apartments (Type F) X 102.0 percent factor

Number of Two-Bedroom Corner Apartments (Type FC) X 102.0 percent factor

Number of Two-Bedroom Deluxe Apartments (Type G) X 107.8 percent factor

Number of Two-Bedroom Deluxe Apartments (Type GL) X 107.8 percent factor

Number of Second Persons X 45.1 percent factor

To take into account the combination or addition of different apartments to The Community, or similar events that require the addition of new factors, factors may be added to or deleted from the above formula. **12.4.2** Multiplying the sum determined in **Section 12.4.1** above by a factor representing the projected apartment occupancy rate.

12.4.3 Dividing the sum of the Cash Requirements determined under Section 12.3 by the amount determined under Section 12.4.2 to arrive at the arithmetic quotient.

12.4.4 The Cash Requirements Allocation for each apartment size and for the second person shall be the product of multiplying the arithmetic quotient determined under **Section 12.4.3** by the factor assigned to each apartment and to the second person in **Section 12.4.1**. The amount of increase or decrease in the Monthly Service Fee will be the difference between the new Cash Requirements Allocation for each apartment size and for the second person and the applicable Cash Requirements Allocation for the preceding year, divided by twelve (12).

12.5 Determination and Implementation of Cash Requirements Allocation. At least thirty (30) days prior to the end of each fiscal year, we will notify you of the new Monthly Service Fee that will go into effect at the start of the next fiscal year.

12.6 Additional Reserves. Evergreen Woods has established reserves as follows: (a) A reserve for replacement (as described in Section 12.3.2); (b) an operating cash reserve (as described in Section 12.3.5); and (c) a reserve required by Connecticut law (as described in Section 12.3.6). The total amount of the operating cash reserves includes operating reserves required by Connecticut law. Any amounts necessary to maintain or increase such additional reserves shall, however, be included in your Cash Requirements Allocation. Copies of these reserve agreements with the bank are available upon request.

13. ARBITRATION.

Any dispute, claim or controversy of any kind between the parties arising out of, in connection with, or relating to this Agreement and any amendment hereof, or the breach hereof, shall be submitted to and determined by arbitration in North Branford, Connecticut in accordance with the commercial arbitration rules then obtaining of the American Arbitration Association. All parties shall be bound by the arbitrator's decision and judgment upon such decision may be entered in any federal or state court having jurisdiction unless the arbitration is fraudulent or so grossly erroneous as to necessarily imply bad faith. If the parties are unable to agree in good faith and within a reasonable time on the selection of an arbitrator, any party may request appointment of an arbitrator by the American Arbitration Association. The parties shall make a reasonable attempt to select an arbitrator with experience in retirement communities, long-term care or health care operations generally. Cost of arbitration shall be shared by the parties equally. If the issue affects more than one resident, we may elect to join all affected residents into a single arbitration proceeding, and you hereby consent to such joinder.

14. AMENDMENTS.

14.1 This Agreement. Subject to **Section 14.2**, this Agreement may be amended only by written agreement of the parties to this Agreement.

14.2 Compliance with Laws. This Agreement may be modified by us at any time in order to comply with laws and regulations upon thirty days' notice to you.

15. MISCELLANEOUS LEGAL PROVISIONS.

15.1 Governing Law. This Agreement will be interpreted according to the laws of the State of Connecticut and will become effective upon acceptance and execution by us. The Glossary which sets forth the definitions of certain terms used in this Agreement is by this reference incorporated herein and made a part of this Agreement.

15.2 Separability. The invalidity of any restriction, condition or other provision of this Agreement, or any part of the same, shall not impair or affect in any way the validity or enforceability of the rest of this Agreement.

15.3 Capacity. This Agreement has been executed on our behalf by our duly authorized agent, and no member, partner, officer, director, agent or employee of Evergreen Woods shall have any personal liability hereunder to Resident under any circumstances.

15.4 Residents. When Resident consists of more than one person, the rights and obligations of each are joint and several, except as the context otherwise requires.

15.5 Resident's Rights. You understand and agree that (a) this Agreement or your rights (including the use of the Apartment) under it may not be assigned and no rights or benefits under this Agreement shall inure to the benefit of your heirs, legatees, assignees, or representatives, except as to reimbursement of the amounts as described in **Section 2**; (b) this Agreement and your contractual right to occupy the Apartment shall exist and continue to exist during your lifetime unless canceled

by you or terminated by us as provided herein; (c) this Agreement grants you a revocable license to occupy and use space in The Community but does not give you exclusive possession of the Apartment as against us and you shall not be entitled to any rights of specific performance but shall be limited to such remedies as set forth herein; (d) this Agreement is not a lease or easement and does not transfer or grant you any interest in real property owned by us; and (e) this Agreement grants to us complete decision making authority regarding the management and operation of The Community.

15.6 Release. We are not responsible for loss of or damage to your personal property. You may want to obtain, at your own expense, insurance to protect against such losses. We strongly suggest you keep and maintain your own insurance for liability and property loss.

15.7 Indemnity. We shall not be liable for, and you agree to indemnify, defend and hold us harmless from claims, damages, settlements, and expenses, including attorney's fees and court costs resulting from any injury or death to persons and any damages to property caused by, resulting from, attributable to or in any way connected with your negligent or intentional act or omission.

15.8 Entire Agreement. This Agreement and any Addenda or Exhibits express the entire agreement between the parties hereto and supersede any prior or contemporaneous written or oral understanding or agreement.

15.9 Reimbursement of Loss or Damage. You or your representative, if applicable, will reimburse us for any loss or damage to The Community as the result of your intentional, careless, or negligent acts or omissions or that of your guests. You agree to obtain insurance for you and your property and agree to provide us with proof of such insurance upon request.

15.10 Subordination. Except for your right of cancellation prior to occupancy, you agree that all your rights under this Agreement will always be subordinate and junior to the lien of all indentures of trust, mortgages or other documents creating liens encumbering The Community or any of the assets of Evergreen Woods, which have been or will be executed by us. Upon request, you agree to sign, acknowledge and deliver to such lender or lenders such further written evidence of such subordination as such lenders may reasonably require. You will not be liable for any such indebtedness.

15.11 Transfers. We may from time to time issue additional equity interests, or sell The Community, provided that in such latter event the buyer shall agree to

assume all existing residency agreements. Your signature hereto constitutes your consent and approval to any such future transfer.

15.12 Law Changes. If changes are made in any of the statutes or regulations applicable to this Agreement, we shall have the right to submit to you a revised Agreement based on the changes in the law, and you agree to accept any such revisions that do not adversely affect you.

15.13 Private Employee of Resident. If you need additional services, you can obtain these needed services from a private employee (privately employed personnel). In such instances, you are encouraged to obtain these needed services from a licensed and/or certified home health agency. Further, you must comply with our policy regarding privately employed personnel and ensure that your private employee complies with our policies and rules of conduct as set forth by us. If you fail to follow the reasonable rules set forth by us, which will be provided to you prior to the time you sign this Agreement, and thereafter in the event of revisions or additions, then we may elect at our sole option to terminate this Agreement in accordance with **Section 10**.

15.14 Responsible Party. You agree to execute and deliver to us within sixty (60) days after assuming occupancy in your Apartment a Durable Power of Attorney, trust documents, or other documentation naming a responsible party for business and financial decision making. You also agree to provide us a copy of, or execute and deliver within the same sixty (60) day time frame, an Appointment of Health Care Representative or a valid Durable Power of Attorney for Health Care Decisions. These documents should be drafted to remain effective notwithstanding your incompetence or disability and shall be in a form acceptable to us, and you agree to keep such documents in effect as long as this Agreement is in effect. The person(s) named as your responsible party shall not be a person(s) employed by us or any other entity engaged in the management of The Community.

15.15 Residents' Association. Residents shall have the right to organize and operate a residents' association at The Community and to meet privately to conduct business of the residents' association.

15.16 Rights of Third Parties. Only parties who have executed this Residency Agreement will have any rights hereunder. Neither this Residency Agreement nor the disclosure statement of Evergreen Woods create any rights for third parties other than those who have executed this Residency Agreement.

15.17 Estate Planning. Evergreen Woods will pay any amounts due pursuant to the Promissory Note to your estate upon your death, or in the case of a double-occupied apartment, upon the death of the surviving resident, in accordance with the repayment terms of the Promissory Note. If you do not wish to have the amounts due pursuant to the Promissory Note paid to your estate, you may choose to assign your right to the Promissory Note to a trust or other person designated by you pursuant to a form of Assignment of Promissory Note Proceeds approved by Evergreen Woods. In the absence of any agreement between the residents of a double occupied apartment which has been provided to Evergreen Woods prior to any repayment due on the Promissory Note to the estate of the surviving resident. No other rights under this Residency Agreement are assignable.

15.18 Notices. All written notices required by this Agreement will be sufficient if addressed; if to you, to your Apartment at Evergreen Woods; if to Evergreen Woods, to the attention of the Executive Director, Evergreen Woods, 88 Notch Hill Road, North Branford, CT 06471.

15.19 Compliance with Laws. You will be afforded all rights and privileges under Conn. Gen. Stat. § 17b-520 <u>et seq</u>., which governs continuing care facilities, and all other applicable laws. Evergreen Woods will comply with all municipal, state and federal laws and regulations regarding consumer protection and protection from financial exploitation.

15.20 Managed Residential Community Residents' Bill of Rights. Evergreen Woods is registered as a "managed residential community" with the Connecticut Department of Public Health. Accordingly, Evergreen Woods is required to provide and explain to you, at the time of execution of this Agreement, the Managed Residential Community Residents' Bill of Rights attached hereto as Exhibit B (the "Bill of Rights"). A staff member will explain the Bill of Rights to you, and you will be asked to separately sign the Bill of Rights to acknowledge that you have received it.

15.21 Rules and Regulations. Evergreen Woods shall have the right to adopt, or amend, either by itself or with or through the residents' council, such reasonable rules and regulations as it deems necessary or desirable for the proper management and operation of the Community and for the safety, health and comfort of the residents. The rules and regulations in effect at the time of execution of this Agreement are set forth in the Resident Handbook. You agree to abide by such rules and regulations, as they may be amended from time to time. The Resident Handbook

includes procedures for you to address any concerns or complaints you may have during your residency.

16. RESERVE POLICY. It will be our policy to maintain reserve funds as required by Connecticut law.

17. RESIDENT REPRESENTATIONS/OBLIGATIONS.

By executing this Agreement you represent and warrant that you are at least sixty-two (62) years of age or older at the time of admission, that you are capable of meeting the requirements of admission as set forth in our current admission policy, that you have assets and income which are sufficient under foreseeable circumstances and after provision for payment of your obligations under this Agreement to meet your ordinary and customary living expenses after assuming occupancy, and that all written representations made with respect to such matters by you or on your behalf to us are true. From time to time, Evergreen Woods may request that you update the financial information you submitted prior to admission. As a condition of continued residency in The Community, you agree to provide such financial information to Evergreen Woods when requested. You agree not to impair your ability, in any manner described in **Section 10.2.3** of this Agreement, to meet your financial obligations under this Agreement.

18. ACKNOWLEDGEMENT OF RECEIPT OF DOCUMENTS.

Under the requirements of applicable law, we are obligated to remind you that:

(1) this Agreement, which is a continuing care contract, is a financial investment and your investment may be at risk;

(2) our ability to meet our contractual obligations under this Agreement depends on our financial performance;

(3) you have been advised to consult an attorney or other professional experienced in matters relating to investments in continuing care facilities before you signed this Agreement; and

(4) the Department of Social Services of the State of Connecticut does not guarantee the security of your investment.

BALANCE OF PAGE LEFT BLANK. SIGNATURE PAGE FOLLOWS.

You hereby certify that you received a copy of this Agreement and a copy of our latest disclosure statement on or before the date hereof, and have been permitted to inspect any additional relevant materials requested to be reviewed by you or your representatives prior to signing this Agreement.

Executed at,,	, this	_ day of
RESIDENT	EVERGREEN WOODS RETIREMENT LLC d/b/a Evergreen Woods	
Witness		
RESIDENT	By: Executive Director	
Witness	Date:	
Date:		
Apartment Number:	Building Name:	
Scheduled Move-In Date:		
Entrance Fee: \$	One Time Capital Charge: \$	
Non-Refundable One-Time Second	Person Entrance Fee: \$	
1 Person Monthly Service Fee: \$	(plus \$75 Working Capital of	charge)
2 nd Person addition to Monthly Serv	vice Fee: \$	
\diamond		



EXHIBIT A (TO BE EXECUTED AT OCCUPANCY)

PROMISSORY NOTE

\$

North Branford, Connecticut

FOR VALUE RECEIVED, the undersigned ("Maker") promises to pay to ______ ("Holder") or order, the principal sum of ______ Dollars (\$_____) payable in lawful money of the United States of America, at such place as the legal holder hereof may designate in writing, subject to the following.

This Note is given to evidence certain obligations of Maker pursuant to an Residency Agreement dated as of ______, ____ between the undersigned and Holder regarding residency at Evergreen Woods (the "Residency Agreement") and is subject to the terms and conditions of the Residency Agreement, which are incorporated into and made a part of this Note by this reference.

Except as provided in Section 2.2 of the Residency Agreement, no interest shall accrue on the outstanding principal balance of this Note. Interest on this Note shall accrue as provided in Section 2.2 of the Residency Agreement. Maker shall make monthly payments of accrued interest, if any, on the tenth day of each calendar month.

The principal amount of indebtedness evidenced by this Note shall be automatically reduced in accordance with Section 2.2 of the Residency Agreement. If not sooner paid, the entire indebtedness, less a credit for any amount owed by Holder to Maker under the Residency Agreement, shall be due and payable on the date described in Section 2.2 of the Residency Agreement.

This Note may be prepaid in full or in part at any time without penalty or premium.

In the event of default in payment of any installment of principal or interest hereof as the same becomes due and such default is not cured within ten (10) days after written notice to Maker, then the Holder of this Note may without further notice, declare the remainder of the principal sum, together with all interest accrued thereon, at once due and payable. Failure to exercise this option shall not constitute a waiver of the right to exercise the same at any other time. The unpaid principal of this Note and any part thereof, accrued interest and all other sums due under this Note shall bear interest at the rate of twelve percent (12%) per annum after default until paid. All parties to this Note, including Maker and any sureties, endorsers, or guarantors hereby waive protest, presentment, notice of dishonor, and notice of acceleration of maturity and agree to continue to remain bound for the payment of principal, interest and all other sums due under this Note notwithstanding any extension or extensions of time for the payment of principal and interest; and all such parties waive all and every kind of notice of such change or changes and agree that the same may be made without notice or consent of any of them.

Upon default the Holder of this Note may employ an attorney to enforce the Holder's rights and remedies and the Maker, principal, surety, guarantor and endorsers of this Note hereby agree to pay to the Holder reasonable attorney's fees not exceeding a sum equal to fifteen percent (15%) of the outstanding balance owing on this Note, plus all other reasonable expenses incurred by the Holder in exercising any of the Holder's rights and remedies upon default. The rights and remedies of the Holder as provided in this Note shall be cumulative and may be pursued singly, successively, or together, in the sole discretion of the Holder. The failure to exercise any such right or remedy shall not be a waiver or release of such rights or remedies or the right to exercise any of them at another time.

Maker holds a leasehold interest in and operates Evergreen Woods. If Maker, (i) after obtaining the prior written consent of the owner of the fee interest in Evergreen Woods, assigns its leasehold interest in Evergreen Woods and (ii) the assignee assumes in writing all of Maker's obligations under the Residency Agreement and this Note, then Maker shall be released from all obligations under this Note and the term "Maker" shall thereafter mean such assignee.

This Note is to be governed and construed in accordance with the laws of the State of Connecticut.

This Note is given to secure a debt.

IN TESTIMONY WHEREOF, Maker has executed this Note under seal as of the day and year first above written.

EVERGREEN WOODS RETIREMENT, LLC, a Connecticut limited liability company (SEAL)				
By:	Senior Living Communities, LLC (Manager	(SEAL)		
By:	Title:	(SEAL)		
PAGE	31 of 34			

EXHIBIT B



MANAGED RESIDENTIAL COMMUNITY RESIDENTS' BILL OF RIGHTS

You have the right to:

- Live in a clean, safe and habitable private residential unit;
- Be treated with consideration, respect and due recognition of your personal dignity, individuality and the need for privacy;
- Privacy within your private residential unit, subject to The Community's rules reasonably designed to promote your health, safety and welfare;
- Retain and use your personal property within your apartment so as to maintain individuality and personal dignity, provided that your use of personal property does not infringe on the rights of other residents or threaten the health, safety and welfare of other residents;
- Private communications, including receiving and sending unopened correspondence, telephone access and visiting with persons of your choice;
- Freedom to participate in and benefit from community services and activities so as to achieve the highest possible level of independence, autonomy and interaction within The Community;
- Directly engage or contract with licensed health care professionals and providers of your choice to obtain necessary health care services in your private apartment, or such other space in The Community as may be available to residents for such purposes;
- Manage your own financial affairs;
- Exercise civil and religious liberties;
- Present grievances and recommend changes in policies, procedures and services to our Executive Director or staff, government officials or any

other person without restraint, interference, coercion, discrimination or reprisal from The Community, including access to representatives of the Department of Public Health or the Office of the Long-Term Care Ombudsman;

- Upon request, obtain the name of the Executive Director or any other persons responsible for resident care or the coordination of resident care;
- Confidential treatment of all records and communications to the extent required by state and federal law;
- Have all reasonable requests responded to promptly and adequately within the capacity of The Community and with due consideration given to the rights of other residents;
- Be fully advised of the relationship that The Community has with any Assisted Living Services Agency, health care facility or educational institution to the extent that such relationship relates to resident medical care or treatment and to receive an explanation about the relationship;
- Receive a copy of any rules or regulations of The Community;
- Privacy when receiving medical treatment or other services within the capacity of The Community;
- Refuse care and treatment and participate in the planning for the care and services you need or receive, provided, however, that your refusal of care and treatment may preclude you from being able to continue to reside in The Community; and
- If you are a continuing care resident, all rights afforded under Conn. Gen. Stat. § 17b-520 <u>et seq</u>. and any other applicable laws. If you rent your apartment, all rights and privileges afforded to tenants under title 47a of the Connecticut General Statutes (Connecticut's landlord tenant laws).

We hope that any complaints or concerns that you have can be resolved by our staff. However, you also have the right to contact the following state agencies regarding complaints or concerns:

Department of Public Health Facilities Licensing and Investigations 410 Capitol Ave., P.O. Box 340308 MS#12HSR Hartford, CT 06134-0308 Information/General: Loan Nguyen, R.N. Supervising Nurse Consultant (860) 509-7400

Office of the Long Term Care Ombudsman Dan Lerman 370 James Street Suite 306 New Haven, CT 06513 (860) 823-3366

If you are receiving nursing or personal care from an Assisted Living Services Agency, you also have other rights set forth separately in the Assisted Living Clients' Bill of Rights.

Please sign below to acknowledge that we have provided you with a copy of the Managed Residential Community Residents' Bill of Rights and explained them to you.

Date

Resident

Resident's Representative

Relationship to Resident

EXHIBIT B

EXAMPLE UNBUNDLED RESIDENCY AGREEMENT

See attached. Note that there are three versions of the Unbundled Residency Agreement that follow this form. The only substantive difference between the three versions is the refund provisions, which are described under "PROMISSORY NOTE" on page 33 of the Disclosure Statement.



UNBUNDLED RESIDENCY AGREEMENT 60% MINIMUM REFUND PLAN

88 Notch Hill Road

North Branford, Connecticut 06471

Telephone: (203) 488-8000

www.Evergreen-Woods.com

PAGE 1 of 30

GLOSSARY

The following terms are used in the accompanying Agreement. Reference to the Agreement and the context in which the terms are used is recommended to provide a fuller understanding of each of the terms:

"Agreement" means this Residency Agreement.

"Apartment" means the apartment at The Community in which you are entitled to live pursuant to the Agreement, as identified in **Section 1** of the Agreement.

"Assisted Living Services" means assistance with activities of daily living and nursing services provided by a licensed assisted living services agency ("ALSA").

"Bill of Rights" has the meaning set forth in Section 15.20 of the Agreement.

"Cash Requirements" has the meaning set forth in Section 12.3 of the Agreement.

"Cash Requirements Allocation" means your portion of The Community's Cash Requirements as determined by the Cash Requirements Allocation Formula.

"Cash Requirements Allocation Formula" means the method for determining your portion of The Community's Cash Requirements as set forth in **Section 12** of the Agreement.

"Entrance Fee" means those fees established by us and payable by a prospective resident to us for acceptance into The Community as described in **Section 1** of the Agreement.

"Evergreen Woods" or "we" or "us" means Evergreen Woods Retirement LLC, d/b/a Evergreen Woods, a limited liability company organized in the State of Connecticut and the operator of The Community.

"Monthly Charges" means all those monthly charges payable by you pursuant to the terms of the Agreement, including the Working Capital Fund Fees, the fees for optional services, fees for health care, if any, and all other fees and charges payable monthly pursuant to the terms of the Agreement, as appropriate in the particular circumstances.

"Monthly Service Fee" means that monthly fee payable in consideration for the services provided to residents, as set forth under **Section 4.2** of the Agreement.

"Move-In Payment" has the meaning set forth in Section 1.3 of the Agreement.

"One-Time Capital Charge" means the one-time non-refundable charge set forth in **Section 3** of the Agreement.

"Operating Cash Requirements" has the meaning set forth in **Section 12.3.1** of the Agreement.

"Promissory Note" means the promissory note executed and delivered by Evergreen Woods to Resident pursuant to this Agreement in the form attached hereto as Exhibit A. Your rights to repayment of amounts due to you pursuant to the Promissory Note may be held by your heirs or others in trust established for your benefit if we consent to such an assignment, and our consent will not be unreasonably withheld.

"**Resident**" or "you" means the resident or residents who is (are) signatory to the Agreement, as identified in **Section 1** of the Agreement. Sometimes a second Resident (if there are two of you) is referred to in the Agreement as the "second person."

"The Community" means the facility known as Evergreen Woods, and which is the subject of the Agreement, including the apartments, The Health Center at Evergreen Woods, and all common areas.

"The Health Center at Evergreen Woods" means the health center forming a part of The Community that is licensed to provide nursing care.

"Working Capital Fund Fee" means that monthly fee payable to owner on a per contract basis for occupancy in The Community, as set forth in **Section 4.1** of the Agreement.

EVERGREEN WOODS

UNBUNDLED RESIDENCY AGREEMENT (60% Minimum Refund)

1. ENTRANCE FEE AND PROMISSORY NOTE.

1.1 Entrance Fee. To assure you, _____ ("Resident"), a place at Evergreen Woods in North Branford, Connecticut (hereinafter "The Community") in accordance with all terms of this Agreement, including payment of the Monthly Charges described below, you will pay to Evergreen Woods an Entrance Fee of <u>\$</u>______

1.2 Initial Deposit. You agree to pay ten percent (10%) of your Entrance Fee in the amount of \$ when you sign this Agreement. We will hold this amount in escrow until such time as it may be released pursuant to Connecticut law.

1.3 Balance of Entrance Fee. You agree to pay the remaining balance (ninety percent (90%)) of your Entrance Fee in the amount of $\underline{\$}$ (the "Move-In Payment") to Evergreen Woods on or before the date you move into The Community. You shall not be required to move into The Community before the expiration of thirty (30) days from your execution of this Agreement. In no event shall the date you move into The Community (or pay the Move-In Payment), extend beyond three (<u>3</u>) months from the date this Agreement is signed by both you and The Community.

1.4 Promissory Note. Upon receipt of the Move-In Payment, we will execute and deliver to you the Promissory Note substantially in the form attached to this Agreement as Exhibit A, and in the original principal amount of ninety percent (90%) of the Entrance Fee. Repayment of the Promissory Note will be made in accordance with its terms and the provisions of **Section 2.2** below. Your interest in the Promissory Note may not be mortgaged, sold, discounted, assigned, or otherwise transferred, except to certain trusts established for your benefit, subject to our prior approval and in our sole discretion.

2. **REIMBURSEMENT OF ENTRANCE FEE**.

PAGE 4 of 30

Payment of your Entrance Fee entitles you to live in the ______ Building, Apartment no. _____ ("Apartment") at The Community for as long as you are capable of meeting the requirements of apartment occupancy as set forth in our current entrance policy and to have priority of admission to the health care services of The Community all in accordance with the terms of this Agreement.

2.1 Reimbursement of Entrance Fee Prior to Occupancy.

2.1.1 If you are not accepted for residency, as evidenced by Community-authorized signature of this Residency Agreement, the full amount of the Entrance Fee that you have paid will be refunded to you without interest within five (5) business days.

2.1.2 You may cancel this Agreement at any time prior to occupying the Apartment for any reason. To cancel, you need to give us and the escrow agent, U.S. Bank, One Federal Street, 3rd Floor, Boston, MA 02110, written notice of cancellation by registered or certified mail and this Agreement will be canceled.

In such event, the full amount of the Entrance Fee that you have paid will be refunded to you without interest within five (5) business days of written notice, except that we will retain an amount equal to any costs specifically incurred by us or The Community at your request (any such costs will be in a written addendum to this Agreement signed by both you and us).

2.2 Repayment of Promissory Note After Occupancy. In the event this Agreement is terminated or canceled after you have occupied your Apartment, the unpaid principal balance of the Promissory Note shall be repaid as stated following:

- a. The unpaid principal balance of the Promissory Note shall become due and payable to you or your estate, as the case may be, in accordance with the terms set forth below only when: (i) this Agreement has terminated; (ii) you have surrendered possession of the Apartment to us as required by this Agreement; and (iii) either we have received the full Entrance Fee from a substitute resident for the Apartment or a substitute resident moves into the Apartment or in accordance with **Section 2.2g**.
- b. We will be entitled to credits which will automatically reduce the principal balance of the Promissory Note in the following amounts, effective on the following dates:

A credit equal to 10% of the total Entrance Fee will be deducted from the principal balance on the Promissory Note if, on the first anniversary of the date that the Move-In Payment was made, this Agreement has not been terminated or you or the second person remains a resident in the Apartment (i.e., we will then owe you an amount equal to 80% of the total Entrance Fee). An additional credit equal to 10% of the total Entrance Fee will be deducted from the principal balance on the Promissory Note if, on the second anniversary of the date that your Move-In Payment was made, this Agreement has not been terminated or you or the second person remains a resident in the Apartment (i.e., we will then owe you an amount equal to 70% of the total Entrance Fee).

A final credit equal to 10% of the total Entrance Fee will be deducted from the principal balance of the Promissory Note if, on the third anniversary of the date that your Move-In Payment was made, this Agreement has not been terminated or you or the second person remains a resident in the Apartment (i.e., we will then owe you an amount equal to 60% of the total Entrance Fee).

The total of the credits which will occur on the first, second, and third anniversary of the date that your Move-in Payment was made is equal to 30% of the total Entrance Fee. Because the Promissory Note will initially be for an amount equal to 90% of the total Entrance Fee, the principal balance owed on the Promissory Note will not fall below 60% of the total Entrance Fee (less any amount you owe us under this Agreement at the time or after it terminates if we have elected to setoff such amount).

- c. We agree to evaluate your financial condition, upon your request, and if need is demonstrated, to apply amounts due to you on the Promissory Note to amounts you would otherwise owe us for services in The Health Center at Evergreen Woods. Any amounts so applied will become a reduction in the unpaid principal balance of the Promissory Note (i.e., a "set-off").
- d. Upon payment to you of the unpaid principal balance due on the Promissory Note, you will execute all such documents and take all action necessary to cancel the Promissory Note.
- e. You will be responsible for payment of your own income taxes on any interest which accrues or is imputed on account of the Promissory Note pursuant to applicable state or federal law.
- f. If we have not received the Entrance Fee from a substitute resident for the Apartment or a substitute resident has not moved into the Apartment within two (2) years after this Agreement has terminated and you have

surrendered possession of the Apartment to us as required by this Agreement, then interest on the unpaid principal balance of the Promissory Note will accrue at the rate of four percent (4%) per annum and interest shall be paid to you monthly in arrears, starting on the tenth day of the month following the second anniversary of the termination date of this Agreement and continuing on the tenth day of each month thereafter until the Promissory Note is paid in full.

- g. If not sooner paid, the unpaid principal balance of the Promissory Note, plus unpaid accrued interest, shall be paid in full on the date which is three (3) years after this Agreement has terminated and you have surrendered possession of the Apartment to us as required by this Agreement.
- h. Refurbishment Costs: You agree that we may deduct from the repayment to you of the principal balance our costs to refurbish the Apartment to like new condition when you leave. At a minimum, we will almost certainly replace the floor coverings and repaint the Apartment. You will be responsible for other costs in addition to paint and floor coverings for damage you, your guests, or your pets have caused to the Apartment, and such other costs will be deducted from the repayment to you of the principal balance.
- i. Tobacco Smoke Damage: You agree that the Apartment is a nonsmoking residence at all times. All of our costs (at any time) to refurbish the Apartment to like new condition which are related to tobacco smoke may be deducted from the repayment to you of the principal balance. The costs to repair smoke damages are substantial and generally include multiple replacements (such as duct work, surface finishes, even ceilings). These costs are in addition to those of **Section 2.2h** above.
- j. Pet Damage: You agree that all of our costs (at any time) to refurbish the Apartment for damages caused by your pet(s) may be deducted from the repayment to you of the principal balance. These costs are in addition to those of **Section 2.2h** above.
- k. The provisions of this **Section 2.2** shall survive termination of this Agreement.

2.3 Right of Offset. We have the right to offset against any Entrance Fee refund or Promissory Note repayment any unpaid Monthly Charges owed by you, any unreimbursed health care expenses we have advanced on your behalf, any amounts deferred by us under **Section 10.2**, and any other sums owed by you to us.

3. ONE TIME CAPITAL CHARGE. On or before the date you move into The Community, you will pay us the amount of <u>\$</u> (the "One-Time Capital Charge"). The One-Time Capital Charge is a one-time, non-refundable charge that will be used to fund apartment refurbishment and other capital improvements as determined by us.

4. MONTHLY CHARGES.

4.1 Working Capital Fund Fee. You shall pay a Working Capital Fund Fee of seventy-five dollars (\$75) per month for occupying your Apartment or The Health Center at Evergreen Woods. If there are two of you, a single monthly Working Capital Fund Fee shall cover both residents. There shall be no increase in your Working Capital Fund Fee during the term of this Agreement. The Working Capital Fund Fee shall be our unrestricted property, and may be used by us for any purposes.

4.2 Monthly Service Fees. You shall pay a Monthly Service Fee for the services and features provided to all residents as set forth in **Section 5**. During the current calendar year, this initial Monthly Service Fee will be $\underline{\$}$ per month for one person and an additional $\underline{\$}$ per month if there are two (2) of you occupying the Apartment. We will give you advance written notice of not less than thirty (30) days before any change in the Monthly Service Fee is implemented. Any change in the Monthly Service Fee may be made at our discretion.

4.3 Monthly Charges. During the current calendar year, you will pay fixed monthly charges of <u>\$</u>, which represent the Working Capital Fund Fee and Monthly Service Fee. In addition to the Working Capital Fund Fee and the Monthly Service Fee, you will also be charged for any additional services you may utilize.

4.4 Payment. On the day you move into The Community or as otherwise provided in Section 1.3 of this Agreement, you shall pay a prorated portion of the Working Capital Fund Fee and Monthly Service Fee. Thereafter, such fixed monthly charges will be payable on the first day of each month in advance, and your

additional charges for optional services will be paid on the first day of each month for the optional services obtained during the preceding month.

4.5 Termination of Charges for Apartment. The Monthly Charges for your Apartment will terminate when you vacate the Apartment and remove your possessions from it. In the event there are two (2) of you who occupy an Apartment and one of you dies, the second person's fees will cease and the remaining person will continue to pay the first person's Monthly Service Fee and Working Capital Fund Fee. Further, if your surviving spouse is not a resident of The Community or a party to this Agreement, then this Agreement will terminate upon your death and any fees or refunds owed will be handled pursuant to the terms herein.

4.6 Late Payment Charge; Interest and Collection Costs. Evergreen Woods reserves the right to impose a late payment charge and to assess interest at the rate of one and one-half percent (1.5%) per month for all balances that remain unpaid ten (10) days after the date on which they are due. Any account balances, including late payment charges and interest, that remain unpaid when this Agreement is terminated shall become a lien against your assets or estate after deducting any refund owed under this Agreement. Evergreen Woods also reserves the right to charge you for the costs of collection, including reasonable attorney's fees.

5. SERVICES AND FEATURES PROVIDED TO ALL APARTMENT RESIDENTS.

Evergreen Woods will furnish at The Community, so long as you reside in an apartment therein, the following services and features, which are included in the Monthly Service Fee:

5.1 Dining services and meals in the dining room per the Flexible Dining Dollar Points program;

- 5.2 Water, sewer, air conditioning, heating and electricity;
- **5.3** Basic cable TV service;
- **5.4** Property taxes, insurance on building and structures;
- **5.5** Grounds keeping, landscaping and snow removal;
- **5.6** Security services;
- **5.7** Building janitorial and maintenance service;

PAGE 9 of 30

5.8 Weekly housekeeping service;

5.9 Weekly flat laundry service;

5.10 Planned activities - social, cultural and recreational, for those who wish to participate;

5.11 One reserved space for open parking;

5.12 Complete kitchen, including refrigerator, range with oven, microwave, garbage disposal, dishwasher and ductless hood fan;

5.13 Coin-free washers and dryers on each floor;

5.14 Scheduled local transportation;

5.15 Urgent call response system on a 24 hour basis;

5.16 Use of all common areas in The Community; and

5.17 Use of The Health Center at Evergreen Woods pursuant to the terms of this Agreement. Note that this is an Unbundled Residency Agreement and that health care (including skilled nursing services) are at an additional charge above the Monthly Service Fee.

6. ADDITIONAL SERVICES PROVIDED FOR AN EXTRA CHARGE.

Evergreen Woods will also make available at The Community, at your request, so long as you reside in an apartment at The Community, for an extra charge at the then prevailing rates, additional meals over those provided in consideration for the Monthly Service Fee, housekeeping, personalized maintenance services, delivered meal service, certain office services such as "fax" and copier, use of guest rooms, catering for private parties, assisted living services, beauty parlor/barber shop, a limited number of garages, personal transportation and certain other services, upon special arrangements.

7. THE HEALTH CENTER AT EVERGREEN WOODS.

If, in the opinion of Evergreen Woods' Medical Director, after consultation with you and your legally authorized representative, if applicable, you need nursing care in The Health Center at Evergreen Woods and if space is available, we will provide you with nursing care services to the extent authorized by our license from the State of Connecticut. Such nursing care services will be provided at an additional charge that is at the then applicable charges for skilled nursing care.

7.1 Release Apartment. If you are receiving nursing care in The Health Center at Evergreen Woods, you may choose to release your Apartment. When you release your Apartment for occupancy by someone else and you have removed your possessions from the Apartment, then the Monthly Service Fee charges for your Apartment will cease. You will pay for your nursing care monthly charges at the then applicable charges for skilled nursing care.

7.2 Retain Apartment. If you choose not to release your Apartment pursuant to **Section 7.1**, you will pay for your nursing care monthly charges at the then applicable charges for skilled nursing care, and in addition, you will continue to pay your Monthly Service Fee for your Apartment and the monthly Working Capital Fund Fee.

7.3 Return to Apartment. If you release your Apartment because you have moved to The Health Center at Evergreen Woods, and if later you are able, in the opinion of the Medical Director, or your attending physician with the concurrence of the Medical Director, to return to an apartment, we will provide you an apartment of the same type as your Apartment as soon as one becomes available. You will be permitted to remain in The Health Center at Evergreen Woods until such apartment becomes available. Upon reoccupying an apartment, your Monthly Charges will be based on the then current charges for that apartment.

7.4 Additional Health Services. We provide health services and supplies in The Health Center at Evergreen Woods such as: therapy, pharmaceutical and medical supplies, personal laundry, and rental of equipment. These services and supplies are not included in the nursing care daily charges and will be available for an extra charge.

7.5 Medical Director. We have designated a member in good standing of the New Haven County Medical Society to serve as Medical Director of The Health Center at Evergreen Woods. You will be at liberty to engage the services of the Medical Director or any other physician at your own expense. We will not be responsible for the cost of medical treatment by the Medical Director, or any other physician, nor will we be responsible for the cost of medicine, drugs, prescribed therapy, and other similar services and supplies. In the event we incur or advance costs for your medical treatment or for medicine, drugs, prescribed therapy, and other similar services and supplies, you will reimburse us for such costs.

7.6 Non-resident Use of The Health Center at Evergreen Woods. We offer nursing services in The Health Center at Evergreen Woods to qualified non-residents for a fee to the extent beds are available. Residents will be given priority to available beds.

8. DURATION OF YOUR RIGHT TO OCCUPY THE APARTMENT.

You can live in your Apartment for as long as you (or either of you) live, unless you (or both of you) require services that cannot be provided or accommodated in the Apartment, or this Agreement is terminated by you or by us. If, in the opinion of the Medical Director, your physical or mental health requires that nursing care be given, you will relocate to The Health Center at Evergreen Woods where Evergreen Woods is licensed to provide such care or to such other health care facility as you choose.

9. YOUR VOLUNTARY CANCELLATION RIGHTS AFTER OCCUPANCY.

After you have moved into The Community you may cancel this Agreement at any time for any reason upon at least ninety (90) days' written notice signed by you (or both of you if there are two (2) of you) and sent to us by registered or certified mail. If you give such notice, you will pay all applicable Monthly Charges and any other daily charges pursuant to the terms of this Agreement until the later of the expiration of such ninety (90) day period or your departure from The Community. The balance due on your Promissory Note will be repaid to you pursuant to its' terms.

Happiness Guarantee: You will have the right to terminate this Agreement within the first six (6) months after the earlier of when you move in or when you paid the Move-In Payment, by giving written notice to us and vacating the Apartment during that six (6) month period. In such event, we will refund the entire Entrance Fee (without interest) that you have paid (minus the cost to restore the Apartment to new condition and the cost for restoration for any damages you have made to the Apartment). This will be paid to you upon the earlier of the following to occur: (i) our receipt of the full Entrance Fee from a substitute resident, (ii) the date a substitute resident moves into the Apartment, or (iii) the third anniversary of the date this Agreement is terminated. This Happiness Guarantee shall not apply for reasons of death. This Happiness Guarantee shall not apply if you move directly from your Apartment to a healthcare facility or you move to a healthcare facility within six (6) months of the date you move out of Evergreen Woods.

10. OUR TERMINATION RIGHTS.

10.1 Just Cause. We will not terminate this Agreement except for just cause, which includes, but is not limited to, the following:

10.1.1 Except as set forth below, failure to pay any Monthly Charges or the Entrance Fee as provided herein;

10.1.2 Creation by you of a disturbance within The Community which in our judgment is detrimental to the health, safety, comfort or peaceful lodging of others;

10.1.3 You become infected with a dangerous and contagious disease or become mentally or emotionally disturbed, and the Medical Director determines that your condition is detrimental to the health, safety, or welfare of others and The Health Center at Evergreen Woods cannot care for you within the limits of our license from the State;

10.1.4 You do not comply with the terms of this Agreement or the published operating procedures, covenants, rules, regulations and policies now existing or later created or amended by us; or

10.1.5 You refuse medical treatment which in the opinion of the Medical Director is medically required for your health or the health or safety of others.

10.2 Limitations on Termination Rights for Financial Inability. If, after you have paid the Entrance Fee and the One-Time Capital Charge, you encounter financial difficulties making it impossible for you to pay the full Monthly Charges and other charges appropriate for the Apartment; then:

10.2.1 You shall be permitted to remain at The Community for ninety (90) days after the date of failure to pay, during which time you shall continue to pay a reduced Monthly Charge based on your current financial condition; and

10.2.2 Because it is and shall continue to be our declared policy to not terminate your residency solely by reason of your financial inability to pay the full Monthly Charges and other charges, you shall be permitted to remain at The Community at a reduced Monthly Charge based on your ability to pay for so long as you establish facts, satisfactory to us in our sole discretion, to justify deferral of the usual charges, and the deferral of such charges can, in our sole discretion, be granted without impairing our ability to operate on a sound financial basis. This provision

shall be rendered inoperative if you have impaired your ability to meet your financial obligations hereunder by making unapproved gifts or other transfers or by not maintaining Medicare Part A, Medicare Part B, Medicare Part D and/or adequate supplemental insurance. Any such charges deferred shall be offset against the refundable portion of your Entrance Fee when it is paid to you or your estate in accordance with the Promissory Note.

10.3 Notice of Termination. Prior to any termination of the Agreement by us, we will give you notice in writing of the reasons, and you will have thirty (30) days thereafter to correct the problem. If the problem is corrected within such thirty (30) day period, this Agreement shall not be then terminated. If the problem is not corrected within such thirty (30) day period, this Agreement will be terminated and you must vacate the Apartment, remove all your possessions, and leave The Community.

10.4 Emergency Termination. Notwithstanding the above, if the Medical Director determines that either the giving of notice or the lapse of time as above provided might be detrimental to the health, safety, or welfare of you or others, then such notice and/or waiting period prior to termination and relocation to a hospital or other appropriate facility shall not be required, and termination of this Agreement shall be deemed to have occurred when you are relocated. In such event, we are expressly authorized to transfer you to a hospital or other appropriate facility and will promptly notify your representative or your attending physician.

11. MISCELLANEOUS PROVISIONS WITH RESPECT TO YOUR APARTMENT.

11.1 Use of Apartment. The Apartment is for living only and shall not be used for carrying on any business or profession, nor in any manner in violation of zoning restrictions or any other municipal, state or federal law.

11.2 Occupancy of Apartment. Except as hereinafter provided, no person other than you (or both of you) may occupy the Apartment except with our express written approval. In the event that a second person who is not a party to this Agreement is accepted for residency under this Agreement at a time subsequent to the date hereof (said acceptance to be in accordance with admission policies governing all other admissions), you shall pay the then current additional Monthly Charges for second persons. If such second person does not meet the requirements for residency, such second person will not be permitted to occupy the Apartment for more than thirty (30) days (except with our express written approval), and you may cancel this Agreement as provided in **Section 9**. If such second person is already a

resident of The Community and the two of you would like to live in the same apartment under one residency agreement, you may cancel this Agreement as provided in **Section 9**.

11.3 Compliance Changes. We may effect changes in The Community at any time to meet the requirements of the law. You agree to temporarily relocate to other facilities provided by us without additional cost to you if it becomes necessary to vacate your Apartment in order to make such changes.

11.4 Furnishings. Furnishings within the Apartment will not be provided by us except to the extent provided in consideration for the Monthly Service Fee. Furnishings provided by you shall not interfere with the health, safety and general welfare of other residents or the staff of The Community. If removal of your furniture and other property is not accomplished upon termination of this Agreement or within thirty (30) days after your death (if there are two of you, the death of the survivor), then we may remove and store such furniture and other property at the expense of you or your estate. We may dispose of such furniture or other property not claimed within ninety (90) days after your death (if there are two of you, the death of the survivor) or termination of this Agreement.

11.5 Alterations by You. You may not undertake any alterations to your Apartment without our prior written approval. We reserve the right, upon your release of the Apartment because you have moved to The Health Center at Evergreen Woods, or upon termination or cancellation of this Agreement, to require that any alterations made by you be removed, and that the Apartment be restored to its original design at your expense. The cost of any such restoration may be withheld from the Promissory Note when it is paid to you or your estate in accordance with its terms.

11.6 Refurbishment. We will provide maintenance and repair to your Apartment for property, furnishings, and equipment owned or leased by Evergreen Woods. Other maintenance services may be available at an additional cost. Should you desire to have the Apartment refurbished in the future (i.e., new paint, floor coverings, wall coverings, counter-tops, etc.), such services may be available at an additional cost to you. You agree to pay for any necessary refurbishments caused by you or your guests that are beyond those which are customary and normal, including, but not limited to, the costs of cleaning and removal of stains, odors, and other damage caused by smoking in your apartment.

11.7 Emergency Entry. We may enter your Apartment should it be necessary to protect your health or safety or the health or safety of other residents.

12. CASH REQUIREMENTS ALLOCATION FORMULA.

The amount of your Monthly Service Fee is determined by application of the Cash Requirements Allocation Formula. The Cash Requirements Allocation Formula has been developed to equitably apportion the costs of operating The Community among its residents who are under an unbundled residency plan.

12.1 General. All projected Cash Requirements (including additions to reserves) necessary to provide accommodations and services to the residents of The Community (excluding residents of The Health Center at Evergreen Woods) are charged to The Community residents through the application of the Cash Requirements Allocation Formula. Generally, the projected Cash Requirements are based upon the costs incurred in connection with the operation of The Community for the preceding year with adjustments for inflation or deflation, occupancy rates, additions to reserves, and nonrecurring expenses, and reduced by projected revenues (such as fees for services, and certain investment income, etc.) for the upcoming year. Once determined, the projected Cash Requirements are then allocated among The Community residents based upon the size of the apartment that each resident occupies and the number of residents in The Community.

12.2 Benefits to Us. No portions of the costs included in the Cash Requirements Allocation Formula are paid to Evergreen Woods as fees. Our compensation under unbundled residency plans for the on-going ownership of The Community is comprised solely of the Entrance Fee, monthly Working Capital Fund Fee (which is fixed for you for the duration of your occupancy), interest earned on deposits and/or certain escrowed monies, the appreciation of The Community, if any, and other benefits generally associated with the ownership of real estate. In the event, however, that we loan funds to operations that would normally be funded through the Cash Requirements Allocation and/or fund any differences between operating revenues and operating expenses, we may, at our option, recoup such amounts funded through the Monthly Service Fees.

12.3 Cash Requirements. As set forth in this **Section 12.3**, your Monthly Service Fee will be adjusted upward or downward each year according to the Cash Requirements of The Community, as allocated to each apartment and all second persons. The "Cash Requirements" to be allocated shall be equal to:

12.3.1 Operating Cash Requirements for the preceding year (annualized if the actual period is less than a full operating year), which shall mean all recurring and nonrecurring costs, expenses and outlays incurred in connection with the operation and administration of The Community. "Operating Cash

Requirements" shall be determined by The Community management and shall include but not be limited to salaries and wages and all fringe benefits paid to employees of The Community, and other persons stationed at The Community, and other persons having day-to-day administrative, marketing and/or management responsibilities; payroll taxes; telephone; utilities; insurance premiums; license fees, occupational taxes, property taxes, sales taxes, permits, etc., associated with the operation of The Community; lease payments, if any; legal and accounting services; bad debts and other losses related to the operation; deductibles related to insurance claims; consulting fees for specialists such as dietitians and medical directors; food and supplies; replacement and repair of equipment and all common furniture and fixtures; individual capital improvements, replacements or additions costing less than fifty thousand dollars (\$50,000) and not covered by Section 12.3.2; all interior building elements such as carpet, doors, repainting, and glass damage; management fees plus costs of reasonable expenses incurred in performing management services for The Community, such as transportation, living expenses, telephone expense and photocopying expense; cost of maintenance and normal replacement of landscaping; principal and interest payments on all loans except loans from residents; and other similar items related to the operation of The Community. Operating Cash Requirements shall not include income taxes, dividends, or individual capital improvements, and other replacements or additions that cost in excess of fifty thousand dollars (\$50,000); plus

12.3.2 An amount to maintain or increase the reserve for replacement described in **Section 12.6** shall be equal to three percent (3%) of the prior year's Operating Cash Requirements. This reserve is in addition to any operating reserves required under Connecticut law. Deposits to this reserve and earnings from this reserve may only be used for improving or replacing capital items of The Community that cost over fifty thousand dollars (\$50,000); <u>plus</u>

12.3.3 An amount projected by us to account for expected changes from the previous year in services, apartment occupancy, marketing, remodeling or any other material factor; <u>plus</u>

12.3.4 A reasonable estimate for inflation or deflation in the cost of operating The Community for the coming fiscal year; <u>plus</u>

12.3.5 An amount to maintain or increase the reserve for operating cash established under **Section 12.6**. Deposits to this reserve may only be used for items related to the operation of The Community; <u>plus</u>

12.3.6 An amount to maintain or increase the amount of any reserve required by the laws of the State of Connecticut; <u>minus</u>

12.3.7 The projected net revenues to be received in the coming fiscal year from (i) The Health Center at Evergreen Woods' operations; (ii) extra fees received for the furnishing of additional services to the residents as provided in **Sections 6** and **7** of this Agreement; (iii) space rentals from such support services as barber/beauty shops; (iv) income from investment of reserves required by the laws of the State of Connecticut; and (v) nonrecurring operating income.

12.4 Allocation of Cash Requirements. Allocation of the sum of the Cash Requirements described in Section 12.3 to each individual resident's apartment and all second persons shall be accomplished by:

12.4.1 Multiplying the number of apartments in The Community of each size and the number of second person residents therein by the following factors and then summing such products:

Number of One-Bedroom Alcove (Type A) Apartments X 70.6 percent factor

Number of One-Bedroom Apartments (Type B) X 84.3 percent factor

Number of One-Bedroom Custom Apartments (Type C) X 86.3 percent factor

Number of One-Bedroom Deluxe Apartments (Type D) X 96.1 percent factor

Number of Two-Bedroom Apartments (Type E) X 100.0 percent factor

Number of Two-Bedroom Custom Apartments (Type F) X 102.0 percent factor

Number of Two-Bedroom Corner Apartments (Type FC) X 102.0 percent factor

Number of Two-Bedroom Deluxe Apartments (Type G) X 107.8 percent factor

Number of Two-Bedroom Deluxe Apartments (Type GL) X 107.8 percent factor

Number of Second Persons X 45.1 percent factor

To take into account the combination or addition of different apartments to The Community, or similar events that require the addition of new factors, factors may be added to or deleted from the above formula.

12.4.2 Multiplying the sum determined in **Section 12.4.1** above by a factor representing the projected apartment occupancy rate.

12.4.3 Dividing the sum of the Cash Requirements determined under Section 12.3 by the amount determined under Section 12.4.2 to arrive at the arithmetic quotient.

12.4.4 The Cash Requirements Allocation for each apartment size and for the second person shall be the product of multiplying the arithmetic quotient determined under **Section 12.4.3** by the factor assigned to each apartment and to the second person in **Section 12.4.1**. The amount of increase or decrease in the Monthly Service Fee will be the difference between the new Cash Requirements Allocation for each apartment size and for the second person and the applicable Cash Requirements Allocation for the preceding year, divided by twelve (12).

12.5 Determination and Implementation of Cash Requirements Allocation. At least thirty (30) days prior to the end of each fiscal year, we will notify you of the new Monthly Service Fee that will go into effect at the start of the next fiscal year.

12.6 Additional Reserves. Evergreen Woods has established reserves as follows: (a) A reserve for replacement (as described in Section 12.3.2); (b) an operating cash reserve (as described in Section 12.3.5); and (c) a reserve required by Connecticut law (as described in Section 12.3.6). The total amount of the operating cash reserves includes operating reserves required by Connecticut law. Any amounts necessary to maintain or increase such additional reserves shall, however, be included in your Cash Requirements Allocation. Copies of these reserve agreements with the bank are available upon request.

13. ARBITRATION.

Any dispute, claim or controversy of any kind between the parties arising out of, in connection with, or relating to this Agreement and any amendment hereof, or the breach hereof, shall be submitted to and determined by arbitration in North Branford, Connecticut in accordance with the commercial arbitration rules then obtaining of the American Arbitration Association. All parties shall be bound by the arbitrator's decision and judgment upon such decision may be entered in any federal or state court having jurisdiction unless the arbitration is fraudulent or so grossly erroneous as to necessarily imply bad faith. If the parties are unable to agree in good faith and within a reasonable time on the selection of an arbitrator, any party may request appointment of an arbitrator by the American Arbitration Association. The parties shall make a reasonable attempt to select an arbitrator with experience in retirement communities, long-term care or health care operations generally. Cost of arbitration shall be shared by the parties equally. If the issue affects more than one resident, we may elect to join all affected residents into a single arbitration proceeding, and you hereby consent to such joinder.

14. AMENDMENTS.

14.1 This Agreement. Subject to **Section 14.2**, this Agreement may be amended only by written agreement of the parties to this Agreement.

14.2 Compliance with Laws. This Agreement may be modified by us at any time in order to comply with laws and regulations upon thirty days' notice to you.

15. MISCELLANEOUS LEGAL PROVISIONS.

15.1 Governing Law. This Agreement will be interpreted according to the laws of the State of Connecticut and will become effective upon acceptance and execution by us. The Glossary which sets forth the definitions of certain terms used in this Agreement is by this reference incorporated herein and made a part of this Agreement.

15.2 Separability. The invalidity of any restriction, condition or other provision of this Agreement, or any part of the same, shall not impair or affect in any way the validity or enforceability of the rest of this Agreement.

15.3 Capacity. This Agreement has been executed on our behalf by our duly authorized agent, and no member, partner, officer, director, agent or employee of Evergreen Woods shall have any personal liability hereunder to Resident under any circumstances.

15.4 Residents. When Resident consists of more than one person, the rights and obligations of each are joint and several, except as the context otherwise requires.

15.5 Resident's Rights. You understand and agree that (a) this Agreement or your rights (including the use of the Apartment) under it may not be assigned and no rights or benefits under this Agreement shall inure to the benefit of your heirs,

legatees, assignees, or representatives, except as to reimbursement of the amounts as described in **Section 2**; (b) this Agreement and your contractual right to occupy the Apartment shall exist and continue to exist during your lifetime unless canceled by you or terminated by us as provided herein; (c) this Agreement grants you a revocable license to occupy and use space in The Community but does not give you exclusive possession of the Apartment as against us and you shall not be entitled to any rights of specific performance but shall be limited to such remedies as set forth herein; (d) this Agreement is not a lease or easement and does not transfer or grant you any interest in real property owned by us; and (e) this Agreement grants to us complete decision making authority regarding the management and operation of The Community.

15.6 Release. We are not responsible for loss of or damage to your personal property. You may want to obtain, at your own expense, insurance to protect against such losses. We strongly suggest you keep and maintain your own insurance for liability and property loss.

15.7 Indemnity. We shall not be liable for, and you agree to indemnify, defend and hold us harmless from claims, damages, settlements, and expenses, including attorney's fees and court costs resulting from any injury or death to persons and any damages to property caused by, resulting from, attributable to or in any way connected with your negligent or intentional act or omission.

15.8 Entire Agreement. This Agreement and any Addenda or Exhibits express the entire agreement between the parties hereto and supersede any prior or contemporaneous written or oral understanding or agreement.

15.9 Reimbursement of Loss or Damage. You or your representative, if applicable, will reimburse us for any loss or damage to The Community as the result of your intentional, careless, or negligent acts or omissions or that of your guests. You agree to obtain insurance for you and your property and agree to provide us with proof of such insurance upon request.

15.10 Subordination. Except for your right of cancellation prior to occupancy, you agree that all your rights under this Agreement will always be subordinate and junior to the lien of all indentures of trust, mortgages or other documents creating liens encumbering The Community or any of the assets of Evergreen Woods, which have been or will be executed by us. Upon request, you agree to sign, acknowledge and deliver to such lender or lenders such further written evidence of such subordination as such lenders may reasonably require. You will not be liable for any such indebtedness.

15.11 Transfers. We may from time to time issue additional equity interests, or sell The Community, provided that in such latter event the buyer shall agree to assume all existing residency agreements. Your signature hereto constitutes your consent and approval to any such future transfer.

15.12 Law Changes. If changes are made in any of the statutes or regulations applicable to this Agreement, we shall have the right to submit to you a revised Agreement based on the changes in the law, and you agree to accept any such revisions that do not adversely affect you.

15.13 Private Employee of Resident. If you need additional services, you can obtain these needed services from a private employee (privately employed personnel). In such instances, you are encouraged to obtain these needed services from a licensed and/or certified home health agency. Further, you must comply with our policy regarding privately employed personnel and ensure that your private employee complies with our policies and rules of conduct as set forth by us. If you fail to follow the reasonable rules set forth by us, which will be provided to you prior to the time you sign this Agreement, and thereafter in the event of revisions or additions, then we may elect at our sole option to terminate this Agreement in accordance with **Section 10**.

15.14 Responsible Party. You agree to execute and deliver to us within sixty (60) days after assuming occupancy in your Apartment a Durable Power of Attorney, trust documents, or other documentation naming a responsible party for business and financial decision making. You also agree to provide us a copy of, or execute and deliver within the same sixty (60) day time frame, an Appointment of Health Care Representative or a valid Durable Power of Attorney for Health Care Decisions. These documents should be drafted to remain effective notwithstanding your incompetence or disability and shall be in a form acceptable to us, and you agree to keep such documents in effect as long as this Agreement is in effect. The person(s) named as your responsible party shall not be a person(s) employed by us or any other entity engaged in the management of The Community.

15.15 Residents' Association. Residents shall have the right to organize and operate a residents' association at The Community and to meet privately to conduct business of the residents' association.

15.16 Rights of Third Parties. Only parties who have executed this Residency Agreement will have any rights hereunder. Neither this Residency Agreement nor the disclosure statement of Evergreen Woods create any rights for third parties other than those who have executed this Residency Agreement.

15.17 Estate Planning. Evergreen Woods will pay any amounts due pursuant to the Promissory Note to your estate upon your death, or in the case of a double-occupied apartment, upon the death of the surviving resident, in accordance with the repayment terms of the Promissory Note. If you do not wish to have the amounts due pursuant to the Promissory Note paid to your estate, you may choose to assign your right to the Promissory Note to a trust or other person designated by you pursuant to a form of Assignment of Promissory Note Proceeds approved by Evergreen Woods. In the absence of any agreement between the residents of a double occupied apartment which has been provided to Evergreen Woods prior to any repayment due on the Promissory Note to the estate of the surviving resident. No other rights under this Residency Agreement are assignable.

15.18 Notices. All written notices required by this Agreement will be sufficient if addressed; if to you, to your Apartment at Evergreen Woods; if to Evergreen Woods, to the attention of the Executive Director, Evergreen Woods, 88 Notch Hill Road, North Branford, CT 06471.

15.19 Compliance with Laws. You will be afforded all rights and privileges under Conn. Gen. Stat. § 17b-520 <u>et seq</u>., which governs continuing care facilities, and all other applicable laws. Evergreen Woods will comply with all municipal, state and federal laws and regulations regarding consumer protection and protection from financial exploitation.

15.20 Managed Residential Community Residents' Bill of Rights. Evergreen Woods is registered as a "managed residential community" with the Connecticut Department of Public Health. Accordingly, Evergreen Woods is required to provide and explain to you, at the time of execution of this Agreement, the Managed Residential Community Residents' Bill of Rights attached hereto as Exhibit B (the "Bill of Rights"). A staff member will explain the Bill of Rights to you, and you will be asked to separately sign the Bill of Rights to acknowledge that you have received it.

15.21 Rules and Regulations. Evergreen Woods shall have the right to adopt, or amend, either by itself or with or through the residents' council, such reasonable rules and regulations as it deems necessary or desirable for the proper management and operation of the Community and for the safety, health and comfort of the residents. The rules and regulations in effect at the time of execution of this Agreement are set forth in the Resident Handbook. You agree to abide by such rules and regulations, as they may be amended from time to time. The Resident Handbook

includes procedures for you to address any concerns or complaints you may have during your residency.

16. RESERVE POLICY.

It will be our policy to maintain reserve funds as required by Connecticut law.

17. RESIDENT REPRESENTATIONS/OBLIGATIONS.

By executing this Agreement you represent and warrant that you are at least sixty-two (62) years of age or older at the time of admission, that you are capable of meeting the requirements of admission as set forth in our current admission policy, that you have assets and income which are sufficient under foreseeable circumstances and after provision for payment of your obligations under this Agreement to meet your ordinary and customary living expenses after assuming occupancy, and that all written representations made with respect to such matters by you or on your behalf to us are true. From time to time, Evergreen Woods may request that you update the financial information you submitted prior to admission. As a condition of continued residency in The Community, you agree to provide such financial information to Evergreen Woods when requested. You agree not to impair your ability, in any manner described in **Section 10.2.2** of this Agreement, to meet your financial obligations under this Agreement.

18. ACKNOWLEDGEMENT OF RECEIPT OF DOCUMENTS.

Under the requirements of applicable law, we are obligated to remind you that:

(1) this Agreement, which is a continuing care contract, is a financial investment and your investment may be at risk;

(2) our ability to meet our contractual obligations under this Agreement depends on our financial performance;

(3) you have been advised to consult an attorney or other professional experienced in matters relating to investments in continuing care facilities before you signed this Agreement; and

(4) the Department of Social Services of the State of Connecticut does not guarantee the security of your investment.

BALANCE OF PAGE LEFT BLANK. SIGNATURE PAGE FOLLOWS.

You hereby certify that you received a copy of this Agreement and a copy of our latest disclosure statement on or before the date hereof, and have been permitted to inspect any additional relevant materials requested to be reviewed by you or your representatives prior to signing this Agreement.

Executed at,,	, this day of
RESIDENT	EVERGREEN WOODS RETIREMENT LLC d/b/a Evergreen Woods
Witness	
RESIDENT	By: Executive Director
Witness	Date:
Date:	
Apartment Number:	Building Name:
Scheduled Move-In Date:	
Entrance Fee: \$	One Time Capital Charge: \$
1 Person Monthly Service Fee: \$	(plus \$75 Working Capital charge)
2 nd Person addition to Monthly Serv	ice Fee: \$

EQUAL HOUSING

EXHIBIT A (TO BE EXECUTED AT OCCUPANCY)

PROMISSORY NOTE

\$

North Branford, Connecticut

FOR VALUE RECEIVED, the undersigned ("Maker") promises to pay to ______ ("Holder") or order, the principal sum of ______ Dollars (\$_____) payable in lawful money of the United States of America, at such place as the legal holder hereof may designate in writing, subject to the following.

This Note is given to evidence certain obligations of Maker pursuant to an Residency Agreement dated as of ______, ____ between the undersigned and Holder regarding residency at Evergreen Woods (the "Residency Agreement") and is subject to the terms and conditions of the Residency Agreement, which are incorporated into and made a part of this Note by this reference.

Except as provided in Section 2.2 of the Residency Agreement, no interest shall accrue on the outstanding principal balance of this Note. Interest on this Note shall accrue as provided in Section 2.2 of the Residency Agreement. Maker shall make monthly payments of accrued interest, if any, on the tenth day of each calendar month.

The principal amount of indebtedness evidenced by this Note shall be automatically reduced in accordance with Section 2.2 of the Residency Agreement. If not sooner paid, the entire indebtedness, less a credit for any amount owed by Holder to Maker under the Residency Agreement, shall be due and payable on the date described in Section 2.2 of the Residency Agreement.

This Note may be prepaid in full or in part at any time without penalty or premium.

In the event of default in payment of any installment of principal or interest hereof as the same becomes due and such default is not cured within ten (10) days after written notice to Maker, then the Holder of this Note may without further notice, declare the remainder of the principal sum, together with all interest accrued thereon, at once due and payable. Failure to exercise this option shall not constitute a waiver of the right to exercise the same at any other time. The unpaid principal of this Note and any part thereof, accrued interest and all other sums due under this Note shall bear interest at the rate of twelve percent (12%) per annum after default until paid. All parties to this Note, including Maker and any sureties, endorsers, or guarantors hereby waive protest, presentment, notice of dishonor, and notice of acceleration of maturity and agree to continue to remain bound for the payment of principal, interest and all other sums due under this Note notwithstanding any extension or extensions of time for the payment of principal and interest; and all such parties waive all and every kind of notice of such change or changes and agree that the same may be made without notice or consent of any of them.

Upon default the Holder of this Note may employ an attorney to enforce the Holder's rights and remedies and the Maker, principal, surety, guarantor and endorsers of this Note hereby agree to pay to the Holder reasonable attorney's fees not exceeding a sum equal to fifteen percent (15%) of the outstanding balance owing on this Note, plus all other reasonable expenses incurred by the Holder in exercising any of the Holder's rights and remedies upon default. The rights and remedies of the Holder as provided in this Note shall be cumulative and may be pursued singly, successively, or together, in the sole discretion of the Holder. The failure to exercise any such right or remedy shall not be a waiver or release of such rights or remedies or the right to exercise any of them at another time.

Maker holds a leasehold interest in and operates Evergreen Woods. If Maker, (i) after obtaining the prior written consent of the owner of the fee interest in Evergreen Woods, assigns its leasehold interest in Evergreen Woods and (ii) the assignee assumes in writing all of Maker's obligations under the Residency Agreement and this Note, then Maker shall be released from all obligations under this Note and the term "Maker" shall thereafter mean such assignee.

This Note is to be governed and construed in accordance with the laws of the State of Connecticut.

This Note is given to secure a debt.

IN TESTIMONY WHEREOF, Maker has executed this Note under seal as of the day and year first above written.

EVERGREEN WOODS RETIREMENT, LLC, a Connecticut limited liability company (SEAL)				
By:	Senior Living Communities, LLC Manager	(SEAL)		
By:	Title:	(SEAL)		
PAGE	27 of 30			

EXHIBIT B



MANAGED RESIDENTIAL COMMUNITY RESIDENTS' BILL OF RIGHTS

You have the right to:

- Live in a clean, safe and habitable private residential unit;
- Be treated with consideration, respect and due recognition of your personal dignity, individuality and the need for privacy;
- Privacy within your private residential unit, subject to The Community's rules reasonably designed to promote your health, safety and welfare;
- Retain and use your personal property within your apartment so as to maintain individuality and personal dignity, provided that your use of personal property does not infringe on the rights of other residents or threaten the health, safety and welfare of other residents;
- Private communications, including receiving and sending unopened correspondence, telephone access and visiting with persons of your choice;
- Freedom to participate in and benefit from community services and activities so as to achieve the highest possible level of independence, autonomy and interaction within The Community;
- Directly engage or contract with licensed health care professionals and providers of your choice to obtain necessary health care services in your private apartment, or such other space in The Community as may be available to residents for such purposes;
- Manage your own financial affairs;
- Exercise civil and religious liberties;
- Present grievances and recommend changes in policies, procedures and services to our Executive Director or staff, government officials or any

other person without restraint, interference, coercion, discrimination or reprisal from The Community, including access to representatives of the Department of Public Health or the Office of the Long-Term Care Ombudsman;

- Upon request, obtain the name of the Executive Director or any other persons responsible for resident care or the coordination of resident care;
- Confidential treatment of all records and communications to the extent required by state and federal law;
- Have all reasonable requests responded to promptly and adequately within the capacity of The Community and with due consideration given to the rights of other residents;
- Be fully advised of the relationship that The Community has with any Assisted Living Services Agency, health care facility or educational institution to the extent that such relationship relates to resident medical care or treatment and to receive an explanation about the relationship;
- Receive a copy of any rules or regulations of The Community;
- Privacy when receiving medical treatment or other services within the capacity of The Community;
- Refuse care and treatment and participate in the planning for the care and services you need or receive, provided, however, that your refusal of care and treatment may preclude you from being able to continue to reside in The Community; and
- If you are a continuing care resident, all rights afforded under Conn. Gen. Stat. § 17b-520 <u>et seq</u>. and any other applicable laws. If you rent your apartment, all rights and privileges afforded to tenants under title 47a of the Connecticut General Statutes (Connecticut's landlord tenant laws).

We hope that any complaints or concerns that you have can be resolved by our staff. However, you also have the right to contact the following state agencies regarding complaints or concerns:

Department of Public Health Facilities Licensing and Investigations 410 Capitol Ave., P.O. Box 340308 MS#12HSR Hartford, CT 06134-0308 Information/General: Loan Nguyen, R.N. Supervising Nurse Consultant (860) 509-7400

Office of the Long Term Care Ombudsman Dan Lerman 370 James Street Suite 306 New Haven, CT 06513 (860) 823-3366

If you are receiving nursing or personal care from an Assisted Living Services Agency, you also have other rights set forth separately in the Assisted Living Clients' Bill of Rights.

Please sign below to acknowledge that we have provided you with a copy of the Managed Residential Community Residents' Bill of Rights and explained them to you.

Date

Resident

Resident's Representative

Relationship to Resident

EXHIBIT C

SWORN STATEMENT OF ENTRANCE FEE ESCROW AGENT

See attached.

bank

The undersigned, U.S. Bank National Association, hereby affirms and swears that it serves as Escrow Agent under the Entrance Fee Escrow Agreement (Phase I) of Shoreline Care Limited Partnership dated January 26, 1990 (the "Entrance Fee Escrow Agreement") between Shoreline Care Limited Partnership, a Connecticut limited partnership, succeeded by Shoreline Life Care LLC, a Connecticut limited liability company (the "Assignor") and Union Trust Company, succeeded by U.S. Bank National Association, as escrow agent (the "Escrow Agent"). Attached is a copy of the Entrance Fee Escrow Agreement and the Assignment and Assumption Agreement between the Assignor and Evergreen Woods Retirement, LLC (the "Assignee") pursuant to which the Assignor assigns all of its right, title and interest in, and all of its obligations as "Provider" under, the Entrance Fee Escrow Agreement and the Assignee assumes and agrees to perform fully all of Assignor's covenants, promises, agreements, obligations and liabilities as "Provider" under the Entrance Fee Escrow Agreement, effective upon the change of ownership of the Evergreen Woods retirement community located in North Branford, Connecticut.

IN WITNESS WHEREOF, the undersigned Escrow Agent has caused this statement to be executed by its duly authorized officer on this 9th day of September, 2019.

U.S. Bank National Association, as Escrow Agent

By: Name:

Title:

David W. Doucette Vice President

STATE OF MASSACHUSETTS

COUNTY OF SUFFOLK

On this 9th day of September, 2019, personally appeared, David W. Doucette, an officer of U.S. Bank National Association, known to me (or satisfactorily proven) to be the person whose name is subscribed to within this document and acknowledged that he/she executed the same for the purposes therein contained.

Notary Public

In witness whereof I hereunto set my hand.

(Notary Seal or Stamp)



My commission expires:

Mar-26-2007 11:04am

9572000189 ACCT

TYPE:

350 INIT ENTRANCE FEE ESCROW AGREEMENT (PHASE I)

of

+97046589

SHORELINE CARE LIMITED PARTNERSHIP

This Entrance Fee Escrow Agreement is made as of the <u>26</u> day of <u>JANUARY</u>, 19**59** by and between SHORELINE CARE LIMITED PARTNERSHIP c/b/a EVERGREEN WOODS ("Provider"), a Connecticut limited partnership with an address at <u>SECM ST</u>, <u>NEW HAVEN</u>, Connecticut, and UNION TRUST COMPANY, a banking institution with its principal place of business in the State of Connecticut ("Escrow Agent").

Recitals:

WHEREAS, Provider is the sponce, of a continuing-care retirement community ("Evergreen Woods") to be located in North Branford, Connecticut, and as such is a "provider" within the meaning of Section 1(d) of Connecticut Public Act No. 86-252 (the "Act"), codified as Sections 17-535 through 17-550 of the Connecticut General Statutes; and

WHEREAS, in order to comply with Section 2 and Section 5 of the Act and to protect residents or prospective residents of Evergreen Woods, the parties desire to enter into this Entrance Fee Escrow Agreement (the "Agreement") setting forth the terms under which certain deposits will be made with the Escrow Agent for the purpose of maintaining an escrow account for entrance fee payments in compliance with the Act. Attached hereto as <u>Exhibit "A</u>" and incorporated herein by reference is a copy of the proposed form of continuing-care contract (the "Residency Agreement") as such may be amended by Provider from time to time, which will be entered into between Provider and each prospective resident of Evergreen Woods (a "Resident").

NOW THEREFORE, in consideration of these premises and the mutual covenants herein set forth, the parties agree as follows:

1. <u>Deposits</u>. In accordance with Section 1 of the Residency Agreement, Provider expects to receive from each Resident a payment of that Resident's Entrance Fee (as such term is described in the Residency Agreement) at the time of execution of the Residency Agreement and to receive from each **Ъ**,

Resident a payment of that Resident's Loan (as such term is described in the Residency Agreement) on or before the date Resident begins to reside at Evergreen Woods. On or after the date of this Agreement, Provider will deposit with the Escrow Agent, within seventy-two (72) hours of receipt thereof by Provider, all Entrance Fees received by Provider in connection with Evergreen Woods (Phase I) together with a letter substantially in the form of FORM 1 attached hereto, including a copy of the signature page of each Residency Agreement signed by a Resident.

2. <u>Escrow Account</u>. The Escrow Agent shall maintain all such deposits made to the Escrow Account separate and apart from any other funds of Resident or of Provider.

The Escrow Agent shall invest all amounts <u>investments</u>. held by it from time to time in such obligations of the United States or its agencies (or in shares of a registered investment company which holds a portfolio of such securities) or such certificates of deposit, savings accounts, or other savings or investment securities as Provider may from time to time direct in writing. No amounts held by the Escrow Agent pursuant to this Agreement may be invested in any building or health-care facility of any kind, or used for capital construction or improvements or for the purchase of real estate or removed from the State of Connecticut if required to be maintained within such State. No amounts held by the Escrow Agent pursuant to this Agreement shall be pledged as collateral or subordinated to other loans or commitments of Evergreen Woods, other than first mortgage loans or other long-term financing obligations of Evergreen Woods.

4. <u>Reports and Certificates</u>. The Escrow Agent agrees to issue periodic statements indicating the status of the Escrow Account maintained by the Escrow Agent pursuant to this Agreement. In any event, Escrow Agent agrees to provide monthly reports to the Provider, which include a summary of all additions to, disbursements from and interest on the Escrow Account during the current period, and a detail of the investments comprising the balance in the Escrow Account as of the end of the current period.

5. <u>Disbursements</u>. The Escrow Agent shall release any amounts then held by it in the Escrow Account in accordance with the following terms and subject to Paragraph 5.5 hereof:

+6176036683

5.1 Disbursement When Resident Rescinds Within Thirty (30) Davs. Upon receipt by the Escrow Agent of written notice from a Resident by registered or certified mail, pertaining to the election by the Resident to rescind his or her obligations under the Residency Agreement prior to occupancy, the Escrow Agent shall immediately notify the Provider of such request (such notice to be given by telephone and confirmed in a writing mailed to Provider at its address first set forth above) and, within five (5) business days of its receipt of a written confirmation of release of Provider (see FORM 2), release to the Resident, such portion of the Entrance Fee paid by such Resident as is required by his or her Residency Agreement, and shall remit the balance of such Entrance Fee, if any, to the Provider. The Escrew Agent shall promptly provide Provider with written notice of any such refund. Any Entrance Fees refundable to Resident pursuant to this Paragraph 5.1, are hereinafter referred to as "Refundable Deposits."

5.2 Disbursements When Resident Cancels For Any Other Reason. In the event that the Apartment at Evergreen Woods is not available for occupancy by the Resident within three (3) years after the date of execution of his or her Residency Agreement, or under any other circumstances permitting a refund of all or any portion of the Entrance Fee, the Escrow Agent shall, upon receipt of written notice thereof from the Provider certifying that such refund shall be made, return forthwith to the Resident such portion of the Entrance Fee paid by such Resident as is required by his or her Residency Agreement and shall remit the balance of such Entrance Fee, if any, to the Provider.

5.3 Disbursements to Additional Reserve. The Escrow Agent shall release amounts to the additional reserve described in Paragraph 6 below, upon receipt by the Escrow Agent of written notice from Provider substantially in the form of FORM 4 attached hereto which certifies that all conditions required by the Act have been met to permit such release, and sets forth the amount to be released to such additional reserve. Upon receipt of such notice, the Escrow Agent shall forthwith release such amount of the aggregate Entrance Fees then deposited with the Escrow Agent for deposit into such additional reserve and the balance, if any, shall be released to the Provider as described in Paragraph 5.4.

-3-

Mar-26-2007 11:05am

From-US BANK CORPORATE TRUST SERVICES

+6176036683

5.4 <u>Disbursements to Provider</u>. The Escrow Agent shall release amounts to the Provider in accordance with the following terms:

> (a) With respect to a previously occupied living unit, upon receipt by the Escrow Agent of written notice from Provider stating that the previously occupied living unit to which the Entrance Fee(s) relates is available for occupancy by the Resident, the Escrow Agent shall forthwith release to the Provider the balance of any Entrance Fee(s) then deposited with the Escrow Agent pursuant to this Agreement for that Resident.

(b) With respect to a living unit which has not been previously occupied, upon receipt by the Escrow Agent of written notice from the Provider certifying that the following conditions have been met:

(i) The sum of the Entrance Fees received or receivable by the Provider pursuant to the Residency Agreements, plus the anticipated proceeds of any first mortgage loan or other long-term financing commitment, plus funds from other sources in the actual possession of the Provider, equals or exceeds the sum of seventy-five per cent of the aggregate cost of constructing or purchasing, equipping and furnishing Evergreen Woods plus seventy-five per cent of the funds estimated in the statement of anticipated source and application of funds submitted by the Provider as part of its disclosure statement on file with the Department of Aging (the "Disclosure Statement") to be necessary to fund start-up loses of Evergreen Woods plus seventy-five per cent of the amount of the reserve fund escrow required to be maintained by the Provider;

(ii) A commitment has been received by the Provider for any permanent mortgage loan or other long-term financing described in the statement of anticipated source and application of funds included in the Disclosure Statement and any conditions of the commitment prior to disbursement of funds thereunder, other than completion of the construction of Evergreen Woods, have been substantially satisfied; and

--<u>+</u>--

0 200

(iii) If construction of Evergreen Woods has not been substantially completed: all governmental permits or approvals necessary prior to the commencement of construction have been obtained; a maximum price contract has been entered into between the Provider and a general contractor responsible for construction of Evergreen Woods; a bond covering the faithful performance of the construction contract by the general contractor and the payment of all obligations arising thereunder has been issued by an insurer authorized to do business in the State of Connecticut with the Provider as obligee; a loan agreement has been entered into by the Provider for an interim construction loan in an amount, when combined with the amount of Entrance Fees then held in escrow under this Agreement plus the amount of funds from other sources (then in the actual possession of the Provider, that will equal or exceed the estimated cost of constructing, equipping and furnishing Evergreen Woods; not less than ten per cent of the amount of the construction loan has been disbursed by the lender for physical construction or site preparation work completed; and orders at firm prices have been placed by the Provider for not less than fifty per cent in value, including installation charges if applicable, of items necessary for equipping and furnishing Evergreen Woods in accordance with the description set forth in the Disclosure Statement; or if construction or purchase of Evergreen Woods has been substantially completed, an occupancy permit covering the living unit has been issued by the local government having authority to issue these permits;

1/28/08 the Escrow Agent shall release to the Provider the balance of any Entrance Fees then deposited with the Escrow Agent less any Refundable Deposits then being held by the Escrow Agent. aggregate amount of Entrance Fees which may be released to the Provider hereunder prior to the date on which any reserve fund escrow that is required to be established pursuant to the Act is established, shall not exceed the aggregate amount of Entrance Fees then received or receivable by the Provider pursuant to binding Residency Agreements less the amount of the WWW be initially maintained in such reserve fund escrow. Entrance Fees received or receivable which may be required to

-5-

Mar-26-2007 11:05am

4/1/05 RAR 6/16/09

(c) Upon receipt by the Escrow Agent of written notice from Provider, the Escrow Agent shall forthwith release to the Provider the aggregate of all interest earned on the Escrow Account, or such portion thereof as Provider desires released.

5.5 Written Confirmation by Provider of Disbursements. In any case in which the Provider believes that funds are required to be disbursed by the Escrow Agent hereunder, the Provider shall promptly give written notice to the Escrew Agent of the Provider's calculation of the amount(s) to be disbursed to Resident and/or Provider, as the case may be (refer to Paragraphs 5.2, 5.3, and 5.4 and FORM 3). In any case in which a Resident has notified the Escrow Agent that Escrowed Funds should be released to such Resident: (i) the Escrow Agent shall immediately notify the Provider of such request (such notice to be given by telephone and confirmed in a writing mailed to Provider at its address first set forth above); and (ii) the Escrow Agent shall not disburse any such funds except in accordance with written confirmation of the Provider, which confirmation shall be given by written notice substantially in the form attached hereto as FORM 2 (refer to Paragraph 5.1).

5.6 <u>Notice to Residents</u>. The Provider shall provide each prospective Resident of Evergreen Woods who has signed a Residency Agreement with the name, address, and telephone number of the Escrow Agent and shall file a copy of this Agreement with the Department of Aging.

6. <u>Additional Reserve.</u> To protect Residents and prospective Residents of Evergreen Woods and to comply with Section 6 of the Act, the Provider will establish the following additional reserve:

<u>Operating Reserve.</u> The Provider will establish an Operating Reserve, to be administered pursuant to a separate agreement. Such Operating Reserve shall be funded from the Escrow Account in accordance with Paragraph 5.3 above. If the Provider has already funded the Operating Reserve as required by Connecticut law, any amounts released from the Escrow Account which would have so funded the Operating Reserve shall be paid to the Provider to such extent.

7. <u>Duties</u>. The Escrow Agent undertakes to perform only such duties as are expressly set forth herein and no

implied duties or obligations shall be inferred or otherwise imposed upon or against the Escrow Agent. The Escrow Agent may act in reliance upon any writing or instrument under signature which it, in good faith, believes to be genuine; may assume the validity and accuracy of any statement or assertion contained in any such writing or instrument; and may assume that any person purporting to give any writing, notice, advice, or instruction in connection with the provisions hereof has been duly authorized to do so, unless Escrow Agent has reason to believe otherwise. The Escrow Agent shall not be liable in any manner for the sufficiency or correctness as to form or manner of execution or validity of any instrument deposited with or delivered to the Escrow Agent hereunder. The duties of the Escrow Agent hereunder shall be limited to the safekeeping of the Deposits and other money, instruments, or other documents received by it as Escrow Agent and for its disposition of the same in accordance with the written instructions accepted by it as Escrow Agent. Each and every determination, payment, action or duty to be taken by the Escrow Agent under this Agreement shall be taken based solely on the written direction from the ... Provider and, in the case of disbursements made pursuant to Paragraph 5.1, from a Resident. The Escrow Agent will be fully released from any liability hereunder if it acts in accordance with such written direction.

5. Indemnity. Provider hereby agrees to indemnify the Escrow Agent and hold it harmless from any and all claims, liabilities, losses, actions, suits, or proceedings at law or in equity, or any other expense, fees, or charges in equity, or other expenses, fees, or charges of any character or nature, which it may incur or with which it may be threatened by reason of its acting as Escrow Agent under this Agreement, unless caused by the Escrow Agent's willful misconduct or gross negligence; and in connection therewith, to indemnify the Escrow Agent against any and all expenses, including attorney's fees and the cost of defending any action, suit, or proceeding or resisting any claim. Any and all expenses incurred by the Escrow Agent bereunder shall be reimbursed by the Provider upon demand.

9. <u>Termination</u>. Either party may terminate this Agreement upon thirty (30) days' written notice thereof to the other party, subject to the appointment of its successor by Provider within such thirty (30) day period. Provider shall give prompt written notice to the Escrow Agent of the name and address of the successor escrow agent. If a successor escrow agent is not appointed by Florider within the thirty (30) duy period, the Escrow Agent may petition the Court to appoint a successor. Upon the appointment of such successor escrow agent, the Escrow Agent agrees to deliver to such successor escrow agent all funds and documents in its possession within thirty (30) days from the date of its notice of termination or resignation, or five (5) business days from the date of appointment of the successor escrow agent, whichever shall later occur. The successor escrow agent shall also be required to have its principal place of business in Connecticut.

10. Fees. Provider shall pay the Escrow Agent its usual and customary fee for services as Escrow Agent as such amounts are set by the Escrow Agent from time to time. Fees for any additional or extraordinary services may be agreed upon by Provider and the Escrow Agent.

11. <u>Amendments</u>. This Agreement may be amended by the parties in writing signed by both, provided that no amendments be made without prior notification by the Provider to the Commissioner of the Connecticut Department on Aging and prior to approval by the Commissioner of the proposed changes, if and to the extent required by the Act or regulations thereunder.

12. <u>Governing Law</u>. This Agreement shall be construed according to the laws of the State of Connecticut.

13. <u>Superseder</u>. This Agreement supersedes and replaces any and all prior agreements between the parties concerning the subject matter hereof.

IN WITNESS WHEREOF, the parties have executed this Entrance Fee Escrow Agreement as of the 26 day of $Aman_{26}$, 1990

UNION TRUST COMPANY

SHORELINE CARE LIMITED PARTNERSHIP d/b/a EVERGREEN WOODS

By U-in-

Trust Uttlin Its M.G.P. Its Samis

-8-

and the second

FORM 1 Entrance Fee Deposit

Date:______

To: [Bank]

In accordance with the Entrance Fee Escrow Agreement entered into between Shoreline Care Limited Partnership d/b/a Evergreen Woods, and ______, please deposit this entrance fee payment into the Established account for Evergreen Woods' entrance fees.

- 1. Date of Residency Agreement:
- 2. Amount: \$
- 3. Name on Residency Agreement:
- Copy of Signature Page of Residency Agreement attached.

As the managing general partner of Shoreline Care Limited Partnership, I am authorized to act on behalf of Shoreline Care Limited Partnership.

Sincerely,

Managing General Partner

- ð -

FORM 2 Recision

Date:_____

To: [Bank]

In accordance with Paragraph 5.1 or 5.2 of the Entrance Fee Escrow Agreement entered into between Shoreline Care Limited Partnership d/b/a Evergreen Woods, and ______, please release the Refundable Deposits paid to Evergreen Woods to the Resident named in the Residency Agreement, as outlined in Section 2 of the Residency Agreement.

- 1. Name on Residency Agreement:
- 2. Amount of deposit: \$
- 3. Any cancellation costs: \$
- 4. Amount to be refunded to Resident: \$
- 5. Amount to be refunded to Evergreen Woods: \$

As the managing general partner of Shoreline Care Limited Partnership, I am authorized to act on behalf of Shoreline Care Limited Partnership.

Sincerely,

Managing General Partner

FORM 3 Release of Deposit/Balance Payment

Date:____

To: [Bank]

In accordance with Paragraph 5.4 of the Entrance Fee Escrow Agreement (the "Escrow Agreement") entered into between Shoreline Care Limited Partnership d/b/a Evergreen Woods, and ______, please release the amount of Entrance Fees indicated below and deposited with you pursuant to the Escrow Agreement to Shoreline Care Limited Partnership. We hereby certify that the conditions set forth in Section 5.4 of the Entrance Fee Escrow Agreement have been satisfied and that the living unit referred to in the Residency Agreement is now available for occupancy by said Resident.

1. Amount to be released: \$

2. Name on Residency Agreement:

As the managing general partner of Shoreline Care Limited Partnership, I am authorized to act on behalf of Shoreline Care Limited Partnership.

Sincerely,

Managing General Partner

1 1

FORM 4 Release to Operating Reserve

Date:_____

To: [Bank]

In accordance with Faragraph 5.3 of the Entrance Fee Escrow Agreement entered into between Shoreline Care Limited Partnership d/b/a Evergreen Woods, and ____ please release the amount set forth below for deposit into the operating reserve escrow for Evergreen Woods. The undersigned hereb/ certifies that all conditions required by applicable statute have been met to permit such release and that the amount to be released, together with other amounts to be deposited in such operating reserve escrow, are sufficient to cover: (1) all principal and interest, rental or lease payments due during the first 12 months after Evergreen Woods is first occupied by a Resident on account of any first mortgage loan or any other long-term financing of Evergreen Woods; and (2) the total cost of operations of Evergreen Woods for a one-month period, excluding debt service, rental or lease payments as described in subsection (1) above and excluding capital expenditures.

Amount of Entrance Fees to be released for deposit into operating reserve escrow:

5_____

As the managing general partner of Shoreline Care Limited Partnership, I am authorized to act on behalf of Shoreline Care Limited Partnership.

Sincerely,

Managing General Partner

0324V

ASSIGNMENT AND ASSUMPTION AGREEMENT

This ASSIGNMENT AND ASSUMPTION AGREEMENT (this "<u>Agreement</u>") is made as of the ______ day of November, 2016, by and between SHORELINE LIFE CARE, LLC, a Connecticut limited liability company ("<u>Assignor</u>"), and EVERGREEN WOODS RETIREMENT, LLC, a Connecticut limited liability company ("<u>Assignee</u>"), and relates to the Entrance Fee Escrow Agreement (Phase I) of Shoreline Care Limited Partnership dated January 26, 1990 (as the same may have been amended, modified or supplemented from time to time, the "Escrow Agreement") between Shoreline Care Limited Partnership, a Connecticut limited partnership, succeeded by Assignor, and Union Trust Company, succeeded by U.S. Bank National Association, as escrow agent (the "Escrow Agent").

RECITALS:

A. Assignor is the owner of a retirement community consisting of approximately 250 independent and assisted living apartment units and 50 skilled nursing beds located at 88 Notch Hill Road, North Branford, Connecticut, and commonly known as "Evergreen Woods" (the "Facility");

B. Assignor has entered into an Operations Transfer Agreement (the "<u>OTA</u>") dated as of August 3, 2016, by and between Assignor and Assignee, pursuant to which Assignee will purchase all of the operating assets of the Facility and become the "Provider" with respect to the Facility under Chapter 319hh of the Connecticut General Statutes (the "<u>Statutes</u>"); and

C. In order to comply with Section 17b-524 of the Statutes, and as part of the transaction contemplated in the OTA, the parties desire that the Assignor assign to Assignee all of its rights, title and interest in, and all of its obligations as "Provider" under, the Escrow Agreement and that Assignee assume and agree to perform fully all of Assignor's covenants, promises, agreements, obligations and liabilities as "Provider" under the Escrow Agreement.

AGREEMENTS:

Accordingly, in consideration of the sale of the Facility, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Assignor and Assignee hereby covenant and agree as follows:

1. <u>Assignment</u>. Assignor hereby assigns to Assignee all of its right, title and interest in, and all of its obligations as "Provider" under, the Escrow Agreement, subject to all of the terms, covenants, conditions and provisions thereof (the "<u>Assignment</u>").

2. <u>Assumption</u>. Assignee hereby accepts the Assignment and assumes and agrees to perform fully all of Assignor's covenants, promises, agreements, obligations and liabilities as "Provider" under the Escrow Agreement, whether arising or existing prior to, on or after the date hereof, including, without limitation, the payment in full of Assignor's indemnification obligations as and when due in accordance with and subject to the terms and limitations of the Escrow Agreement (collectively, the "<u>Assumed Obligations</u>").

3. <u>Representations and Warranties</u>.

(a) Assignor represents and warrants to Assignee that (i) Assignor is duly organized, validly existing and in good standing under the laws of its state of organization, and has the limited liability company power and authority and all governmental licenses, authorizations, consents and approvals, if any, required to carry on its business as now conducted; (ii) Assignor has the requisite limited liability company power and authority to execute and deliver this Agreement and to perform its obligations hereunder; (iii) the execution and delivery of this Agreement by Assignor and the performance by Assignor of its obligations hereunder have been duly and validly authorized and approved, and no other proceedings, consents or authorizations are necessary to authorize the execution, delivery and performance by Assignor of this Agreement and the obligations to be performed by it hereunder; and (iv) this Agreement has been duly executed and delivered by Assignor and constitutes a valid and binding obligation of Assignor, enforceable against Assignor in accordance with its terms, subject to applicable bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights.

(b) Assignee represents and warrants to Assignor that (i) Assignee is duly organized, validly existing and in good standing under the laws of its state of organization, and has the limited liability company power and authority and all governmental licenses, authorizations, consents and approvals, if any, required to carry on its business as now conducted; (ii) Assignee has the requisite limited liability company power and authority to execute and deliver this Agreement and to perform its obligations hereunder; (iii) the execution and delivery of this Agreement by Assignee and the performance by Assignee of its obligations hereunder have been duly and validly authorized and approved, and no other proceedings, consents or authorizations are necessary to authorize the execution, delivery and performance by Assignee of this Agreement and the obligations to be performed by it hereunder; and (iv) this Agreement has been duly executed and delivered by Assignee and constitutes a valid and binding obligation of Assignee, enforceable against Assignee in accordance with its terms, subject to applicable bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights.

4. <u>Notices</u>. Any notices or other communications under this Agreement shall be in writing and either shall be mailed by certified mail, postage prepaid, return receipt requested, or sent by overnight air courier service, or personally delivered to a representative of the receiving party, or sent by facsimile transmission (provided an identical notice is also sent simultaneously by mail, overnight courier or personal delivery as otherwise provided in this Section 4). All such notices and communications shall be mailed, sent or delivered, addressed to the party for whom it is intended at its address set forth below.

If to Assignor:

Shoreline Life Care, LLC c/o Shelter Development, LLC 218 N Charles Street, Suite 220 Baltimore, Maryland 21201 Attention: Jeffrey Hettleman Facsimile: (410) 347-0587

with a copy to:	Gallagher Evelius and Jones LLP 218 N Charles Street, Suite 400 Baltimore, Maryland 21201 Attention: Stephen A. Goldberg Facsimile: (410) 468-2786			
If to Assignee:	Evergreen Woods Retirement, LLC c/o Senior Living Communities, LLC 3530 Toringdon Way, Suite 204 Charlotte, North Carolina 28277 Attention: Donald O. Thompson, Jr. Facsimile: (704) 246-1621			
With a copy to:	Allen K. Robertson Robinson, Bradshaw & Hinson, P.A. 101 North Tryon Street, Suite 1900 Charlotte, North Carolina 28246 Facsimile: (704) 373-3968			

5. <u>Miscellaneous</u>.

(a) This Agreement shall be construed and governed in accordance with the laws of the State of Connecticut. Each of the parties to this Agreement has participated fully in the negotiation and preparation hereof. Accordingly, this Agreement shall not be more strictly construed against either of the parties hereto. Each of the parties hereto irrevocably submits to the non-exclusive jurisdiction and venue of any State or federal court in the State of Connecticut and waives any claim that venue is improper or inconvenient.

(b) If any term or provision of this Agreement is determined to be illegal or otherwise invalid, such term or provision shall be given its nearest legal meaning or, if invalid, be construed as deleted from this Agreement, and the remainder of this Agreement shall be construed to be in full force and effect.

(c) This Agreement may be executed in two or more counterparts each of which shall be deemed an original, and all of which shall constitute one and the same instrument. This Agreement may be executed and delivered by electronic mail, pdf or other electronic means with the same force and effect as delivery of an original.

(d) This Agreement may not be changed, altered or modified except by an instrument in writing signed by the party against whom enforcement of such change, alteration or modification is sought. This Agreement shall be binding upon the parties hereto and their respective successors and permitted assigns. Neither party hereto shall have the right, directly or indirectly, to assign all or any portion of its rights or obligations under this Agreement to any person or entity, without first obtaining the written consent of the other party to any such assignment.

[Signatures Appear on Following Page]

IN WITNESS WHEREOF, Assignor and Assignee have executed this Agreement as of the date first set forth above.

ASSIGNOR:

SHORELINE LIFE CARE LLC,

a Connecticut limited liability company

By: Brightview Evergreen Woods, LLC, a Maryland limited liability company, its managing member

By: Name: Title:

ASSIGNEE:

EVERGREEN WOODS RETIREMENT, LLC, a Connecticut limited liability company

By: Senior Living Communities, LLC, a North Carolina limited liability company its Manager

By:

Name: Donald O. Thompson, Jr. Title: Manager

ACKNOWLEDGED:

U.S. BANK NATIONAL ASSOCIATION, as Escrow Agent

By:	 	 	
Name:			
Title:			

IN WITNESS WHEREOF, Assignor and Assignee have executed this Agreement as of the date first set forth above.

ASSIGNOR:

SHORELINE LIFE CARE LLC,

a Connecticut limited liability company

By: Brightview Evergreen Woods, LLC, a Maryland limited liability company, its managing member

By:

NT	
Name:	
Title	
1 me.	

ASSIGNEE:

By:

EVERGREEN WOODS RETIREMENT, LLC, a Connecticut limited liability company

By: Senior Living Communities, LLC, a North Carolina limited liability company its Manager

The S. Thomas

Name: Donald O. Thompson, Jr. Title: Manager

ACKNOWLEDGED:

U.S. BANK NATIONAL ASSOCIATION, as Escrow Agent

By:	
Name:	
Title:	

IN WITNESS WHEREOF, Assignor and Assignee have executed this Agreement as of the date first set forth above.

ASSIGNOR:

SHORELINE LIFE CARE, LLC,

a Connecticut limited liability company

By: Brightview Evergreen Woods, LLC, a Maryland limited liability company, its managing member

ASSIGNEE:

EVERGREEN WOODS RETIREMENT, LLC, a Connecticut limited liability company

By: Senior Living Communities, LLC, a North Carolina limited liability company its Manager

By:

Name: Donald O. Thompson, Jr. Title: Manager

ACKNOWLEDGED:

U.S. BANK	K NATIONAL ASSOCIATION,
as Escrow.	Agent $\land \land \land \land \land$
By:	All
Name:	John G. Correia
Title:	Vice President

EXHIBIT D

SWORN STATEMENT OF OPERATING AND DEBT SERVICE RESERVE ESCROW AGENT

See attached.



The undersigned, KeyBank National Association, hereby affirms and swears that: (1) it serves as Escrow Agent under the Reserve Escrow Agreement dated October 27, 2016 (the "Debt Service and Operating Reserve Escrow Agreement") between Evergreen Woods Retirement, LLC, a Connecticut limited liability company ("Provider") and KeyBank National Association, as escrow agent (the "Escrow Agent"), and (2) the Debt Service and Operating Reserve Escrow Agreement states that the Provider will make certain deposits with the Escrow Agent for the purpose of maintaining a reserve fund in compliance with Section 17b-525 of the Connecticut General Statutes, as amended from time to time. A copy of the Debt Service and Operating Reserve Escrow Agreement is attached hereto.

IN WITNESS WHEREOF, the undersigned Escrow Agent has caused this statement to be executed by its duly authorized officer on this 04th day of September, 2019.

KeyBank National Association, as Escrow Agent

By:

Name: Lee Ann Habinak Title: Vice President

STATE OF OHIO

COUNTY OF CUYAHOGA

On this <u>4th</u> day of <u>September, 2019</u>, personally appeared, <u>Lee Ann Habinak</u>, an officer of KeyBank National Association, known to me (or satisfactorily proven) to be the person whose name is subscribed to within this document and acknowledged that he/she executed the same for the purposes therein contained.

In witness whereof I hereunto set my hand.

(Notary Seal or Stamp)

BRIAN BANJAC, ATTORNEY NOTARY PUBLIC • STATE OF OHO My commission has no expiration date Section 147.03 O.R.C. My commission expires: <u>My (Ominission</u> has no expiration

RESERVE ESCROW AGREEMENT

This Reserve Escrow Agreement is made as of the 27th day of October 2016 by and between **Evergreen Woods Retirement, LLC** d/b/a **Evergreen Woods** (the "**Provider**"), and **KeyBank National Association**, a banking institution having a place of business in the State of Connecticut (the "**Escrow Agent**"),

Recitals:

WHEREAS, the Provider is becoming the operator of a continuing care facility commonly known as "Evergreen Woods" (herein, the "Community") pursuant to an Operations Transfer Agreement dated as of August 3, 2016 (the "Operations Transfer Agreement") between the Provider and Shoreline Life Care LLC (the "Existing Provider"), and, in connection with that operations transfer, will become a "Provider" within the meaning of Chapter 319hh of the Connecticut General Statutes as amended from time to time (the "Statutes");

WHEREAS, in order to comply with Section 17b-525 of the Statutes and to protect residents of the Community, the parties desire to enter into this Reserve Escrow Agreement (the "**Agreement**") setting forth the terms under which certain deposits will be made with the Escrow Agent for the purpose of maintaining a reserve fund in compliance with the Statutes;

WHEREAS, upon the Effective Date (as hereinafter defined) of this Agreement, the Provider will transfer funds to the Escrow Agent which shall thereafter be subject to the terms and conditions of this Agreement;

WHEREAS, the Provider and Escrow Agent intend that this Agreement supersede and replace any prior agreement as to the same subject matter; and

WHEREAS, this Agreement shall become effective (the "Effective Date") on the Transfer Date (as defined in the Operations Transfer Agreement), which will occur after the Provider has received the acknowledgment from the Connecticut Department of Social Services required under Section 17b-521 of the Statutes with respect to the operations transfer from the Existing Provider to the Provider.

NOW THEREFORE, in consideration of these premises and the mutual covenants herein set forth, the parties agree as follows:

1. **Deposits**. Pursuant to Section 17b-525(a) of the Statutes, the Provider is required to establish and maintain a reserve fund escrow account (the "**Reserve Escrow Account**") sufficient to cover the following, unless the Commissioner of Social Services of the State of Connecticut (the "**Commissioner**") approves a lesser amount to be maintained: (1) all principal and interest, rental or lease payments due during the next six (6) months on account of any first mortgage loan or other long-term financing of the Community, and (2) the total cost of operations of the Community for a one month period, excluding debt service, rental or lease payments, and capital expenditures by the Provider, except that such requirement for the cost of operations for one month may be met in whole or part based on mortgage loan, bond indenture or long-term financing requirements to maintain a certain number of days of cash on hand.

Commencing with the Effective Date of this Agreement and on or before the commencement of each calendar year thereafter:

(a) the Provider shall certify to the Escrow Agent the amount that, based on Evergreen Woods most recent budgets, taking into account existing days of cash on hand requirements, as applicable, will be required to be maintained in the Reserve Escrow Account in order to satisfy the escrow requirements of Section 17b-525(a) of the Statutes, and

(b) the Provider shall deposit with the Escrow Agent the amount necessary, if any, to insure that the total funds held in the Reserve Escrow Account will equal the amount that, based on the Community's most recent budgets, taking into account existing days of cash on hand requirements, as applicable, will be required to be maintained in the Reserve Escrow Account in order to satisfy the escrow requirements of Section 17b-525(a) of the Statutes.

2. **Reserve Escrow Account/Waiver of Lien Rights**. The Escrow Agent shall maintain all such deposits made to the Reserve Escrow Account separate and apart from any other funds of the Provider or the residents and the Escrow Agent hereby waives any and all lien rights it may have with respect to said funds and further waives the right to set off against said funds for or in connection with the obligations of any person or entity.

3. **Investments**. The Escrow Agent shall invest all amounts held by it from time to time in such obligations of any federal or state or local authority (or in shares of a registered investment company which holds a portfolio of such securities) or such certificates of deposit, savings accounts, money market accounts, deposits, or other savings or investment securities including, without limitation, those of Escrow Agent, as Provider may from time to time direct in writing, or absent such direction, in such accounts or similar investments as the Escrow Agent shall deem advisable in its sole discretion. No amounts held by the Escrow Agent pursuant to this Agreement may be invested in any building or health care facility of any kind, or used for capital construction or improvements or for the purchase of real estate or removed from the State of Connecticut if required to be maintained within such State. No amounts held by the Escrow Agent pursuant to this Agreement shall be subordinated to other loans or commitments of the Community other than first mortgage loans or long-term financing obligations of the Community.

4. **Reports**. The Escrow Agent agrees, upon the request of Provider, to issue a statement indicating the status of the Reserve Escrow Account maintained by the Escrow Agent pursuant to this Agreement. In any event, the Escrow Agent agrees to provide monthly reports to the Provider, which shall include a summary of all additions to and disbursements from the Reserve Escrow Account during the current period, and a detail of the investments comprising the balance in the Reserve Escrow Account as of the end of the current period.

5. **Disbursements**. The Escrow Agent shall release any amounts then held by it pursuant to this Agreement in accordance with the following terms:

5.1 Upon receipt by the Escrow Agent of written certification from the Provider that the amount currently required under the Statutes to be maintained in the Reserve Escrow Account is less than the current balance held in the Reserve Escrow Account, the Escrow Agent shall promptly distribute to the Provider upon demand an amount equal to such excess.

5.2 Upon written demand of the Provider, the Escrow Agent shall promptly release to the Provider up to one-twelfth (1/12) of the required principal balance of funds held in the Reserve Escrow Account; provided, however, that the Escrow Agent shall not make any release under this Paragraph 5.2 more than once during any calendar month and provided further that the Provider shall give prompt written notice to the Commissioner of any such release of escrowed funds. The Escrow Agent shall notify the Commissioner if escrowed funds so released are not replaced by the Provider within one (1) year.

5.3 Upon written demand of the Provider, which demand contains evidence of authorization from the Commissioner for the release of additional funds held in the Reserve Escrow Account, the Escrow Agent shall release to the Provider such amounts as are authorized for release, provided that the Provider shall give prompt written notice to the Commissioner of any such release of escrowed funds. The Escrow Agent shall notify the Commissioner if escrowed funds so released are not replaced by the Provider within one (1) year.

5.4 Upon written demand of the Provider, which demand contains a copy of written authorization from the Commissioner for the maintenance of a reserve escrow in a lesser amount than the required reserve amount previously specified by the Provider pursuant to Section 1(a) above, the Escrow Agent shall release any excess escrowed funds, provided that the Provider shall give prompt written notice to the Commissioner of any such release of escrowed funds.

5.5 Upon written demand of the Provider, the Escrow Agent shall release to the Provider any interest or earnings generated from escrowed funds.

5.6 The parties acknowledge that under Connecticut law, the Commissioner must receive prior notification of any withdrawals contrary to the terms of CGS 17b-520 to 17b-535, as amended, or for any change in the terms of this escrow agreement in regard thereto.

6. **Duties**. The Escrow Agent undertakes to perform only such duties as are expressly set forth herein and no implied duties or obligations shall be inferred or otherwise imposed upon or against the Escrow Agent. The Escrow Agent's duties and obligations are purely ministerial in nature, and nothing herein shall be construed to give rise to any fiduciary obligations of the Escrow Agent. The Escrow Agent may act in reliance upon any writing or instrument under signature which it, in good faith, believes to be genuine; may assume the validity and accuracy of any statement or assertion contained in any such writing or instrument; and may assume that any person purporting to give any writing, notice, advice, or instruction in connection with the provisions hereof has been duly authorized to do so. The Escrow Agent

shall not be liable to anyone whatsoever by reason of any error of judgment or for any act done or step taken or omitted by it in good faith or for any mistake of fact or law or for anything which it may do or refrain from doing in connection herewith unless caused by or arising out of its own willfull misconduct or gross negligence. In no event shall the Escrow Agent be liable for any special or consequential damages caused by its gross negligence or willful misconduct even if previously informed of the possibility thereof. The Escrow Agent shall not be liable in any manner for the execution or validity of any instrument deposited with or delivered to the Escrow Agent hereunder, nor as to the identity, authority, or right of any person executing the same. The duties of the Escrow Agent hereunder shall be limited to the safekeeping of the deposits and other money, instruments, or other documents received by it as Escrow Agent, for its disposition of the same in accordance with the written instruction accepted by it as Escrow Agent.

7. Indemnity. The Provider hereby agrees to indemnify and defend the Escrow Agent and hold it harmless from any and all claims, liabilities, losses, actions, suits, or proceedings at law or in equity, or any other expense, fees, or charges of any character or nature, which it may incur or with which it may be threatened by reason of its directly or indirectly acting as Escrow Agent under this Agreement, unless caused by the Escrow Agent's willful misconduct or gross negligence; and in connection therewith, to indemnify the Escrow Agent against any and all expenses, including reasonable attorney's fees and the cost of defending any action, suit, or proceeding or resisting any claim. The parties hereto agree that the Escrow Agent does not assume any responsibility for the failure of any of the parties to make payment or perform the conditions of this Escrow Agreement as set forth herein, nor shall Escrow Agent be responsible for the collection of monies provided to be paid to it. Escrow Agent may consult with counsel of its own choice and shall have full and complete authorization and protection for any action taken or suffered by it hereunder in good faith and in accordance with the opinion of The provisions of this Section 7 shall survive termination of the escrow such counsel. arrangement contemplated hereby

8. **Termination**. Either party may terminate this Agreement upon thirty (30) days' written notice thereof to the other party, subject to the appointment of its successor by the Provider within such thirty (30) day period. Provider shall give prompt written notice to the Escrow Agent of the name and address of the successor escrow agent. If a successor escrow agent is not appointed by the Provider within the thirty (30) day period, the Escrow Agent may petition the court to appoint a successor. Upon the appointment of such successor escrow agent, the Escrow Agent agrees to deliver to such successor escrow agent all funds and documents in its possession within thirty (30) days from the date of its notice of termination or resignation, or five (5) business days from the date of appointment of the successor escrow agent, whichever occurs first. The successor escrow agent shall also be required to have a place of business in Connecticut.

9. **Fees**. The Provider shall pay the Escrow Agent its usual and customary fee for services as Escrow Agent as outlined in Exhibit "A" attached hereto. Any changes to such fees shall be effective if contained in writing signed by the Escrow Agent and the Provider. Fees for any additional or extraordinary services may be agreed upon in writing by the Provider and the Escrow Agent. The Escrow Agent shall be entitled to deduct its usual and customary fee for services as Escrow Agent from the Reserve Escrow Account, so long as the amount remaining in said account after any such deduction is not less than the amount required to be maintained in the

Reserve Escrow Account in order to satisfy the escrow requirements of Section 17b-525(a) of the Statutes.

10. **Authorized Persons of the Provider**. Escrow Agent will receive a list from the Provider of the individuals authorized to act on the Provider's behalf under the terms of this Agreement. The list shall contain the name, title, and signature of the authorized person. Whenever a change occurs in authorized persons, the Provider will provide a new list to the Escrow Agent.

11. **Governing Law/Headings**. This Agreement shall be governed by and construed in accordance with the laws of the State of Connecticut. The headings set forth in this Agreement are for convenience only and shall not affect the meaning of any term or provision of this Agreement.

12. **Successors and Assigns/Binding Effect**. The Escrow Agent shall not assign its rights or obligations under this Agreement without the prior written consent of the Provider, which consent may be granted or withheld in the Provider's sole and absolute discretion. Subject to the foregoing restriction on assignment, this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective heirs, legal representatives, successors and assigns.

13. **Amendments**. This Agreement may be unilaterally amended by the Provider from time to time; provided, however, that no amendment shall alter the duties of the Escrow Agent without the consent of the Escrow Agent, and no amendment shall alter the covenants of the Provider under the residency agreements between the Provider and the residents of the Community. If the Provider wishes to amend this Agreement, no amendments shall be made without prior notification by the Provider to the Commissioner and prior to approval by the Commissioner of the proposed changes, if and to the extent required by the Statutes or regulations thereunder.

14. **Notices to the Parties**. All notices given to the Provider or to the Escrow Agent pursuant to this Agreement shall be in writing and shall be delivered via email, facsimile, first class mail, overnight courier or personal delivery and shall be deemed given and received on the date actually received (or the date that delivery is tendered if delivery is refused). The address to which any such notices shall be sent are as set forth below, unless a different address is specified in writing by either party after the date hereof:

If to the Provider:

Evergreen Woods Retirement, LLC

c/o Senior Living Communities, LLC 3530 Toringdon Way, Suite 204 Charlotte, North Carolina 28277 Attention: Donald O. Thompson, Jr. Fax: (704) 246-1621

With Copy To:

Robinson, Bradshaw & Hinson, P.A.

101 North Tryon Street, Suite 1900 Charlotte, North Carolina 28246 Attention: Allen K. Robertson, Esq. Telephone: (704) 377-8368 Fax: (704) 373-3968

If to the Escrow Agent:

KeyBank National Association

Attn: Lee Ann Habinak 100 Public Square, 9th Floor, Suite 600 Cleveland, OH 44113 Telephone: 216-689-3226 Fax: 800-642-5089

[Signatures begin on the next page]

IN WITNESS WHEREOF, the parties have executed this Agreement as of the day and year first above written.

Evergreen Woods Retirement, LLC, a Connecticut limited liability company

Senior Living Communities, LLC, By: a North Carolina limited liability company its Manager

By: Donald O. Thompson, Jr.

Title: Manager

KeyBank National Association

By: Name: Lee ANN HABINAK Title: VICE PETSIDENT

EXHIBIT A

Fee Exhibit

To Escrow Agreement

Fees:

KeyBank, National Association

Annual Administrative Escrow Fee payable upon execution of agreement, and annually thereafter upon the anniversary date of the account opening:

Annual Administrative Fee: \$5,000.00

*Note: Should the Parties to the Escrow Agreement direct alternative investments other than the Federated Money Market funds (Government Obligations or Treasury Obligations Capital Shares), an additional custodial fee of 5 basis points of assets market value will be charged quarterly against the escrowed funds. The custody fee will be in addition to the annual administrative escrow fee.

EXHIBIT E

AUDITED FINANCIAL STATEMENTS

See attached.

EVERGREEN WOODS RETIREMENT, LLC

North Branford, Connecticut

Financial Statements

At

December 31, 2019 and 2018

And

For The Years Ended

December 31, 2019, 2018 and 2017

* * * * * * *

TABLE OF CONTENTS

Page

Financial Statements:

Independent Auditor's Report. 2 Balance Sheets 3 - 4 Statements of Operations 5 Statements of Changes in Member's Deficit 6 Statements of Cash Flows 7 - 8 Notes to Financial Statements 9 - 22



Moyer, Smith & Roller, P.A.

Certified Public Accountants

7229 Albemarle Rd., Suite A Charlotte, NC 28227 T: 704-566-0222 F: 704-531-6197 cpacharlotte@msr-cpa.com www.msr-cpa.com

Independent Auditor's Report

To the Member of Evergreen Woods Retirement, LLC

We have audited the accompanying financial statements of Evergreen Woods Retirement, LLC (a Connecticut limited liability company), which comprise the balance sheets as of December 31, 2019 and 2018, and the related statements of operations, changes in member's deficit, and cash flows for the years ended December 31, 2019, 2018 and 2017, and the related notes to the financial statements.

Management's Responsibility for the Financial Statements

Management is responsible for the preparation and fair presentation of these financial statements in accordance with accounting principles generally accepted in the United States of America; this includes the design, implementation and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

Auditor's Responsibility

Our responsibility is to express an opinion on these financial statements based on our audits. We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatement.

An audit involves performing procedures to obtain audit evidence about the amounts and disclosures in the financial statements. The procedures selected depend on the auditor's judgment, including the assessment of the risks of material misstatement of the financial statements, whether due to fraud or error. In making those risk assessments, the auditor considers internal control relevant to the entity's preparation and fair presentation of the financial statements in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the entity's internal control. Accordingly, we express no such opinion. An audit also includes evaluating the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluating the overall presentation of the financial statements.

We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Opinion

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Evergreen Woods Retirement, LLC as of December 31, 2019 and 2018, and the results of its operations and its cash flows for the years ended December 31, 2019, 2018 and 2017, in accordance with accounting principles generally accepted in the United States of America.

Changes in Accounting Principles

As described in Note B to the financial statements, effective January 1, 2019, the Company adopted new accounting guidance for goodwill and for leases. ASU 2017-04 simplifies the accounting for goodwill impairments. ASC 842, *Leases* requires lessees to recognize lease assets and lease liabilities in the balance sheets and to disclose key information about leasing arrangements. Effective January 1, 2018, the Company adopted ASC 606, *Revenue from Contracts with Customers*, related to recognition of revenue from occupancy agreements with residents and costs incurred by the Company to obtain those contracts. Our opinion is not modified with respect to these matters.

Moyer, Smith + Roller, P.A.

Charlotte, North Carolina June 25, 2020

EVERGREEN WOODS RETIREMENT, LLC Balance Sheets December 31, 2019 and 2018

ASSETS

	2019		2018	
Current Assets				
Cash	\$	-	\$	7,346
Occupancy Fee Deposits in Escrow		120,600		100
Accounts Receivable, Net		417,013		433,794
Accounts Receivable - Other		20,548		54,114
Current Portion of Deferred Commissions		210,575		164,646
Prepaid Expenses		69,416		123,204
Inventory		23,529		19,310
Total Current Assets		861,681		802,514
Property and Equipment				
Leasehold Improvements		3,028,061		2,461,221
Site Improvements		21,232		16,826
Furniture, Fixtures and Equipment		167,474		122,595
Automobiles and Golf Carts		298,675		303,675
Total Property and Equipment		3,515,442		2,904,317
Less Accumulated Depreciation		(1,242,025)		(617,349)
Property and Equipment, Net		2,273,417		2,286,968
Other Assets				
Deferred Commissions, Net		565,836		493,967
Right-of-Use Assets - Operating Leases		42,695,238		-
Goodwill		56,089,440		56,089,440
Total Other Assets		99,350,514		56,583,407
TOTAL ASSETS	\$	102,485,612	\$	59,672,889

EVERGREEN WOODS RETIREMENT, LLC Balance Sheets (Continued) December 31, 2019 and 2018

LIABILITIES AND MEMBER'S DEFICIT

	2019		2018	
Current Liabilities				
Checks in Process	\$	31,925	\$	14,498
Accounts Payable		989,480		784,453
Monthly Service Fees Received in Advance		203,585		205,549
Accrued Expenses		3,530,943		2,225,240
Resident Deposits		52,945		-
Current Portion of Deferred Revenues		1,724,827		1,439,716
Current Portion of Operating Lease Liabilities		2,366,496		-
Current Portion of Note Payable		54,681		52,357
Total Current Liabilities		8,954,882		4,721,813
Long-Term Liabilities				
Straight-Line Rent Payable		-		2,404,755
Operating Lease Liabilities, Net of Current Portion		43,450,962		-
Notes Payable, Net of Current Portion		88,155		142,910
Deposits on Occupancy Agreements		599,620		404,001
Refundable Occupancy Fees		52,530,927		54,323,752
Deferred Revenues, Net of Current Portion		2,009,422		2,086,415
Due to Related Party		5,172,334		2,912,115
Long-Term Liabilities, Net		103,851,420		62,273,948
Total Liabilities		112,806,302		66,995,761
Member's Deficit		(10,320,690)	1	(7,322,872)
TOTAL LIABILITIES AND MEMBER'S DEFICIT	\$	102,485,612	\$	59,672,889

See Independent Auditor's Report and Accompanying Notes.

EVERGREEN WOODS RETIREMENT, LLC Statements of Operations For the Years Ended December 31, 2019, 2018 and 2017

	2019	2018	2017
Revenues			
Service Fees	\$ 16,855,416	\$ 16,971,998	\$ 16,880,502
Occupancy Fees Earned	2,366,497	1,888,155	1,057,530
Ancillary Income	 472,055	 341,955	 375,393
Revenues, Net	 19,693,968	 19,202,108	 18,313,425
Expenses			
Dining Services	2,092,220	2,027,350	2,212,983
Assisted Living	1,141,575	1,089,570	1,045,460
Skilled Nursing	3,558,366	3,356,227	3,500,404
Facilities Maintenance	1,639,539	1,961,725	1,659,512
Housekeeping	629,647	617,827	666,962
Member Services	438,111	418,739	431,918
Wellness	94,070	96,914	83,532
Marketing	903,101	764,243	1,117,103
Administration and Overhead	1,294,160	1,413,333	1,351,337
Utilities	999,206	1,066,527	1,049,184
Property Taxes	1,650,844	1,507,870	1,366,867
Insurance	508,190	500,447	399,867
Management Services	662,116	644,473	706,227
Provision for Bad Debt Expense	25,399	115,536	34,992
Depreciation and Amortization	629,676	453,890	157,358
Total Operating Expenses	 16,266,220	 16,034,671	 15,783,706
Operating Income Before Rent	3,427,748	3,167,437	2,529,719
Rent Expense	 (6,421,785)	 (6,396,304)	 (6,331,451)
Operating Loss	 (2,994,037)	 (3,228,867)	 (3,801,732)
Other Income (Expense)			
Gain on Sale of Assets	-	-	750
Interest Income	11,834	416	-
Interest Expense	 (15,615)	 (395)	 (47,276)
Other Income (Expense), Net	 (3,781)	 21	 (46,526)
Net Loss	\$ (2,997,818)	\$ (3,228,846)	\$ (3,848,258)

EVERGREEN WOODS RETIREMENT, LLC Statements of Changes in Member's Deficit For the Years Ended December 31, 2019, 2018 and 2017

Balance at January 1, 2017	\$ (715,313)
Net Loss	 (3,848,258)
Balance at December 31, 2017	\$ (4,563,571)
Cumulative Effect of a Change in Accounting Principle Net Loss	 469,545 (3,228,846)
Balance at December 31, 2018	\$ (7,322,872)
Net Loss	 (2,997,818)
Balance at December 31, 2019	\$ (10,320,690)

EVERGREEN WOODS RETIREMENT, LLC Statements of Cash Flows For the Years Ended December 31, 2019, 2018 and 2017

	2019		2018		2017
Cash Flows from Operating Activities					
Net Loss	\$ (2,997,818)	\$	(3,228,846)	\$	(3,848,258)
Adjustments to reconcile net loss					
to net cash used in operations:					
Depreciation and Amortization	629,676		453,890		157,358
Amortization of Right-of-Use Assets	2,741,067		-		-
Amortization of Deferred Commissions	260,764		200,176		-
Amortization of Occupancy Fees	(2,366,497)		(1,888,155)		(1,057,530)
Provision for Bad Debt Expense	25,399		115,536		34,992
Gain on Sale of Assets	-		-		(750)
Interest Expense Added to NHI Line of Credit	-		-		40,658
(Increase) Decrease in:					
Accounts Receivable	(8,618)		(113,461)		108,490
Accounts Receivable - Other	33,566		(54,114)		-
Deferred Commissions	(378,562)		(389,244)		-
Prepaid Expenses	53,788		(9,573)		(83,061)
Inventory	(4,219)		(858)		(63)
Other Intercompany Receivables	-		-		727,384
Increase (Decrease) in:					
Checks in Process	17,427		5,631		(167,567)
Accounts Payable	205,027		232,544		397,801
Monthly Service Fees Received in Advance	(1,964)		(104,246)		(58,029)
Accrued Expenses	1,305,703		277,521		(1,388,792)
Straight-Line Rent Payable	-		958,881		1,120,951
Operating Lease Liabilities	(2,023,602)		-		-
Resident Deposits	52,945		(2,500)		(135,776)
Deposits on Occupancy Agreements	195,619		(31,605)		344,173
Other Intercompany Payables	2,260,219		2,500,066		412,049
Net Cash Used in Operating Activities	(80)		(1,078,357)		(3,395,970)

EVERGREEN WOODS RETIREMENT, LLC Statements of Cash Flows (Continued) For the Years Ended December 31, 2019, 2018 and 2017

		2019		2018		2017
Cash Flows from Investing Activities						
Additions to Property and Equipment	\$	(616,125)	\$	(1,182,285)	\$	(1,391,204)
Proceeds from Sale of Assets		-		-		750
Net Cash Used in Investing Activities		(616,125)		(1,182,285)		(1,390,454)
Cash Flows from Financing Activities						
Occupancy Fees Received		5,942,510		8,068,795		10,090,473
Occupancy Fees Refunded		(5,160,720)		(5,546,656)		(4,491,703)
Repayments on Line of Credit from NHI		-		(302,365)		(722,213)
Principal Payments on Note Payable		(52,431)		(21,073)		(27,934)
Net Cash Provided by Financing Activities		729,359		2,198,701		4,848,623
Net Increase (Decrease) in Cash and Restricted Cash		113,154		(61,941)		62,199
Cash and Restricted Cash at Beginning of Year		7,446		69,387		7,188
Cash and Restricted Cash at End of Year	\$	120,600	\$	7,446	\$	69,387
Supplemental Cash Flow Information:						
Cash paid for interest expensed Cash paid for amounts included in the measurement of lease liabilities:	\$	15,615	\$	395	\$	6,618
Operating cash flows from operating leases	\$	5,725,343	\$	-	\$	-
Significant non-cash transactions:	¢		¢		¢	77 205
Refinancing of Note Payable for vehicles Vehicles acquired with Note Payable	\$ \$	-	\$ \$	- 141,820	\$ \$	77,295

NOTE A - SIGNIFICANT ACCOUNTING POLICIES

Business Activity - Evergreen Woods Retirement, LLC (the Company) was formed in June 27, 2016. It operates a continuing care retirement community (CCRC) in North Branford, Connecticut. It features 229 independent living apartments, 20 assisted living units, and 50 skilled nursing units.

Use of Estimates - The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Reclassifications - Certain accounts in the prior-year financial statements have been reclassified for comparative purposes to conform with the presentation in the current-year financial statements.

Limited Liability Company / Income Taxes - The Company is a single member Limited Liability Company, whose LLC member is Senior Living Communities, LLC. The Company's taxable earnings or losses are passed through to the Company's member who is then taxed based on the taxable earnings or losses. Accordingly, no provision or benefit for federal or state income taxes has been reported by the Company.

As a limited liability company, the member's liability is limited to amounts reflected in the member's capital account.

Cash and Cash Equivalents - The Company considers all highly liquid unrestricted investments with maturities of three months or less to be cash equivalents. The Company had no cash equivalents during the years ended December 31, 2019 and 2018.

Accounts Receivable and Allowance for Doubtful Accounts - Accounts receivable are reported net of an allowance for doubtful accounts. On a periodic basis, management evaluates accounts receivable balances and establishes an allowance for doubtful accounts, based on its analysis of collectability as the receivables age. The allowance for doubtful accounts at December 31, 2019 and 2018 is \$91,676 and \$87,659, respectively.

Accounts receivable includes amounts due from third-party payors. Billings for services under third-party payor programs are recorded at amounts expected to be collected. Subsequent positive or negative adjustments are recorded when known. Contractual or cost related adjustments from Medicare are accrued when assessed. Gross amounts due from third-party payors at December 31, 2019 and 2018, are \$377,070 and \$416,627, respectively.

Inventory - Inventories are stated at cost determined on the first-in, first-out basis.

Property and Equipment - Property and equipment are carried at cost. Depreciation is computed using the straight-line method over the estimated useful lives of the assets. Major renewals and improvements are capitalized while replacements, maintenance, and repairs, which do not improve or extend the life of the assets, are expensed currently. When assets are sold or retired, their cost and accumulated depreciation are removed from the accounts and resulting gains and losses are included in the Statements of Operations.

Deferred Commissions - The Company pays commissions on all occupancy agreements. The Company will occasionally provide move-in services or other incentives to residents to sign an occupancy agreement for independent living units. On January 1, 2018, the Company adopted ASC 606, *Revenue from Contracts with Customers*, which requires all incremental costs of entering into contracts with customers to be recognized as an asset and amortized over the expected life of the contracts. The nonrefundable portion of occupancy fees received from independent living residents prior to moving into the community is earned over a five-year amortization period. The related costs are expensed on a straight-line basis over the same five-year amortization period. If a resident moves out prior to the end of the five-year period, the remaining costs will be expensed in the same period. Amortization of these costs is included in marketing expenses. Accumulated amortization of deferred commissions at December 31, 2019 and 2018 is \$290,074 and \$135,658, respectively.

NOTE A - SIGNIFICANT ACCOUNTING POLICIES (Continued)

Deferred Commissions (continued) - The Company expenses as incurred costs related to acquiring healthcare rental agreements with residents who move into assisted living or skilled nursing under the practical expedient permitted under FASB ASC 340, *Other Assets and Deferred Costs,* because these contracts are considered rentals that are renewed monthly.

Prior to January 1, 2018, the Company expensed as incurred all commissions and move-in incentives in accordance with ASC 954, *Health Care Entities*.

Goodwill - The Company early adopted ASU 2017-04, *Intangibles* – *Goodwill and Other (Topic 350): Simplifying the Accounting for Goodwill Impairment*. This ASU requires entities to calculate impairment as the amount by which a reporting unit's carrying value exceeds its fair value, not to exceed the carrying amount of goodwill. The ASU requires the same test be applied to goodwill for all reporting units, even those with zero or negative carrying amounts and requires entities to disclose the amount of goodwill at reporting units with zero or negative carrying amounts. The Company has a single reporting unit. The carrying value of goodwill at December 31, 2019 and 2018 is \$56,089,440.

Prior to the adoption of ASU 2017-04, the Company followed ASC 350, *Goodwill and Other Intangible Assets*, and tested goodwill for impairment annually or whenever indicators of impairment arise. For the years ended December 31, 2019, 2018 and 2017, no impairment of goodwill was identified.

Checks in Process - The Company has a sweep arrangement with its banks that includes the bank accounts of related entities. Under this arrangement, the Company's operating accounts are zero balance accounts. The daily ending balances are cleared by sweep transfers a to master account held by Senior Living Communities, LLC. When funds are required in the operating account to clear checks and other disbursements, automatic transfers from the master account to the operating account occur for the required amounts. Checks issued and outstanding are shown as a current liability on the Company's Balance Sheets.

Compensated Absences - Employees of the Company are entitled to paid time off depending upon job classification, length of service, and other factors. The amount of paid time off earned but not taken by employees is recorded as a current liability on the Balance Sheets. The amount reflects as of December 31, 2019 and 2018, all unused time off employees have earned at their current rate of pay that would be payable upon termination under Company's payout policy. The applicable share of employer-related taxes payable on the eventual payments has not been accrued as it is not practicable for the Company to estimate. The Company's policy is to recognize the employer-related taxes when the payment is made to the employee. Accrued paid time off at December 31, 2019 and 2018, totals \$333,998 and \$308,976, respectively.

Self Insurance - The Company self insures health related claims for its covered employees up to certain limits. Claims in excess of these limits are insured with stop-loss insurance. The Company has accrued a liability it believes is adequate to cover the outstanding claims and claims that have been incurred but not yet reported as of December 31, 2019. Any subsequent changes in estimate are recorded in the period in which they are determined.

Revenue Recognition - The Company offers various forms of contracts (occupancy agreements) with independent living residents which require an entrance fee to be paid upon moving into its communities. The nonrefundable portion of the occupancy fee paid represents contract revenues to be recognized over the expected length of time a resident will reside in the community's independent living. Monthly services fees are revenues associated with the contracts with residents and are billed monthly in advance. The Company recognizes revenue from monthly services fees as services are provided to residents under guidance that applies to contracts that are monthly with the option to renew. Monthly service fee revenue is recognized at the billing rates that the Company establishes from time to time. Ancillary Income is recognized from services provided to residents on an "as needed" basis and for which a separate charge is assessed and billed in arrears. See Note L for further discussion of the Company's revenues.

Leases - On November 8, 2016, Senior Living Communities, LLC entered into a lease agreement with NHI-REIT of Evergreen, LLC for this community. The lease has an initial lease term that is set to expire on December 31, 2029, and has two options to extend the lease for an additional five years with each extension. Senior Living Communities, LLC entered into a sub-lease agreement with Evergreen Woods Retirement, LLC under the same lease terms it has with NHI-REIT of Evergreen, LLC.

NOTE A - SIGNIFICANT ACCOUNTING POLICIES (Continued)

With the adoption of ASC 842, *Leases*, on January 1, 2019, the Company determined whether an arrangement is or contains a lease at lease inception. Operating leases are included in operating lease right-of-use (ROU) assets, current operating lease liabilities and non-current operating lease liabilities in the Balance Sheets. The right-of-use asset represents the Company's right to use the leased assets over the term of the lease. Lease liabilities represent the Company's obligations to make lease payments over the term of the lease.

For operating leases, the Company determines the present value of the lease payments over the lease term, including any renewal periods that they are reasonably certain they will exercise. To determine the present value of the lease payments, the Company uses the rate implicit in the lease if it is determinable. When the rate implicit in the lease is not determinable, the Company uses its incremental borrowing rate at the commencement date of the lease. Lease expense is recognized on a straight-line basis over the lease term. The Company has elected not to recognize an asset or obligation for leases with an initial term of twelve months or less. The expense associated with short-term leases is included operating expenses in the Statements of Operations.

To the extent that a lease arrangement includes both lease and non-lease components, where the Company is the lessee, the components are accounted for separately.

Prior to the adoption of ASC 842, *Leases*, the Company recognized lease expense on a straight-line basis over the life of the lease. The difference between rent expense recognized and rental payments, as stipulated in the lease, was reflected as straight-line rent payable in the Balance Sheets.

See Note M for further discussion of the Company's leases.

Advertising - The Company expenses advertising costs when the advertising first takes place. Advertising expenses for the years ended December 31, 2019 and 2018 are \$199,704 and \$203,950, respectively.

Fair Value of Financial Instruments - ASC 820-10 defines fair value, establishes a three-level valuation framework and hierarchy for disclosure of fair value measurements. That framework prioritizes the inputs to valuation used to measure fair value of an asset or liability as of the measurement date. The hierarchy gives the highest priority to the lowest level of input that is significant to the fair value measurement. The three levels are described as follows:

Level 1 - Inputs to the valuation methodology are unadjusted quoted prices for identical assets or liabilities in active markets. Level 2 - Inputs to the valuation methodology include quoted market prices for similar assets or liabilities in active markets and inputs that are observable for the asset or liability, either directly or indirectly, for substantially the full term of the asset or liability. Level 3 - Inputs to the valuation methodology are unobservable and significant to the fair value measurement.

Cash and occupancy fee deposits in escrow are carried at amounts considered by management to approximate fair value based on Level 1 valuation.

Recent Accounting Pronouncements - In May of 2014 the Financial Accounting Standards Board (FASB) issued ASU 2014-09, *Revenue from Contracts with Customers (Topic 606)*. In August of 2015, the FASB issued ASU 2015-14, which defers the effective date of ASU 2014-09 to be effective for public entities with annual reporting periods beginning after December 15, 2017. Other entities are required to apply the new standard for annual reporting periods beginning after December 15, 2018. Early adoption is permitted. Under this new standard, entities are required to recognize revenue from contracts with customers as goods or services are provided under the terms of the contract, in amounts expected to be realized from the transfer. The Company adopted this standard on January 1, 2018, using the modified retrospective application to account for the change in accounting principle.

NOTE A - SIGNIFICANT ACCOUNTING POLICIES (Continued)

In February of 2016, the FASB issued ASU 2016-02, *Leases*, that replaces existing lease guidance in U.S. generally accepted accounting principles (GAAP) and will require companies to account for virtually all lease obligations by recording lease assets and lease liabilities on their balance sheets. The new pronouncement has an effective date for private companies for fiscal years beginning after December 15, 2019. Early adoption is permitted. The Company adopted this standard on January 1, 2019, using the modified retrospective application to account for the change in accounting principle.

In January of 2017, the FASB issued ASU 2017-04, *Intangibles – Goodwill and Other: Simplifying the Test for Goodwill Impairment*. ASU 2017-04 simplifies the subsequent measurement of goodwill by eliminating Step 2 from the goodwill impairment test. In computing an implied fair value of goodwill under Step 2, the Company had to perform procedures to determine the fair value at the impairment testing date of its assets and liabilities, including unrecognized assets and liabilities, following the procedure that would be required in determining the fair value of assets acquired and liabilities assumed in a business combination. Under this update, the Company is required to perform its annual goodwill impairment test by comparing the fair value of the reporting unit with its carrying amount. The effective date of this ASU for public companies is for years beginning after December 31, 2019, and for all others for years beginning after December 31, 2020. Early adoption is permitted and is to be applied on a prospective basis. The Company elected to adopt this standard on January 1, 2019.

NOTE B - CHANGES IN ACCOUNTING PRINCIPLES

ASU 2017-04 - Intangibles – Goodwill and Other (Topic 350)

The Company early adopted ASU 2017-04, *Intangibles – Goodwill and Other (Topic 350): Simplifying the Accounting for Goodwill Impairment.* This ASU simplifies the accounting for goodwill impairments by eliminating the requirement to compare implied fair value of goodwill with its carrying amount and requires entities to calculate impairment as the amount by which a reporting unit's carrying value exceeds its fair value, not to exceed the carrying amount of goodwill. The ASU requires the same test be applied to goodwill for all reporting units, even those with zero or negative carrying amounts and requires entities to disclose the amount of goodwill at reporting units with zero or negative carrying amounts. The new standard is effective for annual reporting periods beginning after December 15, 2017. The Company adopted this standard effective January 1, 2019. The adoption of this guidance had no impact on the Company's financial statements.

ASU 2016-02 - Leases (ASC 842)

On January 1, 2019, the Company adopted ASU 2016-02, *Leases (ASC 842)*. This new accounting standard replaces existing lease guidance in GAAP and requires entities to account for virtually all lease obligations by recording lease assets and lease liabilities on their balance sheets. As part of the transition to the new standard, the Company was required to measure and recognize leases that existed on January 1, 2019, using a modified retrospective approach. For leases existing at the effective date, the Company elected to utilize the transition practical expedients and did not reassess whether an arrangement is or contains a lease, did not reassess lease classification, did not reassess what qualifies as an initial direct cost, and elected to continue to apply legacy guidance to the prior years. The Company also elected not to recognize leases in the Balance Sheets that have an initial lease term of twelve months or less.

The adoption of ASC 842 resulted in the recognition of operating right-of-use (ROU) assets and lease liabilities of \$45,436,305 and \$47,829,838, respectively. The Company did not have any finance leases to recognize with the adoption of ASC 842. The difference between the additional lease assets and lease liabilities reduced straight-line rent payable at December 31, 2018 to zero. The standard did not materially affect the Company's net loss and had no impact on cash flows.

NOTE B - CHANGES IN ACCOUNTING PRINCIPLES (Continued)

The effects on the Balance Sheets resulting from the adoption of ASC 842 for the year ending December 31, 2019 are outlined below:

	As Reported	Legacy GAAP	Impact of 842
Assets:			
Current Assets	\$ 861,681	\$ 861,681	\$ -
Property and Equipment, Net	2,273,417	2,273,417	-
Other Assets	99,350,514	56,655,276	42,695,238
Total Assets	<u>\$ 102,485,612</u>	<u>\$ 59,790,374</u>	<u>\$ 42,695,238</u>
Liabilities and Members' Capital (Deficit):			
Liabilities:			
Accounts Payable and Other Current Liabilities	\$ 6,588,386	\$ 6,588,386	\$ -
Operating Lease Liabilities, Current Portion	2,366,496		2,366,496
Total Current Liabilities	8,954,882	6,588,386	2,366,496
Other Long-Term Liabilities, Net of Current Portion	60,400,458	60,400,458	-
Straight-Line Rent Payable	-	3,122,220	(3,122,220)
Operating Lease Liabilities, Net of Current Portion	43,450,962		43,450,962
Total Long-Term Liabilities	103,851,420	63,522,678	40,328,742
Total Liabilities	112,806,302	70,111,064	42,695,238
Members' Capital (Deficit)	(<u>10,320,690</u>)	(<u>10,320,690</u>)	()
Total Liabilities and Members' Capital (Deficit)	<u>\$ 102,485,612</u>	<u>\$ 59,790,374</u>	<u>\$ 42,695,238</u>

ASU 2014-09 - Revenue from Contracts with Customers (ASC 606)

ASC 606, *Revenue from Contracts with Customers*, related to the nonrefundable portion of occupancy fees received in advance associated with occupancy agreements with residents and costs incurred by the Company to obtain those contracts became effective January 1, 2018. In adopting ASC 606, the Company has made no changes to its method of recognizing revenues related to the nonrefundable portion of occupancy fees received in advance or to the monthly service fees and ancillary income recognized as services are provided to residents. The Company continues to recognize the nonrefundable portion of the occupancy fees received in advance on a straight-line basis over a five-year amortization period, which approximates the average length of residency of the independent living residents of all communities owned by Senior Living Communities, LLC.

Beginning January 1, 2018, the Company changed its method of accounting for commissions and other costs directly related to entering into occupancy agreements that require an advanced payment from independent living residents moving into the community. Because the Company expects to recover these costs from the revenues earned under the occupancy agreements signed with residents over a period of five years, the Company believes these costs fall within the definition in the new standard of incremental costs of obtaining a contract. The new standard requires these costs to be capitalized and amortized on a systematic basis that is consistent with fulfilling the Company's performance obligations under the occupancy agreements. The Company records amortization as a marketing expense over the same five-year amortization period that it recognizes the revenue related to the occupancy agreements. If a resident moves out before the end of the five-year period, any unamortized costs related to that resident's occupancy agreement are expensed.

The Company applied ASC 606 using the modified retrospective method applicable to all occupancy agreements requiring an entrance fee from a resident when the resident moves into the community. Results for reporting periods beginning after January 1, 2018 are presented under ASC 606 while prior periods continue to be reported in accordance with legacy GAAP. The Company recorded a net change in beginning member's deficit of \$469,545 as of January 1, 2018 due to the cumulative effect of adopting ASC 606. The transition adjustment is the result of capitalizing commissions associated with the occupancy agreements with residents.

NOTE B - CHANGES IN ACCOUNTING PRINCIPLES (Continued)

The effects on the Balance Sheets resulting from the adoption of ASC 606 for the year ending December 31, 2018 are outlined below:

	As	Reported	Legacy	y GAAP	Imp	act of 606
Deferred Commissions	\$	794,271	\$	-	\$	794,271
Accumulated Amortization of Deferred Commissions	(<u>135,658</u>)		-	(135,658)
Deferred Commissions, Net	\$	658,613	\$	_	\$	658,613

The effects on the Statements of Operations resulting from the adoption of ASC 606 for the year ending December 31, 2018 are outlined below:

	As	Reported	Leg	gacy GAAP	Imp	act of 606
Marketing Expense	\$	764,243	\$	938,273	\$	174,030
Operating Income (Loss) Before Rent from						
Continuing Operations		3,167,437		2,993,407		174,030
Operating Income (Loss)	(3,228,867)	(3,402,897)		174,030
Net Income (Loss)	(3,228,846)	(3,402,876)		174,030

NOTE C - RELATED PARTY TRANSACTIONS

Related Party	How Related
Donald O. Thompson, Jr. and Brenda U. Thompson	Senior Living Communities, LLC members
Maxwell Group, Inc.	Owned and controlled by Donald O. Thompson, Jr.
Senior Living Communities, LLC	Evergreen Woods Retirement, LLC member

In its capacity as the parent company, Senior Living Communities, LLC pays various expenses on behalf of its subsidiaries, principally rent, property taxes and insurance. It also receives construction advances and remits repayments on behalf of the Company. Senior Living Communities, LLC maintains an escrow account designated for certain deposits received from prospective new residents and for refunds currently owed to former residents. Amounts held in this account on behalf of Evergreen Woods Retirement, LLC at December 31, 2019 and 2018 are \$120,600 and \$100, respectively.

Maxwell Group, Inc. provides the Company with management, employee recruitment, accounting, advertising and creative services throughout the year. Amounts paid to Maxwell Group, Inc. for the year ended December 31, 2019 and 2018 are \$1,412,185 and \$1,371,998, respectively. These amounts include management fees and reimbursements for additional services outside the scope of the management agreement.

Related party balances as of December 31 are as follows:

	 2019	 2018
Due to Senior Living Communities, LLC	\$ 5,172,334	\$ 2,192,115

2010

2010

NOTE D - CASH AND RESTRICTED CASH

For purposes of the Balance Sheets and the Statements of Cash Flows, cash and cash equivalents consist of cash and other highly liquid resources, such as investments in certificates of deposit and money market funds with an original maturity of three months or less when acquired. Restricted cash consists of those amounts received from prospective residents required to be held in an escrow account or voluntarily transferred to an escrow account for refunds currently due to former residents. The following table provides a reconciliation of cash, cash equivalents, and restricted cash reported within the Balance Sheets the amounts shown in the Statements of Cash Flows:

	20	2018		
Cash	\$	-	\$	7,346
Occupancy Fees Deposits in Escrow		120,600		100
Total Cash and Restricted Cash shown in the				
Statements of Cash Flows	\$	120,600	\$	7,446

NOTE E - CONCENTRATIONS

Evergreen Woods Retirement, LLC maintains its cash balances at one bank under a sweep investment arrangement. The operating account of Evergreen Woods Retirement, LLC is a zero-balance account. The daily ending balance of the account is cleared by sweep transfers to a master account held by Senior Living Communities, LLC. The balances on hand are combined with those of Senior Living Communities, LLC and its subsidiaries. Funds held in the master account are owned by the respective entities. When funds are required in the operating account to clear checks and other disbursements, automatic transfers from the master account back to the operating account occur for the necessary amounts.

Deposit insurance through the Federal Deposit Insurance Corporation is a function of ownership of the funds on deposit. Each entity participating in the sweep master account is entitled to its own separate deposit insurance up to \$250,000. At times the balances may exceed the insured amounts. The Company periodically reviews the financial condition of the institution and believes the risk of loss to be minimal.

NOTE F - ACCOUNTS RECEIVABLE - OTHER

At December 31,2019, accounts receivable -other represents the amount the Company expects to be reimbursed under its stop loss insurance for its employee benefits plan.

Under the First Amendment to Master Lease between Senior Living Communities, LLC and NHI-REIT of Seaside, LLC, the property owner and landlord, the landlord has agreed to provide funds to the Company to make certain improvements, repairs and renovations to the facilities. Costs incurred for improvements by the Company and expected to be reimbursed under the arrangement at December 31, 2018, total \$54,114.

NOTE G - DEFERRED COMMISSIONS

The Company pays commissions and other move-in incentives directly tied to occupancy agreements that are signed with independent living residents and that require occupancy fees to be paid in advance. Management believes these are recoverable costs. Therefore, the Company has capitalized them as Deferred Commissions in the accompanying Balance Sheets. These costs are amortized over the estimated period the Company will provide services to the residents and over which it earns the nonrefundable portion of occupancy fees received upon move-in. Based on the average residency of an independent living resident across all communities operated by the Company's parent, Senior Living Communities, LLC, the Company expects this to be five years from the date the resident moves in. The related costs are expensed on a straight-line basis over the same five-year amortization period and included in marketing expenses in the accompanying Statements of Operations. The Company recognizes an impairment loss for any unamortized costs related to a contract with a resident who moves out prior to the end of the five-year amortization period. The impairment loss is also recorded as amortization to marketing expense.

NOTE G - DEFERRED COMMISSIONS (Continued)

Amortization of deferred commissions for the years ended December 31, 2019 and 2018 is \$260,764 and \$200,176, respectively.

Prior to January 1, 2018, the Company expensed as incurred all commissions and move-in incentives in accordance with ASC 954, *Health Care Entities*.

The Company expenses as incurred all incremental costs related to acquiring healthcare rental agreements with residents who move into assisted living or skilled nursing under the practical expedient permitted under FASB ASC 340, *Other Assets and Deferred Costs,* because these are considered rental agreements that are renewed monthly.

NOTE H - NOTES PAYABLE

The Company has the following notes payable at December 31:

		2019	2	2018
A note payable to a financing company, monthly payments of \$1,763, including interest at 4.51%, all outstanding principal and accrued interest due October 30, 2021, secured by vehicles.	\$	37,158	\$	56,170
A note payable to a bank, monthly payments of \$3,219,				
including interest at 4.20%, all outstanding principal and				
accrued interest due November 13, 2022, secured by vehicles.		105,678		139,097
		142,836		195,267
Less: current portion	(54,681)	(52,357)
Long-term note payable	<u>\$</u>	88,155	\$	142,910
Maturities of long-term note payable are as follows:				
Year Ending				
December 31		Amount		
2020	¢	F 1 CO 1		

December 31		Amount			
2020	\$	54,681			
2021		53,575			
2022		34,580			
2023		-			
2024		-			
Thereafter		-			
Total	<u>\$</u>	142,836			

The Company is co-borrower on the loan due October 30, 2021. The other borrowers are other subsidiaries of Senior Living Communities, LLC; Wellmore of Daniel Island, LLC whose member is WDI Parent, LLC, which has members in common with Senior Living Communities, LLC; Wellmore of Lexington, LLC, whose member is Wellmore, LLC, which has members in common with Senior Living Communities, LLC; and Stratford Retirement, LLC, which also has members in common with Senior Living Communities, LLC; The total outstanding balance at December 31, 2019 and 2018, as recorded by all borrowers is \$329,432 and \$497,986, respectively. Each borrower has recorded its share of the loan balance based on its proportionate share of the cost of the vehicles that secure the loan. No borrower expects to incur any additional payments on behalf of any of the co-borrowers.

NOTE I - REFUNDABLE OCCUPANCY FEES

The Company recognizes a long-term liability for the unsecured refund obligations associated with occupancy fees received for independent living apartments. The Company offers two forms of occupancy agreements: the "Traditional Plan" and the "Tailored Plan."

The "Traditional Plan" occupancy agreements provide for a refund of 90% or 70% of the occupancy fee paid by the resident, regardless of when the resident moves out. The "Tailored Plan" occupancy agreements provide for a repayment to the resident starting at 90% less 2% per month for each calendar month or fraction of a month from the date the resident paid the occupancy fee until the resident moves out. After forty-five months the refundable portion for the "Tailored Plan" is reduced to zero. The refund obligation is usually due thirty days after a substitute resident has paid the Company the applicable occupancy fee. However, some of these occupancy agreements provide for the refund to be paid one year, eighteen months, or three years after the resident has moved out, even if the unit has not been reoccupied.

NOTE J - DEFERRED REVENUES

A portion of the occupancy fee received from residents represents payment for future services and therefore is nonrefundable. The nonrefundable portion is recorded as deferred revenue and recognized as earned revenue over a period of five years from the resident's move-in date, which approximates the average residency of the residents in independent living across Senior Living Communities, LLC. For "Tailored Plan" occupancy agreements, the deferred revenue is adjusted as amounts become nonrefundable under the terms of the occupancy agreements. Total deferred revenue to be recognized under existing occupancy agreements as of December 31, 2019 and 2018 is \$3,734,249 and \$3,526,131, respectively.

Periodically, the Company re-evaluates the appropriate revenue recognition period for income earned from these contracts.

NOTE K - CONTINGENCIES

It is possible for the Company to incur claims in the normal course of its business. The Company maintains general liability and professional liability insurance policies with coverage and deductibles it believes are appropriate based on the nature of its business and its historical experience. The policies provide coverage on an occurrence basis and are subject to certain deductibles. There are no known claims outstanding as of December 31, 2019 and 2018. The Company routinely reviews the adequacy of its insurance coverage and its accruals for any losses not covered by insurance.

NOTE L - REVENUE

Significant Accounting Policies and Nature of Services

The Company derives all of its revenues from providing an array of services to residents over the period of their residency in its community in North Branford Connecticut. Such services include residential, social, wellness, dining, housekeeping and healthcare in accordance with occupancy agreements which specify the obligations of the community to the resident. The community offers a continuum of level of care units available for residents to reside: independent living, assisted living, and skilled nursing.

The Company offers two types of contracts to residents for independent living: Type B – Modified Contracts, which require an entrance fee upon moving into the community and which include some health services in the initial monthly fee, and market rates are charged for services that exceed those included in the initial fee, and Type C – Fee-for-Service Contracts, which also require an entrance fee, and market rates are charged for health care services on an "as needed" basis. As described in Note I, the Company offers three forms of contracts (occupancy agreements) with independent living residents, the "90% Refund Agreement," the "70% Refund Agreement," and the "Tailored Plan." Type B contracts are referred to as "bundled" plans, and

NOTE L - REVENUE (Continued)

Type C contracts are referred to as "unbundled" plans. The nonrefundable portion of the occupancy fee paid under both contract types represent contract revenue to be recognized over the expected length of time a resident will reside in the community's independent living.

When a resident signs a Type B contract, the monthly service fee charged to the resident includes the same services as the Type C contract plus nursing care in the Health Center for ninety (90) cumulative days without relinquishing the independent living residence. Charges for nursing care beyond the ninety days are dependent upon relinquishing the independent living residence. Therapy, pharmaceutical and medical supplies, personal laundry, equipment rental and additional meals during the healthcare stay are not included, and additional charges at current market rates would be charged for these.

For all Healthcare Rental Agreements, which are with residents who move into assisted living or skilled nursing, a monthly service fee is charged for use of the premises and for the programs, services, and healthcare provided by the community, including direct care, emergency call and fire alarm system monitoring, cable, utilities, daily meal services, housekeeping services, grounds keeping, social and wellness programming, and local transportation. The resident receives the services at a standalone selling price for the service.

The occupancy fee paid by independent living residents gives the resident initial right to the premises and access to additional goods or services though the monthly service fee. Monthly service fees charged to residents are higher on units with a higher percentage refund and on Type B contracts. The nonrefundable portion of the occupancy fee for these contracts is effectively an advance payment on those future services provided by the monthly service fees. As such, a portion should be recognized to revenue as those services are provided.

The Company charges the resident a monthly service fee at the billing rates that the Company establishes from time to time for the programs and services provided by the community, including emergency call and fire alarm system monitoring, cable, utilities, daily meal services, weekly housekeeping services, pest control, grounds keeping, social and wellness programming, parking, and priority admission status to healthcare units. Healthcare services generally are not included for independent living. Monthly services fees are a component of revenue associated with the contracts with residents. The Company recognizes revenue from monthly services fees as services are provided to residents under guidance that applies to contracts that are monthly with the option to renew. Revenue is recognized at the billing rates that the Company establishes from time to time or at the rates established by Medicare for providing services to residents whose costs of care are to be reimbursed under the various programs Medicare administers. Ancillary Income is recognized from services provided to residents on an "as needed" basis and for which a separate charge is assessed.

The nonrefundable portion of the occupancy fees paid represents contract revenues to be recognized over the expected length of time a resident will reside in the community's independent living. The Company recognizes revenue by amortizing a portion of the nonrefundable entrance fee paid in advance for services to be provided over the residents' expected period of residency. The nonrefundable portion is recorded as deferred revenue and is adjusted as it becomes nonrefundable under the terms of the occupancy agreements. This revenue is earned over a period of five years from the resident's move-in date, which approximates the average residency of the residents in independent living across Senior Living Communities, LLC.

NOTE L - REVENUE (Continued)

Disaggregation of Revenue

All revenue is earned as services are provided over time.

	2019	2018	2017
Service Fees:			
Independent Living	\$ 10,404,404	\$ 10,928,970	\$ 10,541,384
Assisted Living	885,486	1,006,237	1,429,194
Skilled Nursing	5,014,920	4,569,534	4,441,583
Other	550,606	467,257	468,341
Total Service Fees	16,855,416	16,971,998	16,880,502
Nonrefundable Occupancy Fees Earned	2,366,497	1,888,155	1,057,530
Ancillary Income	472,055	341,955	375,393
Total Revenues	<u>\$ 19,693,968</u>	<u>\$ 19,202,108</u>	<u>\$ 18,313,425</u>

Skilled nursing service fees are further disaggregated by payor type:

		2019	 2018	 2017
Private Pay	\$	3,051,632	\$ 3,079,204	\$ 3,063,401
Medicare		1,764,824	1,327,160	1,191,319
Other Third-Party Payor		198,464	 163,170	 186,863
Total Skilled Nursing Service Fees	<u>\$</u>	5,014,920	\$ 4,569,534	\$ 4,441,583

Contract Balances

Invoices are sent to residents on the 20th of the month for the following month's monthly service fee and for the prior month's ancillary services that were provided. Medicare is billed in arrears. Amounts due from residents for the prior month's ancillary services, any unpaid monthly service fees for which services have been provided, and the amounts due from Medicare are included in Accounts Receivable in the accompanying Balance Sheets. The Company also records an Allowance for Doubtful Accounts based on its analysis of collectability as the receivables age. The Company records Deferred Commissions for costs related to Type B and Type C contracts.

The Company records a liability for amounts paid in advance by residents for monthly services as well as for refundable occupancy fees and nonrefundable occupancy fees received in advance. Refundable occupancy fees represents amounts due to residents as of December 31 if the resident were to move out of the community. Deferred revenues represents the portion of the occupancy fees received from residents in advance that is nonrefundable as of December 31 that represents advanced payments for services to be provided over the full duration of their residency.

The following table provides information about contract assets related to occupancy agreements with residents at December 31:

	 2019	 2018
Contract Assets:		
Receivables, which are included in Accounts Receivable, Net	\$ 417,013	\$ 433,794
Deferred Commissions, Net	776,411	658,613

NOTE L - REVENUE (Continued)

The following table provides information about contract liabilities related to occupancy agreements with residents at December 31:

	 2019	 2018
Contract Liabilities:		
Monthly Service Fees Received in Advance	\$ 203,585	\$ 205,549
Resident Deposits	52,945	-
Deposits on Occupancy Agreements	599,620	404,001
Refundable Occupancy Fees due currently,		
which are included in Accounts Payable and Accrued Expenses	3,117,530	1,657,345
Refundable Occupancy Fees not currently due	52,530,927	54,323,752
Deferred Revenue, Current Portion	1,724,827	1,439,716
Deferred Revenue expected to be recognized subsequent to		
December 31 of the following year	2,009,422	2,086,415

Significant changes in the contract assets and liabilities during the year ended December 31 are as follows:

	2019		2018	
Contract Assets:				
Receivables:				
Increase (decrease) from revenue billed but not collected	\$	403,863	\$	57,788
(Increase) decrease in Allowance for Doubtful Accounts	(4,016)	(59,863)
Deferred Commissions:				
Increase for commissions capitalized		378,562		858,789
Decrease for amortization and other write-offs	(260,764)	(200,176)
Contract Liabilities:				
Increase (decrease) in cash received, excluding amounts recognized				
as revenue during the year	\$	50,981(\$	106,7	'46)
Increase for Occupancy Fees received from independent				
living residents less amounts recognized as revenue				
during the year		6,138,129		8,037,190
Decrease for Occupancy Fees refunded to former independent				
living residents	(5,160,720)	(5,546,656)

The following table includes estimated revenue expected to be recognized in the future related to performance obligations that are unsatisfied or are partially unsatisfied at the end of the reporting period under occupancy agreements with independent living residents:

Year Ending	
December 31	Amount
2020	\$ 1,881,875
2021	1,842,305
2022	1,579,447
2023	811,819
2024	208,754
Total	<u>\$ 6,324,200</u>

The Company applies the practical expedient permitted in paragraph 606-10-50-14 and does not disclose information about remaining performance obligations that have original expected durations of one year or less.

NOTE M - OPERATING LEASES

The lease agreement between NHI-REIT of Evergreen, LLC and Senior Living Communities, LLC and its sub-lease with Evergreen Woods Retirement, LLC expires on December 31, 2029, with an option to renew for two additional five-year terms. Under the agreement, the lessee is responsible for all operating costs, maintenance and repairs, insurance, and property taxes. The Company also leases various types of equipment with terms ranging from month-to-month to four years.

The components of rent expense for the year ended December 31, 2019, are as follows:

Operating Lease Cost - Facility	\$ 6,421,785
Variable Lease Cost - Facility	
Property Insurance	139,674
Property Taxes	1,648,375
Operating Lease Cost - Equipment	17,961
Short-term Lease Costs	 11,454
Total Lease Cost	\$ 8,239,249

Supplemental Balance Sheet information related to leases for the year ended December 31, 2019, is as follows:

Operating Lease:	
Operating Lease Right-of-Use Asset	\$ 45,436,305
Accumulated Amortization	(2,741,067)
Operating Lease Right-of-Use Asset, Net	<u>\$ 42,695,238</u>
Current Operating Lease Liabilities	\$ 2,366,496
Noncurrent Operating Lease Liabilities	43,450,962
Total Operating Lease Liabilities	<u>\$ 45,817,458</u>
Other information:	
Weighted average remaining lease term - operating leases	10.0 years
Weighted average discount rate - operating leases	8.0 %

The following is a schedule of future minimum base rental payments for the facility over the next five years and in total:

Year Ending				
December 31		Amount		
2020	\$	5,906,990		
2021		6,073,932		
2022		6,248,966		
2023		6,436,434		
2024		6,629,528		
Thereafter		36,252,975		
Total	<u>\$</u>	67,548,825		

NOTE N - 401(k) PROFIT SHARING PLAN

The Company, as part of the controlled group with Senior Living Communities, LLC and its subsidiaries, participates in the 401(k) plan of Senior Living Communities, LLC. Employees who are 21 years of age or older are immediately eligible to participate in the plan and are eligible for matching contributions. Employees become fully vested in the employer contributions to the plan after one year of service. The Company's matching contribution to the plan is discretionary. Currently the Company's matching contribution equals 20% of up to 5% of compensation. Employer matching contributions for the year ended December 31, 2019, 2018 and 2017 are \$44,969; \$44,573 and \$39,666, respectively.

NOTE O - SUBSEQUENT EVENTS

The Company evaluated transactions occurring after December 31, 2019 in accordance with ASC 855, *Subsequent Events* through June 25, 2020, which is the date the financial statements were available for issuance.

The World Health Organization declared the coronavirus (COVID-19) outbreak to be a public health emergency on January 30, 2020 and a pandemic on March 11, 2020. The Company is currently evaluating the impact of the pandemic. As an operator of a continuing care retirement community, the Company has been deemed an essential business by state and local authorities and continues to serve its residents. Although occupancy levels have remained relatively stable, they could be negatively impacted by further weakness in the overall economy and/or disruptions in the demand for senior living communities. This, along with incremental costs to prepare for and respond to the pandemic, as well as price increases to preexisting expenses, could have a material adverse effect on the Company's business, results of operations, financial condition, and cash flows.

In April 2020, the Company applied for and was approved for a loan pursuant to the Paycheck Protection Program (PPP) established by the Coronavirus Aid, Relief, and Economic Security Act (CARES Act) in the amount of 1,221,600. The loan proceeds were received on April 28, 2020. Under the terms of the PPP, payments will be deferred for six months, the loan will bear interest at 1% per annum, and will have a two-year maturity, with an option to modify such to five years upon mutual agreement between the Company and the lender. No collateral or personal guarantees are required. Subject to various eligibility and certification requirements, some or all of the loan may be forgiven; however, the amount and timing of any forgiveness is uncertain.

In April and May of 2020, the Company also received \$266,394 pursuant to the CARES Act Provider Relief Fund targeted towards healthcare entities and providers. Retention and use of these funds are subject to certain terms and conditions. If these terms and conditions are met, payments do not need to be repaid at a later date. Compliance with these terms and conditions is ongoing as of the date the financial statements were available for issuance and therefore the ultimate retention of these funds is uncertain.